Leadership for change management; critical success factors in leading change in medical imaging departments in major teaching hospitals in NSW

> Munyaradzi Gwede Candidate ID: 17311348

A thesis in fulfilment of the requirements for the degree of Doctor of Philosophy

> May 2022 School of Business Western Sydney University

Dedication

This thesis is dedicated to my children Tavonga, Makatendeka and Munotidaishe who have endured the limitations on their play time with me as I completed this thesis. You have been very supportive in a very huge way, forgoing your fair share of time with me while I toiled to see this major milestone to the end. I have a huge amount of catching up to do.

To my wife, Persy, you have been a pillar of strength. Steadfastly patient while I undertook yet another course! Though I said this may be the last one, I am not so sure! You have been a resolute warrior and have unflinchingly supported me through this journey. You have said countless times that you have done the heavy lifting with me. I could not agree with you any less.

To my mum and mother-in-law, your support in this audacious project has been invaluable. I would certainly not have completed it without you.

To my sisters Charity and Brenda being older than you has been enough motivation to do this before you!

And finally, to my young brother Brian, well, I am short of words to describe how you have contributed to the cause.

Acknowledgments

I would like to acknowledge the following people who supported me and have

been invaluable in the production of this thesis:

My supervisors: Associate Professor Terry Sloan Dr Louise Kippist Professor Josephine Chow

My supervisors were the guiding light when I was stuck and struggling to find alternatives. I truly appreciate your guidance throughout this remarkable journey, which at times seemed to be so far away.

Dr Peter Lin, Nuclear Medicine Physician, Clinical Stream Director, Medical Imaging, SWSLHD. You supported me throughout this journey, offering some invaluable insights.

Heads of Departments in Medical Imaging, for opening your departments for my research: Dr Glen Schlaphoff; Dr Sacha Kobilski; Dr Preeta Philips; Dr Leon Lam and Dr Kien Lee.

Arnold Tammekand, General Manager, Fairfield Hospital for supporting me and allowing me the time to balance my work and study requirements.

Leslie Everson, my former manager at Bupa Villages and Aged Care Australia, for your interest in my research and affording me the support to balance my research with work.

To all those that I have not mentioned by name, thank you for your support. It is greatly appreciated.

Statement of Authentication

The work presented in this thesis is, to the best of my knowledge and belief, original except as acknowledged in the text. I hereby declare that I have not submitted this material, either in full or in part, for a degree at this or any other institution.



Date

Declarations

I declare that this study has not been funded by any external agencies other than the Commonwealth of Australia through the Higher Degree Research Program.



Date

.....

Table of Contents

List of Tables 2
List of Figures and Illustrations
Abbreviations4
Abstract5
Chapter 1 – Introduction8
1.1 Introduction
1.2 Background to the Research9
1.2.1 The Australian Healthcare System9
1.2.2 The NSW Health System 11
1.2.3 Local Health Districts (LHDs) 14
1.2.4 Organisational Structure15
1.2.5 Healthcare Changes and Medical Imaging Services
1.2.6 Activity Based Funding (ABF) 21
1.2.7 Emergency Treatment Performance
1.2.8 The Growth of Medical Imaging25
1.2.9 Medical Imaging Departments in NSW
1.2.10 Equipment Procurement in Medical Imaging
1.2.11 Patient Considerations
1.3 Research Problem, Propositions and Contributions

	1.4 Motivation for This Research	. 35
	1.5 Justification for The Research	. 35
	1.6 Research Outcomes	. 36
	1.6.1 Additional Outcomes to Practice	. 37
	1.7 Methodology	. 37
	1.7.1 Scope	. 37
	1.8 Outline of The Report	. 38
	1.8 Conclusion	. 40
С	hapter 2 – Literature Review	41
	2.1 Introduction	. 41
	2.2 Literature Review	. 41
	2.3 Leadership	. 42
	2.4 Management	. 43
	2.5 Clinical Leadership and Management in Health Care	. 45
	2.5.1 Medical Leadership	. 46
	2.6 Leadership Theories	. 48
	2.6.1 Behavioural Theories	. 48
	2.6.2 Trait Theories	. 49
	2.6.3 Contingency Theory	. 50
	2.6.4 Path-Goal Theory	. 52
	2.6.5 Leader-Member-Exchange (LMX Theory)	. 53
	2.6.6 Power and Influence Theories	. 54
	2.7 Leadership Styles	. 57
	2.7.1 Consultative Leadership	. 57
	2.7.2 Authentic Leadership	. 58

	2.7.3 Transactional Leadership	60
	2.7.4 Transformational Leadership	63
	2.7.5 Laissez - Fairer Leadership	65
	2.7.6 Servant Leadership	65
	2.7.7 Variety of Leadership Styles in Healthcare	67
	2.8 Research Questions 1 and 2	68
	2.9 Change Management	68
	2.10 Research Questions 3 and 4	73
	2.11 Learning Organisations	73
	2.12 Research Question 5 and 6	74
	2.13 Critical Success Factors	74
	2.14 Conclusion	75
С	Chapter 3 - Methodology	77
	3.1 Introduction	77
	3.2 Overview of Research Model	78
	3.3 Justification for The Paradigm and Methodology Design	80
	Step 1. Ethics Approval	81
	Step 2. Participant Considerations	82
	Step 3. Population Data Review	83
	Step 4. Identification and Development of Data Collection Tools	83
	Step 5. Identification of Change Management Projects	83
	Step 6. Establish a Participant Cohort	84
	Step 7. Obtain Informed Consent	85
	Step 8. Undertake Baseline Leadership Style Inventory (LSI)	85
	Step 9. Design of Interview Tools	. 86

Step 10. Conduct Pilot Interviews 87
Step 11. Participant Interviews 88
Step 12. Review and Measure Project Success
Step 13. Thematic Data Analysis 89
Step 14. Undertake Final Leadership Style Inventory
3.4 Research Methodology Further Considerations
3.4.1 Further Ethical Considerations
3.4.2 Research Process and Multiple Data Collection Rounds
3.4.3 Researcher Ongoing Communication with Participants
3.4.4 Baseline Qualitative and Quantitative Project KPIs
3.4.5 Sampling Techniques Utilised95
3.4.6 Leadership Style Inventory/Questionnaire
3.4.7 Data Collection Rationalisation101
3.4.8 Discussion 101
3.5 Summary of Tools Used for Data Collection
3.6 Congruency of Methodology, Data Collection and Research Questions 103
3.7 Ethical considerations 104
3.7.1 The Researcher's Knowledge of Medical Imaging Departments 104
3.7.2 Selection Bias 105
3.8 Confounders 105
3.8.1 The Element of Luck on Project Success
3.8.2 External Factors Impacting on Project Success
3.9 Conclusion 106
Chapter 4 – Results 107
4.1 Introduction and Overview

4.2 Sample Results	108
4.2.1 Characteristics	108
4.2.2 Project Demographics	110
4.3 Pilot Study	114
4.4 Leadership Style Inventory (LSI)	114
4.4.1 Authentic Leadership Questionnaire	114
4.4.2 Multifactor Leadership Questionnaire	117
4.5 Round One Interviews	120
4.6 Round Two Interviews	121
4.7 Round Three Interviews	122
4.8 Dominant Leadership Styles in the Local Health District	123
4.8.1 Nature of Leadership and Impact on Success	128
4.9 Reasons for Project Success Or Failure	129
4.9.1 Stakeholder Engagement	130
4.9.1 Stakeholder Engagement 4.9.2 Planning	130 134
4.9.1 Stakeholder Engagement4.9.2 Planning4.9.3 Education and Training	
 4.9.1 Stakeholder Engagement 4.9.2 Planning 4.9.3 Education and Training 4.9.4 Resetting Expectations 	
 4.9.1 Stakeholder Engagement 4.9.2 Planning 4.9.3 Education and Training 4.9.4 Resetting Expectations 4.9.5 Processes and Procedures 	
 4.9.1 Stakeholder Engagement 4.9.2 Planning 4.9.3 Education and Training 4.9.4 Resetting Expectations 4.9.5 Processes and Procedures	
 4.9.1 Stakeholder Engagement	
 4.9.1 Stakeholder Engagement	
 4.9.1 Stakeholder Engagement	
 4.9.1 Stakeholder Engagement 4.9.2 Planning 4.9.3 Education and Training 4.9.4 Resetting Expectations 4.9.5 Processes and Procedures 4.9.6 Other Reasons for Project Success 4.10 Strategies Utilised for Project Success 4.10.1 Consultation and Engagement 4.10.2 Workload Management 4.10.3 Education and Training 	
 4.9.1 Stakeholder Engagement	
 4.9.1 Stakeholder Engagement	

	4.11 Leadership Style Influence from Learned Experience	162
	4.12 Conclusion	164
С	hapter 5 - Discussion	168
	5.1 Introduction	168
	5.2 Research Framework and Recap	168
	5.2.1 Research Conceptual Framework	170
	5.3 Overview of Findings	171
	5.3.1 Family Values and Lifestyle Choices	174
	5.3.2 Organisational Support	175
	5.3.3 Project Types Found in Medical Imaging	176
	5.3.4 Communication Styles That Work	177
	5.4 Dominant Leadership Styles in Medical Imaging Departments	179
	5.5 Leadership Styles and Change Management Outcomes	184
	5.6 Strategies Used for Successful Change Management	186
	5.6.1 Consultation and Engagement	186
	5.6.2 Effective Leadership	188
	5.6.3 Workload Management	189
	5.6.4 Explained Communication	192
	5.7 Why Do Some Projects/Processes Fail or Succeed?	193
	5.7.1 Stakeholder Engagement	193
	5.7.2 Planning	196
	5.7.3 Education and Training	197
	5.7.4 Resetting Expectations	198
	5.7.5 Processes and Procedures	198
	5.8 Leadership Style and Learned Experience	202

5.9 How Do "Learning" Leaders Fare in Future Projects?	5
5.10 Conclusion	3
Chapter 6 – Contribution, Conclusions, and Implications)
6.1 Introduction 210)
6.2 Research Conclusions 210)
6.2.1 Dominant Leadership Styles 210)
6.2.2 Strategies Used for Successful Change Management	2
6.2.3 Nature of Leadership and Change Management Processes	5
6.2.4 Why Do Some Projects/Processes Fail or Succeed?	3
6.2.5 Learning Leaders, Organisations, and Project Outcomes	3
6.3 Scholarly Contribution	3
6.4 Implications for Policy and Practice in Medical Imaging 222	2
6.4.1 Stakeholder Mapping 222	2
6.4.2 Project Documentation 223	3
6.4.3 Change Management and Project Management Training 224	1
6.4.4 Private Sector Managers 224	1
6.4.5 Public Sector Policy Analysts and Managers 225	5
6.5 Limitations 229)
6.6 Further Research 229	9
6.7 Conclusion)
References Cited 232	
List of Appendices 275)
Appendix A – Functions of the NSW Entities	3

Appendix B - Ethics Approval
Appendix C - Ethics Approval – Site Specific Authorisation
Appendix D – Participant Consent Form 284
Appendix E - Multifactor Leadership Questionnaire
Appendix F - Authentic Leadership Questionnaire
Appendix G – MLQ sample Results Interpretation
Appendix H – MLQ international normative samples – Oceania
Appendix I - Categorisation of Results from The Completed MLQ
Appendix J - Interview Questionnaire Round 1 306
Appendix K - Interview Questionnaire Round 2
Appendix L - Interview Questionnaire Round 3
Appendix M – Interview Transcript Coding
Appendix N - Round 1 – Sample Interview Transcript

List of Tables

Table 1 – Leadership Theory Summary 56
Table 2 – Authentic Leadership Domains 59
Table 3 – Transformational Leadership Dimensions 62
Table 4 – Leadership Styles Key Characteristics 66
Table 5 – Dimensions Of Transformational Leadership
Table 6 - Summary Of Tools Used For Data Collection 103
Table 7 – Project Duration By Round
Table 8 – Projects And Stakeholders Involved 112
Table 9 - Project By Type And Driver 113
Table 10 – Participant Results For The ALQ 116
Table 11 – Baseline And Final MLQ Results And Differences Over Time 117
Table 12 – Participation In Data Collection Processes
Table 13 - Self Reported Leadership Styles 127
Table 14 – Stakeholder Types 132

List of Figures and Illustrations

Figure 1: NSW Health Organisational Structure	13
Figure 2: Organisational Structure Of A Local Health District	17
Figure 3: Budget Illustration A Local Health District	22
Figure 4: Reporting Patient Costs Through The NHCDC	23
Figure 5: Thesis Outline	39
Figure 6: Sequence Of Research Methodology Steps	81
Figure 7: Data Collection Processes	91
Figure 8: Research Methodology Matrix	104
Figure 9: Participant By Role Type	109
Figure 10: Participants' Occupations By Classification	110
Figure 11 – Project Success And Leadership Style Utilised	129
Figure 12: Key Reasons For Project Success	130
Figure 13: Strategies Used For Project Success	146
Figure 14: Conceptual/Research Framework	170

Abbreviations

ABF	Activity Based Funding
ACI	Agency for Clinical Innovation
CEC	Clinical Excellence Commission
eMR	Electronic Medical Record
HREC	Human Research Ethics Committee
IMTD	Information Management and Technology Department
MRI	Magnetic Resonance Imaging
NEAT	National Emergency Access Target
NHCDC	National Hospital Cost Data Collection
NHHRC	National Health and Hospital Reform Commission
NHRA	National Health Reform Agreement
NSW Health	New South Wales Ministry of Health
PBS	Pharmaceutical Benefits Scheme
SLHD	Sydney Local Health District
SME	Subject Matter Expert
SWSLHD	South Western Sydney Local Health District
ТАТ	Turnaround Time

Abstract

Successfully leading change in public medical imaging departments is an enigma that has eluded most public health practitioners particularly in the Australian healthcare context. Research in other contexts has shown that at least 70% of change projects fail to achieve their intended objectives. The area of leadership in medical imaging is an under researched area of healthcare leadership. While various scholars (Barnard 2020; Grant-Smith & Colley 2018) have proposed that strong leadership is required for change management, none of them have fully discussed what this means.

This study setting is the medical imaging departments of public hospitals in a New South Wales Local Health District. The primary purpose of this research was to explore the leadership styles and critical success factors that led to successful change management projects or processes. The research utilised a mixed methods approach in a longitudinal study to answer six research questions. There is scant evidence of similar Australian studies utilising the same approach in medical imaging in NSW. In this research, leadership style changes of medical imaging leaders were studied as leaders undertook cycles of successive change management projects over a three-year period. Five data collection rounds were undertaken with the same participant cohort, including two leadership style inventories and three rounds of semi structured interviews.

This research has contributed to scholarship in three main ways: the use of a new approach in studying the research phenomena; answering the six research questions providing new insights into leadership in medical imaging in NSW; and by exploring the concept of strong leadership, defining, and understanding it. This research defined strong leadership as the ability to successfully implement change within an organisation. Leaders achieved change through the mechanism of a systematic process though which leadership activities are

undertaken, which influenced the attainment of organisational goals through the management of conflict, competing demands and divergent views within a complex organisation.

The research found that the dominant leadership style for successful change management in medical imaging in NSW was consultative leadership, contrary to the pre-existing literature on the widely practised leadership styles in healthcare situational, transactional, and transformational styles were the other prominent leadership styles found in this research.

This research further found that the use of the key strategies of consultation and engagement; workload management; education and training; communication and planning were critical enablers to project success. Similarly, stakeholder engagement; planning; education and training; resetting expectations and processes and procedures were the key reasons that drive project success in medical imaging.

This research also found that the nature of leadership impacted on change management processes in healthcare. For example, a consultative leadership approach positively influenced the change management processes. Following previous change management projects, this research found that medical imaging leaders' leadership styles were influenced by prior experience. Their leadership style was reshaped by learned experience of past change management. This meant that learning leaders achieved better outcomes in successive change management processes.

The significance of this research is that it presents new knowledge on leadership in medical imaging in healthcare. In doing so, it proposes new ways and strategies of successfully leading change management projects in healthcare. This research also presents new knowledge on what effective leadership in medical imaging is and how it is characterised.

Chapter 1 – Introduction

1.1 Introduction

Rapid and timely diagnosis of diseases is a crucial tenant in the management of diseases (Li, J et al. 2021). Medical imaging tests including diagnostic and interventional radiology, and nuclear medicine facilitate the early diagnosis and intervention in disease management. Such services are offered in major tertiary public and private hospitals within New South Wales (NSW) and other parts of the world, for this purpose. Like other healthcare services, medical imaging processes are led by clinical and non-clinical personnel, who undertake a range of service delivery functions.

Under the research topic: "Leadership for change management; critical success factors in medical imaging departments in major teaching hospitals in NSW", this study investigated how the leadership and change management processes implemented in healthcare affect the outcomes of the change management process in medical imaging departments. The research was conducted at a micro setting, meaning with floor leaders or frontline leaders such a Senior MRI Radiographers as opposed to senior Chief Executive level leaders. The setting of the research is a NSW Health Local Health District (LHD), one of 18 such entities providing healthcare to the NSW public. This LHD called South Western Sydney Local Health District is the largest and fastest growing district in metropolitan Sydney, which is also one of the most geographically spaced areas in NSW (South Western Sydney Local Health District 2019). This chapter, will include, firstly, a background to the research including a synopsis of the Australian healthcare system and secondly discussion of the NSW Health System which also includes the LHD structure is discussed. Thirdly, a brief discussion on healthcare changes follows and before activity-based funding (ABF) and the associated disease classification process is introduced. Then the research problem, motivation and justification for the research follows. The research questions are then introduced, together with the anticipated research outcomes. The conclusion of the chapter provides an outline of the research thesis. By exploring the leadership styles employed by medical imaging leaders, this research will provide a better understanding of the leadership styles employed by such leaders which lead to successful change management projects or processes. The next section provides some background information about the Australian private and public healthcare systems as well as medical imaging.

1.2 Background to the Research

1.2.1 The Australian Healthcare System

The Australian healthcare system follows the federal government system with the 3 governance levels at national (Commonwealth), State and local government levels. These levels manage different components of the healthcare system. According to Australian Institute of Health and Welfare (2021) the Commonwealth provides the broad functions of: developing national health policy; funding Medicare through the Pharmaceutical Benefits Scheme; funding state and territory public hospitals and other functions. State and territory governments; fund and manage public hospitals; regulate and licence private hospitals and other health premises and deliver community based and preventative programs. Finally, local governments undertake the delivery of community and home-based health and support services, environmental health services and other public health services. Within the Australian health context, private and public hospitals provide care to citizens. According to Australian Institute of Health and Welfare (2009a), private hospitals are privately owned and operated institutions catering for patients treated by their doctor of choice, where patients are charged fees for accommodation and practitioner fees. Similarly, a public hospital is defined as one that is operated by, or on behalf of, the government of the state or territory in which it is established. This includes hospitals which are owned by private or charitable groups but are authorised or contracted by the government to deliver public hospital services (Australian Institute of Health and Welfare 2009b).

This research is only focussed on public hospitals. The key instrument underpinning the operational arrangements of public hospitals in Australia is the National Health Reform Agreement 2011, signed by all states and territories in August 2011. According to the NHRA (2011), the reforms' primary objectives were:

- Reforming the fundamentals of our health and hospital system, including funding and governance, to provide a sustainable foundation for providing better services now and in the future.
- Changing the way health services are delivered, through better access to high quality integrated care designed around the needs of patients, and a greater focus on prevention, early intervention, and the provision of care outside of hospitals.
- Providing better care and better access to services for patients, through increased investments to provide better hospitals, improved infrastructure, and more doctors and nurses.

The NHRA (2011), set the tone for further refinements in the organisational structures implemented within the states. It also led to the establishment of the Local Health Networks (now known as Local Health Districts) within states and territories. Profligate expenditure on healthcare has continued (Briggs 2017), as life expectancy continues to increase for Australians. The Australian Institute of Health and Welfare (2014) estimated that 9.67% of GDP¹, or \$147.4billion, was spent on healthcare in 2012/2013 an increase of 1.5% from the prior year. This was significantly lower than the average 5.1% increase recorded since the 1980s.

The Department of Health (2020b) states that chronic illnesses have also continued to grow. However, new technology and the advancement of medicine has resulted in improved management of such diseases and consequently increased life expectancy for Australians to about 80 years for males and 84 years for females born in 2016/2018 (Australian Insititute of Health and Welfare 2021).

The evolution of the healthcare system has necessitated attempts to improve, accountability, performance, value for money and mere political expediency in the Australian healthcare system (Briggs & Isouard 2016; Duckett & Willcox 2015; Javanparast et al. 2018; Kay & Boxall 2015) including states, such as the NSW Health system, which is discussed in the next section.

1.2.2 The NSW Health System

The NSW public health system has undergone significant organisational metamorphosis, which include the previous Area Health Service model (Javanparast et al. 2018; Liang et al. 2013; NSW Health 1997, 2015). The key influences on the current structure include the Garling Enquiry (Garling 2008) and

¹ Gross Domestic Product

the National Health and Hospital Reform Commission's (NHHRC) (Jowsey et al. 2011) National Health Reform Agreement (NHRA) 2011, as discussed above. Following the establishment of the LHDs across the State, the next paragraphs discuss their structures and governance arrangements.

The NSW government is part of state and territory parliaments and local government with shared responsibilities for healthcare policy, financing and delivery (Briggs 2017). The governance of the parliaments reflect that of elected government and such governance is known as the Westminster model (Chaplin 2020). As such, the NSW Ministry of Health being responsible for public health in the State. Figure 1 below details the structure of NSW Health, generally comprising of 15 LHDs and three specialty networks (NSW Health 2015). The LHDs are supported by six pillars namely, Agency for Clinical Innovation (ACI), Bureau of Health Information (BHI), Cancer Institute NSW, Clinical Excellence Commission (CEC), Health Education and Training Institute (HETI) and NSW Kids and Families. In addition, other NSW Health entities provide operational support and include state-wide services such as: NSW Ambulance, Health Infrastructure, Health Protection NSW, HealthShare NSW, NSW Health Pathology, Office of Health and Medical Research, St Vincent's Health Network, and eHealth. The detailed nature and function of these entities is presented in Appendix A.

FIGURE 1: NSW HEALTH ORGANISATIONAL STRUCTURE



*Service Compact — Instrument of engagement detailing service responsibilities and accountabilities [†]No Service Compact between Ministry of Health and NSW Ambulance

Source: NSW Health 2022. Accessed on 5 April 2022 from: http://www.health.nsw.gov.au/about/nswhealth/Pages/chart.aspx

1.2.3 Local Health Districts (LHDs)

As shown in Figure 1 above, NSW currently has a total of 15 LHDs and three specialty networks. Eight cover the metropolitan regions and seven cover rural/regional areas. Of the three specialty networks, one covers three public hospitals operated by St Vincent's health while the other two provide children and paediatric services and forensic mental health, respectively. LHDs are led by a Board of directors appointed by the shareholder, who is the Minster for Health. The Board's role is focused on leading, directing and monitoring the activities of the LHDs and specialty networks and driving overall performance. Section 28 of the Health Services Act 1997 describe these functions which are summarised below (NSW Health 2015):

- The establishment of effective clinical and corporate governance frameworks.
- To support the maintenance and improvement of standards of patient care and services by the LHDs and to approve those frameworks.
- To ensure strategic plans to guide the delivery of services are developed for the LHD and to approve those plans.
- To approve systems that support efficient, economic, and budget conscious service delivery models that meet performance targets in accordance with the state-wide performance framework.
- To ensure equitable allocation and resource utilisation.
- To make recommendations regarding Chief Executive (CE) appointments or dismissals.
- To provide the Chief Executive of the LHD with the operational performance targets and performance measures to be negotiated in the service agreement for the district under the NHRA.
- To approve the service agreement for the LHD under the NHRA.
- To seek and engage providers, consumers, and other key stakeholders in the provision of health services.

Some of these functions are delegated to the Chief Executive (CE) who exercises them operationally under Board oversight. The CE's role is also espoused in legislation, under section 24 of the Health Services Act 1987 and is to manage and control the affairs of the LHD. Accountable to the Board, the CE is the principal officer of the LHD, with the delegation for contractually and legally committing the organisation. In contrast to the Board member appointment process by the shareholder, represented by the Minister for Health, the CE is appointed by the Secretary for Health (formerly the Director-General) in terms of section 116 of the Health Services Act 1997. All 18 CEs, the Secretary for Health and other Ministry of Health senior executives form a group called the NSW Health Senior Executive Forum, which drives the NSW Health agenda with their support teams.

1.2.4 Organisational Structure

Ahmady, Mehrpour and Nikooravesh (2016), building on the work of Mintzberg (1972) suggest that organisational structure is a framework or system where tasks and duties are determined, divided, and coordinated to achieve organisational goals. The administration of the LHDs differs widely despite a common overarching governance framework. Several reasons including responsiveness to local requirements necessitates these differences. This research was conducted in a large LHD in NSW, which has undergone several restructures as discussed earlier. Like many Australian healthcare organisational structure during the research was undertaken had a matrix organisational structure during the research period (Figure 2). Under this structure, there are 17 clinical directorates and networks, each headed by a Clinical Stream Director. The Clinical Stream Director reports directly to the CE and their function is to provide strategic advice to the CE and support effective governance across the facilities (hospitals and community health centres). While some literature supports clinical directorates due to their advantages in adapting and providing closer interaction

with the patient (Tuckett et al. 2015); facilitate a collaborative multidisciplinary approach and reduce interdisciplinary rivalry (Clarke, A et al. 2017); improve efficiency (Parthasarthy 2018), resource management (Gerring 2016) and service delivery (Stefanini et al. 2020), some other authors are not so convinced. Braithwaite and Westbrook, as reported by Long, Cunningham and Braithwaite (2013) found that the introduction of clinical directorates had increased organisational politics, while Sorensen et al. (2013) suggested that conflict arose between professions working in multidisciplinary teams. Coyle, Degeling and Gray (2017) further suggested that hybrid clinician managers' clinical autonomy was eroded in these structures.

Braithwaite (2018) found there are no discernible differences in organisational efficiency for restructured organisations either to clinical directorates or otherwise. It could be argued that there is no single acceptable organisational structure as different organisational structures have their advantages and disadvantages. Key considerations for organisations contemplating organisational structural changes are the balance between the pros and cons of alternative courses of action.

Braithwaite et al. (2017) and Edwards, N and Saltman (2017) recognised the complex nature of public health organisations as political; multi-stakeholder, and multi-dimensional, which suggests that different organisational structures that consider this complexity could work. For example, the presence or absence of a large Aboriginal population or migrant community may mean that one LHD may need to consider a structure that focuses on primary healthcare and health promotion strategies compared to another LHD with a different demographic composition.

FIGURE 2: ORGANISATIONAL STRUCTURE OF A LOCAL HEALTH DISTRICT



Source: South Western Sydney LHD Intranet. Accessed on 15 April 2015.

Figure 2 above depicts the organisational structure discussed in section 1.2.4. This hierarchical structure shows an overview of the matrix structure within the LHD. Areas of responsibility for each directorate and executive are summarised in the organisational structure above. Of note, under the purview of the Director Operations are the hospitals and other services that provide direct patient care (facilities and services). While most services are provided at the local hospital level, integrated networks at the clinical stream level exist to provide a strategic role for the clinical and non-clinical services, including standardising discipline specific clinical practices across the LHD (South Western Sydney Local Health District 2020). Healthcare changes have invariably made an impact on medical imaging services, and these are discussed in the next section.

1.2.5 Healthcare Changes and Medical Imaging Services

Health reforms in Australia between 2008 and 2011 by the Australian Labour governments precipitated a further number of changes in the operations of hospitals. The implementation of the Health Reform in NSW and wider Australia had major implications in the provision of healthcare generally and medical imaging specifically, through increasing demand among other considerations. Hollnagel and Braithwaite (2019), Kuziemsky (2016) and Erasmus et al. (2014) found that wider political and social contexts also influence complex adaptive systems such as health facilities. Over the past 25 years, medical imaging services' growth as service centre has been accelerated, particularly within the political context and health reforms. Compared with their position in the recent past, they now form an integral part of healthcare provision in public hospitals. Funding implications, timely patient access to services and a renewed focus on efficient resource utilisation have meant that imaging services have rapidly For example, Interventional Radiology and Interventional grown. Neuroradiology have become significant medical imaging specialties. Previously, medical imaging could have been absorbed as under a Medical or Surgical directorate's budget, however, the changes outlined above mean that these service centres (medical imaging) are standalone more autonomous services with their own directorates, as is the case in the South Western Sydney LHD.

According to the Council of Australian Governments (2011), the National Health Reform Agreement (NHRA) 2011 resulted in several significant changes to hospital funding and operations in most Australian jurisdictions including the introduction of key performance indicators (KPIs). As most states were embracing Activity Based Funding (ABF), a new layer of complexity was introduced for medical imaging managers. Veronesi et al. (2014) discussed ABF as a new, transparent funding model where hospitals are funded according to activity performed with hospitals with potential to reduce waiting lists. This means that, if a hospital undertakes 50,000 episodes of care, the funding received by the hospital is a subset of all inputs in the episode of care. While this concept may seem quite straightforward, this is not necessarily the case for medical imaging departments. If each episode of care is allocated for example a CT scan, the cost of a CT scan is factored into the funding model. However, if an episode of care requires more than a single CT scan, then it creates a funding shortfall for the extra medical imaging activity. As the overall funding available to hospitals is activity dependent, how the funding is distributed within health services is problematic for medical imaging as it is not necessarily linked to the number of examinations performed per year. The implications for medical imaging services in this scenario would be the provision of multiple tests and only receiving funding for a single study. ABF is more fully discussed in section 1.2.5.

The NHRA linked hospital funding to performance. For example, Emergency Treatment Performance (ETP) was set as a performance benchmark within which hospitals were expected to perform. For health services to meet these mandated benchmarks, a team approach is required from all services involved including medical imaging departments, to minimise delays in service provision and contribute to the timely provision of imaging services. A detailed discussion on ETP is provided in section 1.2.7. Today's medical imaging leader therefore faces new challenges in the provision of medical imaging services. The skillset previously adequate in the past to manage medical imaging departments is no longer adequate to manage these emerging challenges. Such challenges include the need to successfully lead change, optimising patient outcomes with minimal resources and increased innovation in service delivery among others.

Since the emergence of newer imaging technology, rationalisation of processes and appropriate resources utilisation is progressively becoming more important. Today's medical imaging manager is increasingly considering value for money propositions in financial decision making, which has led to the establishment of new roles in medical imaging such as business managers and operational managers. The business manager in medical imaging is responsible for establishing and maintaining systemic commercial processes; leading the establishment of new processing; providing commercial advice to other medical imaging leaders and leading the financial management function within medical imaging services. The operational manager is responsible for the leadership of operational processes across the medical imaging stream within the LHD. Working collaboratively, medical imaging leaders, are consistently required to manage quality in imaging. This includes the reduction in radiation exposure, procurement of appropriate equipment and clinical practices employed in the acquisition of images.

When x-ray technology commenced over 100 years ago, it became a gold standard for the provision of care where it was indicated. Griffin and Treanor (2017) found that improvements in how this technology was utilised have resulted in efficiency gains from financial and workflow perspectives. In the last 25 years CT usage has increased dramatically year on year. For example, data from the LHD shows that over the 10 year period between 2011 and 2019 the number of CTs ordered across the hospitals have increased by an average 19%

compared to hospital admissions that increased by about 3-4% annually (Lin 2021). This demonstrates that CT has become the test of choice, possibly due to the superior image quality provided when compared to x-rays. It can be reasonably expected that MRI will move in the same direction in the future for appropriate patients. While the demand for x-rays remains due to some patients (such as those with pacemakers or metal prosthesis) being unable to access MRI which uses magnet technology, it is evident that consumers are considering the significant radiation exposure risks prior to utilising these services (Government of Western Australia 2021d). Further discussion on the growth of medical imaging is undertaken in section 1.2.8.

The health reform sought to make improvement in the health system. For example, the National Health Reform Agreement (2011) aimed at improving patient access to services and public hospital efficiency through ABF among other objectives. Other initiatives stipulated as a result included the National Emergency Access Targets (NEAT) now known as Emergency Treatment Performance (ETP). A brief discussion on these key initiatives follows.

1.2.6 Activity Based Funding (ABF)

According to the Independent Health Pricing Authority (2019) (IHPA), ABF is a funding model where hospitals get paid by the funding agency, for the patient number and mix. A simplistic view of ABF is that funding is aligned to hospital activity, unlike the traditional block funding models that existed prior to the introduction of ABF in NSW and other states. Medical imaging is funded for inpatients through this process and through Medicare for outpatients. The next paragraphs describe a high-level application of this process. Except for Victoria where ABF has been a key part of public health expenditure and funding, the rest of the states had to make significant changes to their funding models.

In an ABF environment, there is a provider/purchaser arrangement where the NSW Ministry of Health and the Commonwealth are the purchasers of a set amount of annual activity from the provider (LHD or hospital). An illustration of this is shown in Figure 3 below:

Hospital						Activity					Activity					
	Activity					purchased					purchased					
	purchased by the					by NSW					by the LHD					
	Commonwealth	Pu	rchase	Sta	te Health ABF	MoH from	Pu	rchase	AB	C LHD ABF	from the	Pur	chase	Hos	pital	
	from NSW Health		Price		lget	ABC LHD	Price		Bu	dget	hospital	hospital Price		ABF	ABFBudget	
Х	65,000	\$	4,971	\$	323,115,000	65,000	\$	4,500	\$	292,500,000	65,000	\$	4,200	\$	273,000,000	
Y	50,000	\$	4,971	\$	248,550,000	50,000	\$	4,500	\$	225,000,000	50,000	\$	4,200	\$	210,000,000	
Z	40,000	\$	4,971	\$	198,840,000	40,000	\$	4,500	\$	180,000,000	40,000	\$	4,200	\$	168,000,000	
TOTAL ABF BU	DGET			\$	770,505,000				\$	697,500,000				\$	651,000,000	
Other services	s: Block funded															
Mental Health	ı			\$	20,000,000				\$	18,000,000				\$	17,000,000	
Community H	ealth			\$	10,000,000				\$	9,000,000				\$	8,000,000	
Population He	alth															
Total Budget (Total Budget (ABF and Block Funding) \$ 80			800,505,000				\$	724,500,000				\$	676,000,000		

FIGURE 3: BUDGET ILLUSTRATION A LOCAL HEALTH DISTRICT

Source: Developed for this research.

In this simplified example, the purchaser (Commonwealth/State) requires 65,000, 50,000 and 40,000 episodes of care from hospitals X, Y and Z, respectively. These services are purchased by the State at a National price (\$4,971) denoted in terms of a National Weighted Activity Unit (NWAU). The states then procure similar activity from the LHD at a lower price than the NWAU. Similar negotiations and pricing occur until services are delivered at the department level. The reduction in price is done to accommodate overheads for managing this process. The next section discusses the NSW determination process.

The determination of the NWAU is a cyclical process undertaken by the IHPA from costing data provided by all states through the National Hospital Cost Data Collection (NHCDC) and the Diagnosis Related Groups (DRG) system for disease classification. This process is summarised by Figure 4 below:



FIGURE 4: REPORTING PATIENT COSTS THROUGH THE NHCDC

Source: Developed and adapted for this research

The DRG classification will identify the proportion of an episode of care out of the NWAU. For example, the classification of knee reconstruction may have a weight of 0.5 with 1 being the baseline. The weight for the knee reconstruction may be adjusted by factors such as 0.05 for rural and remote locations. The total funding for these episodes of care is a product of the adjustment factors and the activity undertaken, as agreed between the provider and purchaser. Within an ABF environment, medical imaging services must be more efficient by providing the right study, safely, efficiently the first time.

When LHDs receive their funding allocations, under the budget process, they have a revenue budget as well. This is a portion of funding that hospitals are required to generate to support their state budget allocation. As such, departments with an income earning capacity, such as medical imaging through outpatient work, play a critical role in revenue generation. LHDs are expected to utilise their funding allocations in the fiscal year in which it was allocated. Failure to spend within a fiscal year may result in a lower budgetary allocation in succeeding years, i.e., "use it or lose it concept", unless specific exemption is sought from the funder.

At both the LHD and hospital levels, funding is generally allocated for specific purposes or to specific accounts, such as: salaries and wages; equipment under \$10,000; equipment over \$10,000 and so on. There is usually a capital limit for equipment over \$10,000 imposed for each LHD by the Ministry of Health which is dependent on the LHD size and budget. For SWSLHD this cap was \$3.5million in the 2019/2020 fiscal year. Capital expenditure exceeding this amount would require Ministry of Health approval. While this capital cap constrains the ability of LHDs to expeditiously replace equipment, it also provides a structure for prudent financial management within LHDs. Funding is usually linked to the attainment of certain goals. One of these key deliverables is Emergency Treatment Performance, which is discussed next.

1.2.7 Emergency Treatment Performance

Between 2011 and 2019 the Commonwealth of Australia implemented the concept of National Emergency Access Target (NEAT) and later Emergency Treatment Performance (ETP) as part of the Health Reform. Both initiatives were aimed at treatment or discharge of patients presenting to emergency departments (EDs) within four hours. According to the Emergency Care Institute (2014), NEAT was aimed at having 90% of patients presenting to the ED being discharged, admitted to hospital or transferred to another hospital for treatment within four hours. Key to this approach is that clinical decision making will override these targets which facilitate safe patient care. Research by several authors (Di Somma et al. 2015; Guttmann et al. 2011; Morley et al. 2018; Pines et al. 2011; Sun et al. 2013) suggested that poorer patient outcomes resulted from delays in patients accessing treatment. NEAT/ETP was a crucial driver in clinical practice change with key performance indicators (KPIs) used to measure hospital performance across public health systems. These KPIs are tied to health funding in addition to the ABF discussed earlier in section 1.2.6.
The shift in hospital funding through ABF and NEAT/EAP resulted in unprecedented pressure in EDs and consequently radiology departments. Radiology departments provide crucial diagnostic services to support clinician decision making, through services such as computed tomography (CT), X-rays, Fluoroscopy, Ultrasound and Magnetic Resonance Imaging (MRI). Additionally, due to the evolution of clinical practice over the past few years, CT demand has continued to grow, replacing a general x-ray as the modality of choice. Better Health Victoria (2015) defined Computed Tomography (CT) as a medical imaging procedure that uses x-rays and digital computer technology to create cross sectional images of all the body structures including bone and soft tissue. At a major NSW public hospital, 60% of the total CT demand is driven from the emergency department (Lin 2021). For such a hospital, undertaking at least 1,600 CT examinations every month, 960 of these studies are ED referrals. Moreover, consistent with the peak period for ED presentations of between 6pm and 10pm, CT activity in an afterhours setting contributed two thirds of the total CT usage at the hospital. As pointed out earlier in section 1.1, medical imaging provides the invaluable diagnostic tools which assist physician disease diagnosis and treatment in an emergency department. Radiologists and ED physicians have to collaborate to ensure that imaging tests are prioritised for EDs or there exists a process for prioritising and performing imaging studies, taking into account ETP as a performance and quality metric. As the setting of this research project is in medical imaging, it is important to review its history and origins as presented in the next section.

1.2.8 The Growth of Medical Imaging

Medical Imaging has grown from predominantly x-ray and fluoroscopy usage 25 to 30 years ago to a greater use of new technologies such as computerized axial tomography (CAT), or simply computed tomography CT, and magnetic

resonance imaging (MRI). These modalities utilise the tomography technology to acquire images and will now be discussed in turn.

CT technology, commencing in the 1970s (Shaw, AS & Dixon 2015), is based on the ability to acquire multiple slices of body images versus one plain offered by the conventional x-ray technology (Cox & Lynch 2015; Government of Western Australia 2021a). These images are then reconstructed using computer software and interpreted by specialist doctors called radiologists. Parallel to the development of CTs is MRI technology, discussed in the next paragraph.

Also, in the 1970s MRI developed as a modality. Unlike x-ray based CTs, MRI utilises magnets for their technology thereby reducing radiation exposure. While still an expensive technology, the superior resolution and diagnostic quality from this technology is increasingly making clinical use of MRI more widespread (Government of Western Australia 2021c). The use of technology in assessing the structure or organs needed to be supported by the study of organ functions, which will now be discussed.

While radiology focuses on the structure of organs, there was a need to understand organ function. Again, x-rays played a crucial role in the 1950s, in the establishment of nuclear medicine as an imaging technique. It was discovered that some radioactive compounds such as Technicium 99m could be combined with méthylène diphosphona and be absorbed by bones being invaded by tumours or cancerous cells (Government of Western Australia 2021b). This meant that cancers that typically spread to bones could be detected in the body though a nuclear bone scan and disease diagnosis achieved. Further technology advances have resulted in new technology such as positron emission tomography (PET). The premise of PET scanning is the use of chemical compounds such fluorine (F-18) that is incorporated into a glucose analogue called fluorodeoxyglucose (FDG)(Gaertner et al. 2015). In most cancers, glucose uptake is increased, and this increased uptake assists in the diagnosis of most cancers (Hoffmeier et al. 2014). PET and CT technology continues to evolve with new medical technology of PET CTs becoming increasingly available.

With the advancement of technology, many other imaging disciplines have emerged such as Interventional Radiology. The British Society of Interventional Radiology (2020), defined interventional radiology as a range of techniques relying on the use radiological image guidance (X-ray fluoroscopy, ultrasound, computed tomography [CT] or magnetic resonance imaging [MRI]) to precisely target therapy. These techniques reduce the need for open surgery which increases the risk of infection, reduction in hospital stay, lower costs, provide greater patient comfort and reduced risks (British Society of Interventional Radiology 2020).

While technology has continued to evolve, so has management practice in medical imaging departments. The increased technology costs, advancement in technology has resulted in continuous changes in the medical imaging landscape. Gone are the days when medical imaging departments used to be only support departments, but they are now viewed as being more integrated in the holistic approach to healthcare provision through multidisciplinary approaches within NSW and Australia generally (Chew, Cannon & O'Dwyer 2020; Phillips, Smith & Straus 2013; Sacks et al. 2019). Not only do medical imaging departments have to continuously manage their equipment pool to keep pace with technological advancement, they must also manage the changing general healthcare landscape. Like other specialists in medicine, medical imaging practitioners continue to specialise in specific areas such as CT; MRI; Ultrasound; Interventional Radiology (IR) and the Nuclear Medicine specialties such as PET and Bone Studies. Medical imaging practice has been confounded by state and federal policies which were introduced since 2008 such as the Diagnostic Imaging Accreditation Scheme (DIAS), and the Diagnostic Imaging Quality Practice Program (DIQPP)(Department of Health 2016, 2020c). These initiatives are intended to enforce service standards and quality through the incentivisation of investment in new equipment.

As technology has developed, anecdotally, patients and consumers are accessing information through the widespread usage of the internet or mobile technologies, medical imaging managers must increasingly manage more sophisticated customers who have a better understanding of the procedures they are undertaking (Kayser et al. 2015; Patel, Barker & Siminerio 2015). While customer experience was a consideration in the past, it is a major consideration for today's medical imaging manager. A significant number of customer complaints can adversely affect the perception of service provided by a medical imaging service. As such, the medical imaging manager is now required to proactively manage customer engagement processes to reduce reputational risk. Workforce considerations therefore play a key role in this process as discussed in the next paragraph.

The healthcare system generally is laden with an ageing workforce (Armstrong-Stassen et al. 2015; Callander et al. 2021; Gong & He 2019; Kagan & Melendez-Torres 2015; Lenthall et al. 2011) Medical imaging is not an exception. Managers are now required to proactively plan for succession (Howe et al. 2012), ensure staff maintain their skills and experience in accordance with professional bodies' accreditation requirements (Australian Health Practitioner Regulation Agency 2021)² and foster a good working environment for staff retention. In the NSW public health system, wages and remuneration conditions are governed by industry awards.

² Australian Health Practitioner Regulation Agency

1.2.9 Medical Imaging Departments in NSW

In NSW, there are two types of medical imaging practices, private and public ones. Private practices are independently owned and operated mostly by radiologists, while public imaging departments are associated with public hospitals in Australia. Unless practices are collocated within a private hospital where they perform inpatient studies, they predominantly undertake outpatient studies, compared to imaging departments in public hospitals. According to Royal Australia and New Zealand College of Radiology (2019)³, about 72% of diagnostic tests are undertaken by private practices with the remainder provided in public hospital departments. Anecdotal evidence suggests that public practices that manage inpatients, perform more complex and trauma type studies compared to outpatients who are usually mobile to seek services themselves. This research focuses on public imaging departments which are discussed next.

Most NSW public hospital radiology departments operate 24 hours a day, with a limited range of services out of business hours due to historical, resource rationalisation and other socio-economic and political considerations. These challenges are not unique to NSW LHDs, but also to most other public hospitals in Australia and internationally (Marino & Quattrone 2019; McCullough et al. 2020; Meng, Q et al. 2019). The demand for imaging in EDs significantly impacts on the effectiveness, timeliness, and efficiency of medical imaging departments. Coupled with changes in the hospital funding models in NSW as discussed in section 1.2.5, the case for changing work processes and workflows to improve efficiency, scarce resource management and managing emerging demands such as increasing community expectations cannot be more compelling (Bhattacharyya et al. 2019; Roberts et al. 2016).

³ Royal Australia and New Zealand College of Radiologists

Medical imaging departments like other health service departments, require a departure from the status quo to refocus healthcare delivery to different, more efficient ways in Australia and other jurisdictions. The complexity of healthcare does not make this task any easier, with multiple and usually non-mutually exclusive considerations at play. In attempting to refocus the provision of medical imaging services, effective leadership for change management is required. There does not seem to be an agreement in literature on what is strong leadership in healthcare, both generally and in medical imaging departments. Medical imaging departments for example, are faced with different situations requiring change through policy reforms. These changes include but are not limited to equipment procurement policies that recognise the capital-intensive nature of medical imaging. Existing policies, while encouraging investment in new equipment, do not go far enough or are incongruent with other budgetary processes utilised in the public health system, which include caps on capital expenditure as fully discussed in sections 1.2.6 and 1.2.11 of this chapter.

management implementation, the leadership and In change change management approach utilised is imperative in achieving success (Mehta, Maheshwari & Sharma 2014; Nging & Yazdanifard 2015). It is crucial for change management leaders to utilise an appropriate strategy that will result in the attainment of the intended objectives. The context within which public health organisation leaders must implement change, is complicated by several factors including state and federal government regulatory frameworks: incentives and disincentives for particular activities. For example, long waiting times in the emergency departments may necessitate hospitals offering GP clinics to colocate next to EDs. These clinics may then attend to patients in higher triage categories (that is patients who are less sick). The funding of these services may not be clearly defined. These factors result in health services preferring to undertake approaches aligned to funding models in the provision of healthcare,

which may not necessarily be advantageous to the patient but are more pragmatic for health services.

1.2.10 Equipment Procurement in Medical Imaging

Following the budget discussion earlier in this chapter, it is important to recognise that medical imaging is a capital-intensive service as it requires expensive state of the art equipment. Moreover, with technological advancement, equipment often requires replacement within a 10 to 15-year period or sooner due to usage, obsolescence of existing equipment and other considerations. Equipment replacement in Australia is also encouraged by the Medicare capital sensitivity rules. According to the Department of Health (2020a), diagnostic equipment is allocated life ages for Medicare benefits purposes, a process known as capital sensitivity. The intend of this process is to encourage practices to replace diagnostic equipment regularly to ensure that patients have access to high quality diagnostic imaging services. To achieve this, Medicare will stipulate how much it will pay for a diagnostic imaging test, using Medicare item numbers. If tests are done using equipment within the life age of the equipment (mostly up to 10 years), then Medicare will pay 100% of the schedule item number. However, if the equipment used is older than 10 years then the Medicare rebate item number is reduced to 50%. Public and private hospitals rely on this funding for their outpatients, as it is a revenue stream for them. Equipment replacement becomes a key consideration for public hospitals as another income stream. The budgetary process and a primer of equipment budgeting was provided in section 1.2.6 above. In this section the process for capital procurement is further explored.

Different departments within the hospital and LHD generally maintain assets lists and risk rate their assets. As part of the budget process in the months leading to a new fiscal year, departments are requested by the LHD finance directorate to

nominate their equipment requirements or provide their "wish lists". These are aggregated, with department representations made to the LHD executive and decisions made on the equipment to be procured at the LHD level. Once equipment procurement decisions are made, the department with approved purchases are notified and they go through the necessary procurement processes. Regularly, the Ministry of Health will have a centralised budget for high-cost equipment for example, CT scanners for medical imaging or Cardiac Catheter Labs. They will offer these funds to LHDs with caveats to "spend the funds within this financial year" or directly negotiate the procurement of these machines with equipment vendors. Where equipment procurement is done centrally, there is usually the risk that local requirements are rarely met, as consultation may not be at an appropriate level. Non-end users may make procurement decisions for inappropriate equipment type, or equipment that is non-compatible with an organisation's existing fleet could be bought. The former process ("spend it within the financial year"), also adds to the complex dynamic when funds are availed in May for a 30 June financial year end when the procurement process includes several activities, such equipment testing. In most cases, equipment is shipped from overseas where there is a 4-6-week turnaround time from the time a purchase order is generated, for the acquisition. These timeframes are far longer in the COVID19 pandemic due to supply constraints caused by lockdowns (Magableh 2021). NSW Health has implemented state-wide contracts where equipment can be procured, however, not all products and vendors may be on these contracts due to the compliance requirements to be on such panels.

Service managers in medical imaging reported that equipment leasing was previously considered and was prominent in healthcare in the early 2000s in NSW. LHDs are now reluctant to lease equipment for several reasons, such as lack of expertise and proactivity in managing leases. Additionally, leases impose a significant recurrent burden on the operating budgets of hospitals if not managed well, hence their unpopularity and restriction from the Ministry of Health in recent times. The patients as end users of equipment have some service expectations as they receive healthcare. These considerations are discussed in the next section.

1.2.11 Patient Considerations

The study was conducted in hospital settings and inherently most of the projects considered had some impact on patient welfare. The study was primarily investigating project leadership and management in medical imaging and their leadership behaviour, rather than project outcomes. Therefore, this research did not expressly examine patient opinions, as patients were not part of the study. However, a consideration of patient welfare was undertaken as part of this research, in so far as the projects undertaken by the LHD impacted on patient welfare. For example, change processes involving workflow reconfiguration while not directly targeting patient satisfaction, considered improving the ultimate patient experience as a goal of the change. These change projects undertaken were service improvement and quality improvement projects, viewed as low-risk projects within the SWSLHD and therefore requiring no ethics approval. However, these projects had relevant senior leadership approval and executive sponsorship. Section 1.2 provided a background to this research and discussed several issues impacting on medical imaging services including the Australian and NSW health context, funding arrangements; implications of health reforms in NSW; change drivers such as ABF and the growth of medical imaging. The next section discusses the research problem.

1.3 Research Problem, Propositions and Contributions

Section 1.2 above outlined the increasing challenges in a dynamic environment, faced by medical imaging department, which require addressing through research informed leadership processes. This research investigated how leadership in healthcare affects the outcomes of the change management process in medical imaging departments. It investigated the leadership styles that optimise change management processes and lead to successful change in public hospitals. In so doing, this research highlights the critical success factors for leading change, the strategies implemented for leading change and the key reasons for project success or failure. Through this research, learning leaders or organisations were explored, to understand whether learning leaders or organisations achieved better outcomes in subsequent change management processes, compared to those who did not leverage their learning from prior projects or processes.

As noted earlier, the literature does not seem to agree on what effective leadership is in medical imaging departments. More appropriate may the research of strong leadership as effective leadership. This research further investigated what effective leadership means for leaders in medical imaging. Medical imaging departments are faced with different situations requiring change through policy reforms for example. Some authors, (Glouberman & Zimmerman 2016; Ratnapalan & Lang 2020) argued that healthcare processes are heterogeneous in terms of range and variety of tasks. Medical imaging processes, however, can be both homogeneous and heterogeneous due to the customisation of service delivery or the emergence of personalised medicine therapies. For example, a radiology department may have a set process for undertaking CT studies. However, where there are patients with acute renal failure, requiring CTs with contrast, these patients must be accommodated, and special precautions undertaken to provide the CT safely (Government of Western Australia 2021a).

This research investigated the following questions:

- 1. What are the dominant leadership styles for change management approaches in medical imaging departments?
- 2. What strategies are used for successful change management in medical imaging?
- 3. How does the nature of leadership and change management processes in healthcare impact on the success of change management processes?
- 4. Why do some projects/processes fail or succeed?
- 5. How is the healthcare leader's leadership style influenced and reshaped by learned experience in prior change management processes?
- 6. How do "learning" leaders achieve better outcomes in successive change management processes?

1.4 Motivation for This Research

The researcher was employed within the LHD at the commencement of the research. When seeking literature to support evidence practice or ways of providing leadership, there was an observation that most of the leadership literature within the healthcare setting was broad, general and did not pertain to medical imaging. Medical imaging, like other diagnostic testing services, is quite capital intensive and change management when undertaken, needs to be effective and with minimal downtime. Coupled with the researcher's interest in leadership theory, this set the scene for the need to undertake medical imaging specific leadership research.

1.5 Justification for The Research

This research aims to identify those leadership styles that can be utilised in the health system to progress improvement and change projects. There is scant literature on leadership and change management within medical imaging in healthcare, as fully discussed in chapter 2 of this thesis. This thesis examined the leadership styles for leadership and change management processes in healthcare and seeks to add to the theory and scholarly literature on leadership styles in medical imaging departments. It will make these contributions by answering the research questions outlined above. The implications of this research include informing public medical imaging departments attempting to manage increasing demands for diagnostic imaging tests and effect how medical imaging departments will configure their workflows to achieve sustained performance improvement.

Strong leadership for change management was identified as being required (Ajayi 2002; Ajmal et al. 2012; Grant-Smith & Colley 2018). Chapter 2 will identify that literature does not seem to define what this effective leadership is, in healthcare generally and medical imaging departments specifically.

This research will provide guidance on the use of evidence informed change and leadership management processes in public hospitals with diverse stakeholders competing for scarce resources. The implications of this project will have an impact on the way local medical imaging departments undertake change management process both within Australia and internationally.

1.6 Research Outcomes

Informed by the answers to the research questions, this research study will lead to the following research outcomes, which will contribute to the outlined areas of leadership theory:

> Addition to the knowledge on the relationship between leadership and change management in healthcare and specifically medical imaging.

- ii. Whether certain types of leadership styles optimise outcomes to change management processes in healthcare and specifically medical imaging.
- iii. Assess the applicability of leadership theory to complex environments such as healthcare and specifically medical imaging departments.
- iv. Identify critical success factors for successfully leading for change management in healthcare organisations.

1.6.1 Additional Outcomes to Practice

This research was conducted by a practitioner within medical imaging as discussed in section 1.4. It is therefore important that the research considers the adoption of these additional implications on practice such as:

- i. Leadership practices that result in improving patient access to imaging services, patient safety, and quality.
- Interpretation of whether power and influence theory and other leadership styles are crucial to successful change management process in medical imaging.
- iii. Identification of the critical success factors of effectively leading for change in major public institutions.
- iv. Contribution to leadership and change management capacity for practice changes in medical imaging departments.

1.7 Methodology

1.7.1 Scope

This research is a mixed methods longitudinal study that sequentially examines the leadership style and change management behaviours of leaders through change management processes in medical imaging, over several successive projects or processes. The research investigated these through case studies of change management processes in medical imaging departments and draws intra organisation change management process comparisons and those led external to the organisation, state, country and internationally.

1.8 Outline of The Report

This thesis consists of six chapters as outlined in Figure 5. Figure 5 outlines the chapter on the right-hand side with the section heading contained in that chapter outlined on the left-hand side. A summary of each chapter is provided in the next paragraphs.

The thesis was introduced in this chapter, with a background of the research context provided. A brief synopsis was provided for the research problem together with the justification for this research. The research questions were introduced in this chapter, along with an overview of the contribution to literature together with the research outcomes was also undertaken in this Chapter. A broad methodology overview was also introduced in this chapter. The next chapter, Chapter 2 presents the research issues. In this chapter, a literature review will be provided. Chapter 3 discusses the research approach identifying the Methodology, data collection and analysis to be conducted, and the participants of the research. Chapter 3 also outlines the scope of this study. Chapter 4 discusses the results of the research both from a quantitative and qualitative perspective as this is a mixed methods study. The next chapter, Chapter 5 discusses the findings of the study and provides clarifications and further commentary on the previous chapter. This chapter also answers the research questions and discusses gaps in literature. Lastly, Chapter 6 provides a conclusion to the research and provide implications to practice. It also outlines the limitations of this research while also propose future research directions resulting from this study.

FIGURE 5: THESIS OUTLINE



Source: Developed for this research

1.8 Conclusion

This chapter has provided a background to the research study and highlighted the increasing pressure on the LHD. There is scarcity of resources which necessitates the need for the LHD to continually undergo change processes seeking value for money propositions, improved patient outcomes or to meet emerging patient expectations. An illustration of the funding processes for hospitals in NSW has been provided together with a discussion for emergent demands such as the Emergency Treatment Performance.

An overview of the research problem was provided in tandem with the research questions together with the research justification, expected outcomes and brief introduction of the methodology to be utilised. A research outline was also provided which provides details about the structure and what the reader should expect to find in this research. The next chapter provides a literature review of the literature relevant to this research.

Chapter 2 – Literature Review

2.1 Introduction

The previous chapter introduced this research, research questions and expected theoretical contribution. This chapter reviews leadership theory as it applies to this study. The literature review presented in this chapter, discusses the various leadership models as proposed by different scholars. Firstly, this chapter discusses the research problem. In this part of the discussion, the research questions will be introduced, building on the justification for the research as discussed in section 1.4 of the previous chapter. When the literature review is presented, the research gaps identified will be highlighted. The research questions immediately follow the relevant literature review, with a summary of the anticipated research outcomes provided at the end of the chapter. Secondly, the propositions or research issues are discussed and finally the anticipated scholarly contribution of this research outlined. This chapter also defines leadership, and various theories are reviewed with a focus on the leadership in healthcare, so that the later chapters can build on this review.

2.2 Literature Review

An extensive literature review was undertaken for this research project. More than 800 articles were reviewed and due to the breadth of the leadership literature, a structured approach was considered, and discussions were conducted under the following broad headings: Leadership Theory; Leadership Styles; Measurement tools for leadership style. Further discussion will also be conducted for change management and learning organisations.

2.3 Leadership

Kotter (2008) as referred to in Western (2019) and others (Bratton & Gold 2017; DuBrin 2015) defined leadership in terms of coping with change, responding to or creating chaos to some extent, with Stanley (2016) being more explicit in identifying the purpose of leadership as setting the vision; values and communicating these to their followers. Stanley (2016); Redknap et al. (2015) and Hutchinson and Jackson (2013) further suggested that leaders in this context, motivated, set direction and inspired a group for a shared goal. This results in leadership's functionality being; innovation, direction, communication and keeping people on track in difficult times. Similarly, Rittinger (2014) defined leadership as guiding and assisting followers to the right path as it is vital for goal achievement through employee engagement. Others such as Bui (2020); Goh (2020); Hsu and Lamb (2020); and Rebelo, Lourenço and Dimas (2019) suggested that leadership is associated with the stimulation and motivation of people to reach common goals. Northouse (2021) suggested that leadership is a process where an individual influences a group of individuals to achieve a common goal. The common theme running through these definitions is that leadership involves the influence of others to achieve set objectives. The way these objectives are attained varies for several reasons including the personality of the individuals or groups leading and that of the followers. Depending on one's school of thought, it is fair to say that leadership development is a lifelong process as discussed by Liu, Z et al. (2020) who suggested that leader development is an ongoing process across the entire lifespan. The Northouse definition of leadership is adopted in this study. This researcher is interested in healthcare leadership particularly leadership from medical imaging leaders, who are the subject of this research. While leadership has been studied in healthcare, for example in nursing, medicine, leadership in medical imaging departments remains an under researched area. This study seeks to contribute to this gap in literature.

The literature also discussed that for project or processes to be successful, effective leadership is required. Such literature, however, does not discuss what this entails. Strong leadership has been described in terms of the results it yields, for example, the International Institute of Directors and Managers (2020) identified the five main results that great leaders produce, which are: high percentage of top performers; higher employee retention; higher productivity; consistent growth and consistent results. However, some of these results are not applicable within public health settings, where for example organisational growth is not a key consideration. This study proposes that strong leadership is leadership that results in successful change management projects or processes. It is important for this research to explore what effective leadership entails, particularly what it means for medical imaging leaders, its characteristics and how it is enacted. Leadership should be viewed as distinct from management, which is discussed in the next section.

2.4 Management

Management is defined as the attainment of organisational goals in effective and efficient manner through planning, organising, leading and controlling organisational resources (Daft & Marcic 2016; Gomes et al. 2014; Grisaffe, VanMeter & Chonko 2016). Similarly, Burrell and Morgan (2017) and Fayol (2016) defined management as forecasting and planning, organising, commanding, co-ordinating and controlling. Both definitions are distinct from influencing others in the attainment of organisational goals. The former definition is the one adopted for this research. In medical imaging, participants who perform a management function and are recognised as managers by the LHD are the leaders who have been selected to participate in this research. Further discussion on the sample selection is provided in Chapter 3 following. These managers are the health service managers that are discussed in the next paragraphs.

Kippist (2012); McGivern et al. (2015) and Mangold et al. (2014) found that the role of the health service manager was increasingly complex, requiring professional, well-trained managers to have management skills and knowledge. This has led to the evolution of health service management increasingly becoming professional to the extent of having a professional college, the Australasian College of Health Service Managers (ACHSM), educating and upholding standards, like other colleges in healthcare, such as the Royal College of General Practitioners (RACGP). Brooke-Sumner et al. (2019) and Nyikuri et al. (2015) found that health service managers' ability to rapidly adapt to changing policy directives, landscapes and build relationships and staff resilience are crucial to weathering the complex challenges they face from healthcare systems and communities. Uvhagen et al. (2018) and Brooke-Sumner et al. (2019) also found that in top-down hierarchies, health service managers have a key role in determining task prioritisation and their perceptions of policies and innovations.

Within the LHD, health service managers are usually non-clinicians who are distinct from clinical leaders as fully discussed in section 2.5. below. Health service managers (or managers who are non – clinicians) usually possess healthcare leadership skills and other skills, that allow them to support or lead the provision of healthcare, in areas such as finance (resource allocation); corporate services type functions such as hotel services (cleaning; patient transport; porters and meal management), human resource management and other specialist areas (Knol, Janssen & Sol 2014; Lacity, Khan & Yan 2017). An understanding of health service management is important and relevant for this study as some of the leaders who are the subject of this research, are health service managers. Leadership within medical imaging departments is not only from health service managers as discussed above, but also from clinical leaders who are discussed in the next section.

2.5 Clinical Leadership and Management in Health Care

Edmonstone (2017) and Stanley et al. (2017) identified key skills that clinical leaders such as doctors; nurses; radiographers and other allied health professionals are recognised to possess. Such elements include clinical expertise; effective communication and interpersonal skills; empowerment, respect for others, team building; driving change, making care better, providing quality care; and vision. While Edmonstone (2017) and Stanley et al. (2017) suggest that some of these attributes are evident in clinical nurses, some need to be developed. For example, the demonstration and translation of their clinical values and beliefs into actions and functions of their management role (Démeh & Rosengren 2015; Doherty 2013; Erol et al. 2015; West et al. 2015). Such skills would fully harness the potential of frontline clinical nurses, radiographers, and other clinicians to be effective leaders, who are engaging and can make a positive and significant contribution to the quality of healthcare to their patients.

Kang et al. (2020); Maghsoudi, Cascon-Pereira and Beatriz Hernández Lara (2020); Okpala (2018) and VanVactor (2012), suggest a collaborative leadership model in healthcare organisations. VanVactor suggests the definition of collaboration management as involving multiple elements that can involve simple coordination, synergistic behaviour, mutual work processes and partnerships (VanVactor 2012). These are crucial tenets of change management projects or processes at major public hospitals in NSW. While the VanVactor (2012) study correctly identified key aspects in leadership processes that distinguish it from management, its limitations include the very high-level discussion that does not offer practical solutions in application for healthcare managers considering implementation of these approaches. For example, this study does not provide examples of where collaborative leadership has been practiced in healthcare, such as new multidisciplinary approaches in the provision of healthcare including physicians, nurses, pathology, radiographers, physiotherapists, and other health professionals. A review of clinical leadership literature identified that practitioner literature such as ltri and Lawson (2016); Kruskal et al. (2012); Matalon et al. (2018) & Shah et al. (2013) does not adequately discuss management or leadership theories or critical success factors for change management in medical imaging departments. In such practitioner literature, targeted primarily for medical imaging practitioners, gaps exist where there is no adequate discussion on leadership theories, its congruency and applicability with practice and change management processes in public hospitals. For example, while Kruskal et al. (2012) discussed LEAN⁴ methodology and the value addition proposition for patients, they did not adequately discuss leadership theory and linkage on leadership styles such as transformational leadership.

A subsection of clinical leadership is medical leadership, discussed in the following paragraphs.

2.5.1 Medical Leadership

An emerging type of leadership in healthcare is Medical Leadership. Andersson (2015); Sebastian et al. (2014); and Mintz and Stoller (2014) defined medical leadership as consisting of fully trained physicians occupying leadership roles relevant to the practice of medicine. Medical leadership can include resource management, decision making, recruitment and medical consulting as well as implementing changes and improvements in hospital and clinical settings. Sebastian et al. (2014) also suggested that medical leadership is evident at both strategic and operational levels with cases where doctors are executive leaders and are no longer delivering clinical care. Sebastian et al. (2014) also suggested that medical level where doctors possessing no formal management leadership position lead operational activities because of their clinical expertise and credibility. The medical imaging setting has medical leaders as defined by (Andersson 2015); Mintz and Stoller

⁴ A systematic process for reducing waste in a manufacturing or service process.

(2014) and Sebastian et al. (2014) and also some hybrid doctor-managers as discussed by (Coleman 2017) and Kippist, Hayes, and Fitzgerald (2012) who assume both a clinical and a managerial role to influence the strategic and operational levels.

Medical leadership often leads itself to medical dominance in clinical practice. Medical dominance is the use of selective resources to restrict and exclude other professions from gaining legitimacy in healthcare, through the three modes of subordination, limitation and exclusion (Kippist 2012). She also found that in Australia for example, only medical clinicians can refer patients to medical specialists in the public healthcare system, thus taking control of their own work and that of other professionals. This allows medical doctors to remain dominant over other professions. Such dominance thus transcends other areas of leadership with the capacity and ability to directly influence the leadership styles at the ward and department levels. It becomes apparent that medical leadership has an influence on the leadership styles of followers of those medical doctorled-departments. This research is set in a LHD where medical doctors lead many departments.

As discussed in Chapter 1, the structure of medical imaging departments is such that it can lead to medical dominance in the leadership of clinical departments in the LHD including medical imaging departments. Coupled with the high technological requirements of medical imaging departments, an engagement of the entire team is necessary, and this is often seen through the medical leaders who provide the driving force in the shape and operations of the service. An understanding of medical leadership is relevant to the leadership culture that may be observed in medical imaging departments.

2.6 Leadership Theories

Several leadership theories with relevance to this research exist and include Behavioural Theory (de Graaf 2019), Trait Theories (Ghasabeh, Soosay & Reaiche 2015), Contingency or Situational Theories (Mango 2018), Path-Goal Theories (Malik, Aziz & Hassan 2014), Power and Influence Theories (Olley 2021) and Leader-Member Exchange Theories (Orr & Cleveland-Innes 2015). These theories have been chosen and are discussed in this chapter as they contribute to the exploration of the leadership styles utilised by medical imaging leaders within NSW public hospitals. The theories will be discussed in the paragraphs that follow. In the discussion of these theories, the researcher will explain their additional relevance to this research.

2.6.1 Behavioural Theories

Behavioural theorists such as Lewin, Lippitt and White (1939) and McGregor (1960) focused on the actual behaviour of the leaders, for example whether a leader makes autocratic decisions without consulting others. Such leaders are called autocratic leaders whereas democratic leaders allow for team or follower participation in the decision-making process. Laissez-faire leaders on the other hand, have a "hands off" approach and do not interfere, allowing the team or followers to make most of the decisions. One such behaviour is consultative leadership. Kovach (2018) and Deshwal and Ashraf Ali (2020) suggested that these theories are premised on the assumption that an effective leader is one who exhibits behaviours that are most conducive to group productivity and psychosocial growth in any achievement environment. Robbins and Judge (2017) extended this theory, believing that people could be taught leadership through education and instruction.

As this research explores the nature of leadership within the healthcare setting, it is useful to study behavioural theories and how they impact on healthcare leadership and particularly medical imaging leaders. For example, if behavioural studies of other healthcare leaders have shown that certain traits lead to successful leadership practices, it is important to examine this for medical imaging leaders and assess such traits against leadership success or failure. An examination of the behavioural characteristics of healthcare leaders was undertaken in this research through the interview process and utilisation of the leadership style inventory (LSI) as part of the data collection process, fully described in Chapter 3 of this thesis.

2.6.2 Trait Theories

Trait theories are based on the premise that leaders have certain common characteristics or traits that make them effective leaders. Trait theorists, Stogdill (1948); Stogdill and Bass (1981); Mann, RD (1959) and Lord, De Vader and Alliger (1986), believed that leadership was an innate quality, someone was born with or without and was not a quality that could be learnt. This theory was further adopted by Northouse (2019) and further developed by Yukl (2019) who suggested that the big five broadly defined personality traits are:

- i. Surgeoncy in a leader requires extroversion, energy and a high activity level, and a need for power (assertiveness).
- ii. Dependability Conscientiousness in leaders requires dependability, personal integrity, and the need for achievement.
- iii. Agreeableness in leaders requires a need to be cheerful and optimistic, nurturing (sympathetic, helpful) and a need for affiliation.
- iv. Adjustment means that a leader must show emotional stability, selfesteem, and self-control and
- v. Intellectual outlook outlook in leaders indicates that they must be curious and inquisitive, open minded and learning oriented.

However other theorists have developed different views, for example, Robbins and Judge (2017) and Kovach (2018) discussed how the theory was able to forecast leadership outcomes but indicated trait theories' inconclusiveness in rationalising leader behaviours. As this study explores the nature of leadership in medical imaging departments, this theory was utilised in the exploration of the characteristics of healthcare leaders. The trait theoretical lens was utilised to evaluate the specific characteristics of medical imaging leadership. Trait theory also presupposes that leaders are inquisitive and learning oriented. This theory's relevance is also underscored by the need to understand the learning leader and organisation. For example, this research draws on the experience of medical imaging leaders as they undertake successive change management projects over time.

2.6.3 Contingency Theory

Contingency theory, also known as situational theory, is based on the premise that there is no right or wrong leadership style. The leadership style to be employed depends on the situation and the maturity of the individuals being led.

Leading proponents of contingency theory include Hersey, Blanchard and Natemeyer (1979), Fiedler (1967) and House and Mitchell (1975). Concurring with the last three authors, Utecht and Heier (1976) postulated that there are three main situational variables that that determine the favourability or unfavourability of a situation to a leader. These are: leader member relations, task structure and leader position power.

The first, is concerned with the amount of tension involved in the interpersonal relationship between the leader and the group members and are in part dependent on the leader's personality (Utecht & Heier 1976). The second refers to whether the job is done a certain way, such as a step-by-step procedure

(structured), or whether there is no particular way to do the task and the requirements are quite vague (unstructured) (Utecht & Heier 1976). The third one posits that position power is the degree to which the position itself enables the leader to get his group members to comply with and accept his/her direction and leadership (Utecht & Heier 1976).

Later literature from Cherry (2012) and Amanchukwu, Stanley and Ololube (2015) on contingency leadership states that success is dependent on several variables such as leadership style, qualities of followers and situational variables. Kovach (2018) and Deshwal and Ashraf Ali (2020) also discussed that effective leaders can adapt their leadership style based on the nature of the group, situation and objectives to be achieved. This is an important consideration in a complex environment such as healthcare. This view is supported by Ghazzawi, Shoughari and Osta (2017) who suggested that situational leadership demands that leaders vary their behaviour and leadership style according to their followers; commitment to the goals. For example, if followers in the medical imaging department were unable or unwilling to carry out some tasks, then clear and specific directions must be given by the leader. This makes contingency theory relevant for this study, which explores the critical success factors for leading change in medical imaging departments and healthcare.

Healthcare is hierarchical in nature both from an organisation wide, leadership perspective as well as from a professional level for clinicians. For example, the Chief Executive of a LHD has overall responsibility in the management of resources and outcomes for a population within a local governance area. Clinicians on the other hand play both a dual role as hybrid clinical managers and as clinicians only. Kippist (2012) found that the goal of senior management may be to minimise costs while clinicians are more concerned with having state of the art equipment for clinical practice. Contingency theory thus posits this study within the context of the healthcare leader and explores the medical imaging leader's leadership style within different contexts with different variables. As discussed in section 1.2, healthcare in Australia undergoes regular significant change hence this theory is relevant for this study in such a context.

2.6.4 Path-Goal Theory

House and Mitchell (1975) outlined the functions of leaders as; clarifying pathways for subordinates; removal of obstacles impeding subordinates from attaining goals and increasing rewards when subordinates achieve excellent results. To achieve organisational goals is one key requirement for enterprises. McKenna (2020) theorised that the effective leader is one who supports subordinates through paths which ultimately lead to organisationally desired and individually valued outcomes.

According to Boonyachai (2011), the Path-Goal theory was developed to describe the way that leaders encourage and support their followers in achieving the goals they have been set by making the path that they should take clear and easy. Robbins, Bergman and Stagg (2014) theorised that leaders must support their subordinates in achieving the goals of their organisations. The leaders in an organisation must all provide similar pathways to achieve goals of the organisations. Jones and George (2017) believed that the path goal model was a way in which leaders provided clarity to motivate subordinates to achieve specific results. They recognised four types of behaviours: directive, supportive, participative, and achievement oriented, which they believed, the success of each was hinged on the subordinate.

The path goal theory may be beneficial for leaders and followers to follow and was chosen to explore the healthcare leadership styles in medical imaging. In clinical practice, evidence informed protocols and practices are exercised. For example, in medical imaging, protocols exist on the process for conducting a CT Brain or a CT abdomen. Some studies require chemicals or reagents known as contrast and the patient preparation for those studies differs accordingly. Medical doctors lead the way in establishing these protocols and directing their teams to follow these prescribed pathways. Through this process, the path goal theory is seen in action and its understanding is crucial in understanding the leadership behaviour of medical imaging leaders.

2.6.5 Leader-Member-Exchange (LMX Theory)

Nie and Lämsä (2015) proposed that the LMX is routed in role theory, popularised by, Graen (1976); Dienesch and Liden (1986); Sparrowe and Liden (1997) and social exchange theory whose main proponents were, Wayne and Green (1993) and Erdogan, Kraimer and Liden (2002).

Graen and Uhl-Bien (1991) postulated that in LMX, there are three elements; the leader, follower and the exchange relationship. Yukl (2013b) further explained that the manager/subordinate relationship develops over time because of the role making processes and social exchange between them, with managers developing stronger relationships with some subordinates than others. Where mutual trust and loyalty exists, subordinates are provided with more responsibility with those trusted by the manager seen to be given more responsibility (Yukl 2013b). Interviews were utilised in this healthcare context research and sought to test the assertion by Yukl (2013a).

According to Chen et al. (2014), the more positional power an individual has, the more authority the individual presents, the more obedience employees will show. This view may be contrary to the highly specialised roles within a healthcare setting where medical dominance exists in the majority of clinical specialties (Germov 2019; Willis 2020). Additionally, Hsiung and Bolino (2018)

postulated that leaders have relatively high-quality exchange relationships with some employees and relatively low-quality relationships with others. This approach may not work in healthcare due to the interdependencies of sub specialities and the key role played by each profession.

Recent research by Gottfredson, Wright and Heaphy (2020) suggests that new and valid constructs and measures of leader follower relationships must be developed which involve; clearly conceptualising leader-follower relationship constructs; developing valid measures aligned with the construct and ensuring the research design surrounding the construct limits which affect endogeneity bias. Despite the emerging view discussed earlier, the LMX theory is relevant for this research as it is at the intersection of the leader- follower relationship and the exchange relationship. Okello and Gilson (2015) found that trusting relationships between frontline staff and management exerted a strong influence on staff retention, motivation and quality of care, with Scott, Mathews and Gilson (2012) further suggesting it may influence the adoption of innovations. Within healthcare, medical doctors' professional standing through their leadership role in clinical practice is quite high and is reinforced though a multidisciplinary approach to healthcare, where medical doctors take a leadership role and are quite influential (Edwards, N & Saltman 2017). This leadership role transcends to the ward and departmental levels. The LMX theory could explain why medical imaging doctors influence how their departments are run and consequently the outcomes of projects within those departments. The relevance of this theoretical lenses for this research cannot be understated.

2.6.6 Power and Influence Theories

Power and Influence theorists' approach is to view the way in which leaders use their power and influence to get things done. This then informs the leadership styles that emerge. French, Raven and Cartwright (1959) suggested that there are five sources of power: three positional – legitimate, reward and coercive and two personal sources – expert and referent.

Legitimate power comes from the belief that someone has the formal right to demand and expect compliance and obedience from others (Bolman & Deal 2017). Reward power is premised on the ability of someone's ability to compensate for compliance for example a manager's ability to influence a subordinate through rewarding desired behaviour in the form of a pay rise (Daft & Marcic 2016). Expert power is based on one's superior skill and knowledge, for example a doctor has expert power over a patient (Thompson, Zald & Scott 2017). Referent power on the other hand relates to the perception of attractiveness, worthiness and right to respect from others (Luhmann 2018). Finally, coercive power stems from the belief that someone can punish others for non-compliance, for example a line manager will usually hold coercive power over their subordinate. Raven's model suggested that a leader should use personal power while building their expert power.

Williams, Perillo and Brown (2015) found that the lack of authority to change practise is a barrier to the implementation of change. They also found that this barrier is linked to having a supportive employment culture that break down hierarchical workplace structures. If staff feel they can make changes by having their inputs and ideas accepted, they feel empowered to enact changes within an organisation. For example, in medical imaging, this translates to staff members feeling that they have the decision-making ability to effect positive patient outcome impacting changes.

The use of power and influence leads to several leadership styles including transactional leadership, transformational leadership, and authentic leadership (Cummings et al. 2018), with these discussed further in the sections that follow.

Conducting this study with an understanding of power and influence theory is relevant for this research. This is because leadership in healthcare is hierarchical, and this structure is derived from the distribution of power and influence within the different organisational levels. For example, health service managers control resources, budgets, and ultimately patient outcomes through the allocation of such resources. Clinicians through medical dominance, on the contrary exert their influence through clinical practice and their excellence in their clinical field (Braithwaite et al. 2017; Kippist 2012). An exploration of the extent to which their coercive power and influence is demonstrated, is crucial in how they lead in their departments hence the relevance of power and influence theory in this research.

Table 1 below summarises the discussed leadership theories outlining their characteristics.

Leadership Theory	Main proponents	Focus
Behavioural	Lewin, Lippitt, and White (1939) and McGregor (1960)	Actual behaviour of the leaders, for example whether a leader makes autocratic decisions without consulting others.
Trait	Stogdill (1948); Stogdill and Bass (1981); Mann (1959); Lord, De Vader, and Alliger (1986) and Kirkpatick and Locke (1991)	Leadership is an innate quality; one is born with or without it.
Contingency	Hersey, Blanchard, and Natemeyer (1979), Fiedler (1967) and House and Mitchell (1975). Utecht and Heier (1976)	There is no right or wrong leadership style.
Path-Goal	House and Mitchell (1975)	Encourage and support followers in achieving the goals by making the path that they should take clear and easy.
Power and Influence	French, Raven and Cartwright (1959)	How do leaders use power and influence to get things done?

TABLE 1 – LEADERSHIP THEORY SUMMARY

Leader-Member- Exchange (LMX)	Graen and Lid	(1976); en (1986)	Dienesch	Manager/subordinate relationship which influences
				trust and allocation of work.

Source: Developed for this research

Thus far, discussion and summarisation in Table 1 has occurred on the main leadership theories that exist and their main proponents. Discussion has also occurred regarding the relevance of these leadership theories in so far as they contribute to the leadership styles of medical imaging leaders. The next sections will discuss the leadership styles that have been developed from these leadership theories.

2.7 Leadership Styles

Leadership styles emanate from broad leadership theory groups. This section will discuss the following leadership styles as they continue to evolve and due to their relevance to this research; consultative leadership; transformational leadership; transactional leadership; authentic leadership; situational leadership; servant leadership; clinical leadership.

2.7.1 Consultative Leadership

Hornickel (2012) suggested that consultative leadership is a leadership style where the leader considers that team members have a lot of knowledge and experience, that if one garners, would provide one with information for decision making. He also suggested leaders do not lead in a vacuum and this is ever more important in complex organisations such as healthcare.

Additionally, Kissane-Lee et al. (2016) found that surgical trainees preferred consultative leadership during their training compared to the autocratic leadership they actually received in their training.

Consultative leadership is relevant for this study due to the nature of contemporary healthcare provision, through multidisciplinary teams. As such, it is important to consider consultative leadership as a style practiced by healthcare leaders. For example, Vainieri et al. (2019) and Oostra (2016) found that clinical engagement promotes a greater connection between the medical profession and the organisational environment.

2.7.2 Authentic Leadership

Authentic leadership theory is an emerging style inspired by Shakespearean Hamlet, "to thine own self be true". This has been embraced by practitioners such as Blake (2020); Ulrich (2020) and academics alike (Dinh et al. 2014; Gardner et al. 2011; Wang, D-S & Hsieh 2013). Authentic leadership can be described as a process that draws from both positive psychological capabilities and a highly developed organisational context, resulting in a greater self-awareness and self-regulated positive behaviours on leaders and associates, fostering positive self-development (Anderson, MH & Sun 2017; Haslam, Reicher & Platow 2020; Luthans, Youssef & Avolio 2015) and Walumbwa and others as cited in Schnackenberg and Tomlinson (2016) described authentic leadership as requiring four key components of: fostering greater self-awareness, an internalised moral perspective, balanced processing of information and relational transparency for leaders working with followers, to foster self-development.

Dinh et al. (2014) and Haslam, Reicher and Platow (2020) described authentic leadership as a pattern of transparent and ethical leadership behaviour encouraging openness and input between leaders and followers in the decisionmaking process. This process requires balanced consideration of followers input and negative and positive perspectives prior to making important decisions. In this model, the authentic leader thus sets high ethical and moral standards. Despite the relative infancy of authentic theory, compared to say LMX theory, there have been a handful of studies on authentic leadership in general industry and a very few studies in medical imaging. For example, Anitha (2014) and Leroy, Palanski and Simons (2012), linked work attitudes and outcomes, trust in management, organisational commitment (Peus et al. 2012) and nurses perception of their manager's authenticity, (Laschinger & Smith 2013; Wong & Spence-Laschinger 2013) who found a positive correlation between trust and authentic leadership.

Studies on authentic leadership have been in the Canadian healthcare context (Dwyer & Revell 2016; Wong & Spence-Laschinger 2013) with scant Australian studies found in the researcher's literature review. The applicability of authentic leadership within the Canadian healthcare system raises transferability and applicability considerations of this leadership style within an Australia and New South Wales context due to the federal or state-based jurisdiction similarities. For this reason, it is worth considering whether authentic leadership is evident in NSW notwithstanding the socio-political and economic differences between the former and later countries.

Authentic leadership is measured by the Authentic Leadership Questionnaire (ALQ) as developed by Walumbwa et al. (2008) (Avolio, Wernsing & Gardner 2018; Roof 2014). Like the Multifactor Leadership Questionnaire, introduced in section 2.7.4, the ALQ measures four domains of; self-awareness; relational transparency; internalised moral perspective; balanced processing. The aggregate score of these measures, decides whether a leader practices authentic leadership or not. The higher the score, the more the leader is viewed as practising authentic leadership. The discussion in the following paragraphs will focus on the four authentic leadership domains as discussed by Walumbwa et al. (2008).

TABLE 2 – AUTHENTIC LEADERSHIP DOMAINS

Domain	Discussion
Self-Awareness	The self-awareness domain refers to a leader who seeks feedback to improve interactions with others and accurately describes how others view his or her capabilities.
Relational Transparency	The relational transparency domain refers to a leader who says exactly what he or she means and is also willing to admit mistakes when they are made.
Internalised Moral Perspective	The internalised moral transparency domain refers to a leader who demonstrates beliefs that are consistent with actions and makes decisions based on his/her core beliefs.
Balanced Processing	The balanced processing domain refers to a leader who solicits views that challenge his or her deeply held positions and carefully listens to different points of view before coming to conclusions.

Source: Developed for this research

2.7.3 Transactional Leadership

Golla (2012) considered transactional leadership to be a style in which leaders promote employee self-interest through the promise of rewards in return for performance. This was supported by Sparh (2016) who said that transactional leadership focused on results, conforms to an existing organisational structure and utilises the organisation's reward or penalty system. She also suggested that this type of leadership is for maintaining routine with leaders possessing formal authority and positions within the organisation.

Sparh (2016) distinguished transactional leadership from transformational leadership by saying that the former is a reactive approach that features both positive and negative reinforcement while the latter is a proactive approach that emphasises motivation and inspiration. Further differences include that transactional leadership is a "telling" style while transformational leadership is a "selling" style. Zehir et al. (2011) also found that transactional leaders basically operate within an existing culture as opposed to those who try to change the culture to meet new expectations.
Bass and Avolio (1990) developed the tool used to measure transactional and transformational leadership. Transactional and transformational leadership were found to contain several dimensions. Smith, TD, Eldridge and DeJoy (2016) defined transactional leadership as a corrective approach with two dimensions of contingent reward and management by exception (both passive and active). The researcher discusses the transactional leadership dimensions of Contingent Rewards; Management by Exception (Active) and Management by Exception (Passive) in the paragraphs that follow as they are measured on a scale and as developed by Bass and Avolio (1990). The transformational leadership dimensions are tabulated in Table 3 below with a summary provided.

Dimension	Discussion	
Contingent Rewards	The Contingent reward dimension refers to leaders who discuss clearly, the responsibilities for specific tasks and projects, stating the objectives clarifying rewards and punishments and express satisfaction when they get the correct output.	
Management by Exception (Active)	The Management by Exception (Active) dimension refers to leaders who focus on careful monitoring of deviations, mistakes, and errors and on quick and thorough corrective measures, if appropriate. Leaders practicing an active management by exception use to keep track of mistakes, concentrate on errors and exceptions from the rule and on the treatment of these oversteps mistakes, concentrate on errors and exceptions from the rule and on the treatment of these oversteps.	
Management by Exception (Passive)	This refers to leadership behaviours typical where the leader takes corrective measure, but only after the fire has started. The behaviour is thus not proactive, but reactive and focused on punishment. Passive leaders do not; clarify misunderstandings, make their expectations clear; set clear objectives and performance standards for their followers. This style has very often a serious negative effect upon individual, group and organisational results. The rewards system proposed in this type of leadership is oxymoronic to the rewards system within the public health system where wages are set through employee bargaining and an awards system.	

TABLE 3 – TRANSFORMATIONAL LEADERSHIP DIMENSIONS

Source: Developed for this research

As this research seeks to ascertain the leadership styles practiced by medical imaging leaders in healthcare, it is relevant to understand how transactional and transformational leadership is measured. Tables 2 and 3 provide the domains and dimensions respectively that are measured in this research in determining the extent to which that leadership style is utilised.

2.7.4 Transformational Leadership

The adoption of transformational leadership has gained prominence in most healthcare settings particularly among the nursing workforce (Daly, J et al. 2015; Northouse 2021; Simons & Lomax 2021; Stanley & Stanley 2018). Burns (1978); Sadiq (2020) and Emere, Aigbavboa and Thwala (2018) defined transformational leadership as an approach that causes change in individuals and social systems through creating valuable and positive change in followers so that they becomes leaders themselves. Tajasom et al. (2015) also defined transformational leadership as helping followers to accomplish organisational goals and mission by working with them and through them.

Expanding on the work by Burns (1978), Stanley (2016) discussed the four themes of vision, communication, trust and self-knowledge as crucial in effecting transformational leadership. In a healthcare setting, transformational leadership empowers the clinician to make meaningful changes to their clinical practice. However, Northouse (2012) defined transformational leadership as a process focused on creating significant changes in followers transforming others and creating significant changes in people's lives and organisations. This more inclusive definition is adopted throughout this thesis.

In implementing transformational leadership, Yergler (2012); Meng, J and Berger (2019) and Vito (2020) argued that the creation of culture and its management are the only thing of real importance for leaders to do. Transformational leadership is important in healthcare due to the continuous state of changes. Previous solutions to issues are no longer adequate, due to the increasing influence and role of governments through policy and funding; the evolution of medicine and the increasing consumer or patient expectations from healthcare providers make it necessary for healthcare organisations to continuously evolve. For these reasons, transformational leadership in healthcare has been viewed as the most appropriate type of leadership (Antonakis & Day 2017; Lunenburg 2013; Sadiq 2020; Silva

2016; Uhl-Bien et al. 2014). Simons and Lomax (2021) reviewed transformational change in a general health care setting and discussed its importance in improving health care quality and safety. They highlighted the need for a bottom up, transformational leadership in the NHS⁵. The notion that all transforming or changing from the status quo continues to be an appealing concept to healthcare practitioners. For example, Kessler (2017), Simpson (2012) and Güven (2013) suggests that transformational leaders commit people to action and converts leaders into agents of change. In the healthcare environment of continuous change, transformational leadership may be the best approach for effective change management.

The National Institute for Health Research (2013), identified that there was a lack of empirical research focusing on organisational context. Their research suggests that most research tends to focus on descriptive and anecdotal research on leadership traits and personal characteristics, underplaying the importance of both internal and external organisational contexts. Also, a plethora of studies that has focused on transformational leadership in nursing exists such as Doody and Doody (2012); Drenkard (2013); Luzinski (2012) and Hutchinson and Jackson (2013). This literature review has identified a lack of longitudinal studies, transformational leadership in medical imaging. This research project will identify leadership styles used in change management processes in medical imaging settings.

The five dimensions of transformational leadership of; Idealised Influence; Idealised Behaviour; Inspirational Motivation; Intellectual Stimulation and Individualised Consideration as outlined by Bass and Avolio (1996) are discussed in section 3.5 in the next chapter where data collection and the methods utilised is fully discussed.

⁵ National Health Service

The MLQ is thus a tool that measures both Transactional; Transformational and Laissez Faire leadership styles and is fully described in section 3.5.6. Febres (2017); Meeks Jr (2020) and Salary (2019) found that the transformational leadership style was better than the transactional CR and the transactional MBEA leadership styles and results in a highly satisfied workforce.

2.7.5 Laissez - Fairer Leadership

This leadership style is premised by a group that makes decisions on its own, as the leader has no real authority, according to Boonyachai (2011); Liphadzi, Aigbavboa and Thwala (2017) and Emere, Aigbavboa and Thwala (2018). Yukl (2010) and Yahaya and Ebrahim (2016) suggested that Laissez-faire is the absence of effective leadership. The MLQ as discussed earlier is one tool that measures this type of leadership hence becomes relevant for this study. According to Bass and Avolio (2004) and Asrar-ul-Haq and Kuchinke (2016), this leader avoids decision making and responding to queries, with the typical behaviours being that the leader avoids involvement completely, even non responsiveness to threats and challenges arising. Such leaders are of the view that the challenges will go away on their own. According to Muenjohn and Armstrong (2015), subordinates of laissez -faire leaders are left to their own devices and have to seek assistance and supervision from alternative sources.

2.7.6 Servant Leadership

First articulated by Greenleaf (1977), servant leadership focuses on serving the highest need of others to help them achieve their goals. The servant leader leads due to their desire to serve people. According to DuBrin (2015), a servant leader is a moral leader whose purpose is accomplished by the degree to which their subordinates become autonomous. Also, according to Trastek, Hamilton and Niles (2014) and Hanse et al. (2016), servant leadership focuses on the leader's development through

awareness and self-knowledge, which enable the leader to understand their purpose, beliefs and characteristics.

The qualities and characteristics of servant leaders as identified by Spears and Lawrence (2016); Gandolfi and Stone (2018) and Parris and Peachey (2013) include; listening, empathy, healing, awareness, persuasion, conceptualisation, foresight, stewardship, commitment to the growth of people and community building. It may be suggested that healthcare workers in the profession to primarily serve other and possess most of these characteristics. However, the researcher is not asserting that healthcare leaders are servant leaders, in the absence of further research. This research project seeks to investigate the nature of that leadership hence the relevance of the servant leadership discussion in this chapter.

Hunter et al. (2013) and Samaibekova and Kocherbaeva (2019); suggested that people follow servant leaders because they trust them. However, this may not be the only reason why people follow servant leaders. Within the healthcare setting, followers do not necessarily choose their leaders. They work with leaders' positions with little choice regarding who they work with.

Following the discussion on Authentic; Consultative; Transactional; Transformational; Laissez Fair; Servant; Clinical and Medical leadership, Table 4 below summarises the key descriptors and dimensions.

Leadership Style	Dimensions	Descriptions
Authentic	Self-awareness; Relational	Process that draws from
	transparency; Internalised	both positive psychological
	moral perspective; Balanced	capabilities and a highly
	Processing	developed organisational
		context, resulting in a
		greater self-awareness and

TABLE 4 – LEADERSHIP STYLES KEY CHARACTERISTICS

		self-regulated positive behaviours on leaders and associates, fostering positive self-development
Consultative	Consultative Leadership	Leader consideration of team members' expertise for decision making.
Laissez-Faire	Laissez-Faire Leadership	Absent leadership, where a group that makes decisions on its own, as the leader avoids decision making.
Transactional	Contingent Rewards; Management by Exception (Active); Management by Exception (Passive)	Results oriented style premised on contingent reward for employees in return for performance.
Transformational	Idealised Influence (IIA); Idealised Behaviour (IIB); Inspirational Motivation (IIM); Intellectual Stimulation (IS); Individualised Consideration (IC)	Process focused on creating significant changes in followers transforming others and creating significant changes in people's lives and organisations.
Clinical	Clinical Leadership	Clinical leaders such as doctors and nurses possessing clinical expertise and other skills in the provision of healthcare.
Medical	Medical Leadership	Fully trained physicians occupying leadership roles relevant to the practice of medicine

Source: Developed for this research

2.7.7 Variety of Leadership Styles in Healthcare

Within the healthcare setting, there is a variety of leadership styles utilised by medical imaging leaders. All leadership styles depicted in Table 4 exist within healthcare or medical imaging. As discussed earlier, Cook et al. (2012); Daire and Gilson (2014) and Gilson (2016) found that health service managers and clinical managers are expected to manage facility resources, shape the organisational culture, influence individuals and relationships, motivate all cadres of staff,

sustain links with the various tiers of the health system and lead organisational change. To undertake these tasks, it is therefore important that they use a range of skills to meet the demand of complex health systems. In medical imaging for example, it would be expected that there is a range of skills and leadership styles utilised from the variety discussed in this chapter. Chapman, Johnson and Kilner (2014) found that there was a variety of styles used by medical leaders in conducting their roles in the NHS in the UK. Findings of this study will be compared to these previous findings to understand the similarities or differences between the UK and Australian healthcare systems. The next paragraph highlights the gaps in literature and motivation for this study, immediately followed by the research question.

The impetus created by the lack of research on leadership styles implemented in medical imaging departments motivates this research as it influences clinical leadership and general leadership practice. The preceding literature gaps drive the need for this research and leads to the first research question below. Further research questions for this thesis are preceded by relevant literature reviews. This research will explore the leadership styles applied in medical imaging departments and attempts to address the following questions 1 and 2:

2.8 Research Questions 1 and 2

- 1. What are the dominant leadership styles for change management approaches in medical imaging departments?
- 2. What strategies are used for successful change management in medical imaging?

2.9 Change Management

This research is not researching change management per se. However, it is investigating the leadership of change management projects or processes in medical

imaging departments. According to Thomson III et al. (2016), about 70% of organisational change management projects fail, and failed innovation projects are particularly common in medicine. With this level of project failure, it is critical to understand the critical success factors of implementing change in public health in medical imaging in New South Wales. Edwards, N and Saltman (2017) suggested that hospitals are hard organisations to change, particularly publicly owned and operated hospitals. Australian public health systems are heavily influenced by the socio-political and election cycles, with the major political parties using healthcare as a key battle ground. As such change fatigue may be argued as a major barrier to change among the dominant medical and nursing workforces (Aguirre, von Post & Alpern 2013; Stensaker & Meyer 2012). Additionally, there is an apathy to change in most countries with both managers and staff, despite constant calls for improvements in efficiency, quality, and responsiveness, that little will be different tomorrow or next year. This makes a compelling case for researching change management techniques that will allow leaders to implement change, even in the most challenging of circumstances.

The evolution of the Australian and NSW Health system discussed in section 1.2 also demonstrated that the public hospital system is in a continuous state of flux (NSW Health 1997). The review of change management literature suggests that there are a few distinct models of managing change such as the eight step process by Kotter (1995) and the three step change management model by Lewin (1951). These models are extensively used in most change management literature.; Aggarwal and Swanwick (2015); Doherty (2013) and Spurgeon et al. (2017), concluded that within hospitals and primary care practices, the engagement of clinicians was necessary to bring about changes, development and strengthening of clinical leadership. In healthcare, leaders often find themselves allocated to projects by necessity, either through legislation or being tapped on the shoulder. This is contrary to Rosenbaum, More and Steane (2018) and Tudor (2014) who suggested that when we choose

projects for ourselves, we are far more committed to the outcome (almost by a factor of five to one).

In healthcare, change management is also commonly undertaken following case study reviews. For example, the Bundaberg Hospital enquiry (Day, G & Casali 2015; Delaney 2015; Edwards, MS, Lawrence & Ashkanasy 2016) conducted a case study review following very adverse events and outcomes to patients. Their study was preemptive and aimed at informing decision makers to effect structural changes in the governance arrangements of their entities before the occurrence of adverse events. Their approach of process review at Bundaberg hospital is symptomatic of change management processes in healthcare (Casali & Day 2015; Wu et al. 2021).

While there is value in clinician literature in quality improvement and a practical application of initiatives, there is paucity of research on leadership and change management theory and its applicability in healthcare, particularly on leading change effectively and successfully in medical imaging departments. For example, ltri and Lawson (2016) proposed that situational leadership may be the more suitable approach for managing change, dictated by events on the ground. This view was not supported by Gareis (2013) and Silvius and Schipper (2015) who suggested that transformational leadership, which challenges the status quo, then involves planned change and implementation may be more appropriate. He also proposed radical repositioning which involves an organisational change management process in response to a crisis. It can be noted therefore that most practitioner literature or models remain quite general and remain vague about how planning, implementation and diffusion occur and its linkage to theory.

While some work has been undertaken on the value chain of business models in Radiology by Enzmann (2012), limited work has been undertaken by researchers on radiological processes and how they impact on patient flow, access and improved patient care. Mascia, Morandi and Cicchetti (2014) postulated that the increasing demand for better clinical safety outcomes, need for efficient economic resources, has encouraged substantial renovation of practices, models, and structures adopted within healthcare organisations. Seltzer et al. (1994) undertook a study to assess whether total quality management systems could be used for expediting reporting in a radiology department. They looked at a variety of interventions including reviewing workload, team reporting processes and additional resourcing to improve radiology reporting. They concluded that total quality management systems accelerated radiology reporting. However, the limitations of this study are that it only reviewed a component of patient access to CTs. Other components that might impact on reporting speed such as responsible ordering of CT tests by emergency departments or other clinicians were not addressed by this study.

For change to occur, there should be an enabling environment and an organisation's readiness to change. Brooke-Sumner et al. (2019); Cresswell and Sheikh (2013) and Williams, Perillo and Brown (2015) found that an organisation readiness to change comprised of an organisation's staff's joint commitment to implement a change, and staff members' shared belief in their capability to achieve the change. Aarons, Hurlburt and Horwitz (2011) found that attitudes to evidence-based practices have an influence in the adoption of innovations.

In implementing sustained change, Gilson (2016) suggested that pursuing small wins was important as these tended to build participants' confidence to develop an understanding of each other and tackle further long term change. For example, in medical imaging, this may be the development of a clinical protocol by a change team, that is a precursor to the procurement, installation and commissioning of a new MRI machine to conduct specific heart studies. The development of the protocol ensures that the team members understand what is required of them. When the bigger project is implemented, participants can visualise the end goal and can therefore fully participate in the intervening steps to bring the project to fruition. While change is necessary Thomson III et al. (2016) suggested that radiology lacks the necessary change management skills to effectively undertake change. This study contributes to the skills development process and adds to leadership literature in medical imaging departments.

Leadership and change management processes that address these burgeoning costs are not only crucial in Australia but also other western countries, managing increasing radiological and healthcare costs. Increasing diagnostic tests usage cannot be underestimated in modern day medicine. Ahn et al. (2014); Gilbert et al. (2012) and Morton and Korley (2012) researched the increased usage of CT and MRI imaging services in EDs in the US between 1998 and 2007 without a corresponding increase in life threatening conditions. However, work by Rodriguez et al. (2013) and Kea et al. (2013) indicates that less than 2% of CT tests were therapeutically inconsequential indicating appropriate CT head usage in their study. Brady, Cain and Johnston (2011) found that the usage of CT scans increased by two to three times between 1985 and 2005 in Australia. Considering the findings by Ahn et al. (2014); Gilbert et al. (2012) and Morton and Korley (2012), there is evidence of the increased healthcare costs due to increased CT usage. In an environment of financial austerity currently prevailing in NSW and Australia, taming healthcare costs is crucial as well as research on change management processes that will address these ballooning healthcare costs is of paramount importance. Australian Institute of Health and Welfare (2019) advised that the health expenditure to GDP increased by 1.2% in 2017/2018 to 10% the previous year, representing AUD185.4 billion spent on health goods and services. In the same period, hospitals spend an additional AUD3.5billion compared to the previous year. Additionally, a quarter of the taxation revenue nationwide goes into health expenditure, which by any measure is significant.

These gaps in the change management literature has resulted in this researcher asking the research questions 2 and 3;

2.10 Research Questions 3 and 4

- 3. How does the nature of leadership and change management processes in healthcare impact on the success of change management processes?
- 4. Why do some projects/processes fail or succeed?

2.11 Learning Organisations

Hsu and Lamb (2020); Schneper (2020) and Wen (2014), defined learning organisations as those that required employees skilled at creating, acquiring and knowledge transfer. Bevan et al. (2012) also defined organisational learning as the capacity or processes within an organization to maintain or improve performance based on experience. Edmondson (2018) and Tannenbaum and Cerasoli (2013) suggested that such individuals could assist firms to cultivate tolerance, foster open discussion and think holistically and systematically, which would allow organisations to be able to adapt to unpredictable environmental change leading to competitive advantage. Ayres (2012); Mafle Ferreira Duarte, Sahakian and Ferreira Neto (2021); Scharmer and Kaufer (2013); and Cousins (2018) supported this view and suggested that organisations and their leaders have to be learning organisations and leaders. Leaders are the change agents in the implementation of change processes and projects, and it is crucial that they utilise the lessons learnt from previous projects to improve future leadership capability.

The existing literature reviewed, discussed learning organisations in most environments other than healthcare. Donaldson (2018) and Franklin et al. (2014) suggested that the NHS did not have an adequate systematic process for adverse incident reporting. In healthcare, such systems are crucial to identify systemic shortcomings, trends and hopefully learn from mistakes of the past. The Bundaberg hospital case study (Day, G & Casali 2015; Edwards, MS, Lawrence & Ashkanasy 2016) is an example where other organisations can learn from the mistakes of others due to the tragic patient outcomes highlighted in the inquiry.

From this literature and discussion on learning organisations, this researcher defines "learning leaders" as those that create or acquire knowledge from past change management processes and apply such learning in future settings, to improve the outcomes of later change management processes. This study is a longitudinal study examining the change in leadership styles of medical imaging leaders over time as they lead change projects. A discussion on learning leaders and learning organisations is relevant as it allows an exploration of how learning affects future outcomes.

Despite the benefits identified in literature above, this researcher's literature review has identified scant medical imaging literature correlating the learning organisations and successive change management processes hence the research questions 5 and 6 that follows:

2.12 Research Question 5 and 6

- 5. How is the healthcare leader's leadership style influenced and reshaped by learned experience in prior change management processes?
- 6. How do "learning" leaders achieve better outcomes in successive change management processes?

2.13 Critical Success Factors

In change management, critical success factors usually exist, that impact on the success or failure of projects or processes. Javadin, Raei, Iravani, and Safari (2015) defined critical success factors as the limited number of areas in which satisfactory results will ensure successful competitive performance for the individual or organisational unit. For example, a critical success factor for an information technology change project may be the existence of a support team pre and post implementation to support end users with training and embedding knowledge. These authors also suggested that critical success factors are unique conditions indicating success. Mehregan et al. (2012) and Bhuasiri et al. (2012) viewed critical success factors as something, controllable, measurable and with mandatory implementation if organisations seek to achieve success in specific fields companies want to be successful in a specific field. For example, if consultation could be viewed as a critical success factor within a healthcare setting. As projects impact on several stakeholders, consultation may be viewed as mandatory to ensure that all constituent stakeholders provide feedback prior to project implementation. A lack of consultation would mean that the stakeholders not consulted may not sign off on a project, thereby negatively impacting on the project.

The discussion on critical success factors in this literature review is crucial in that it not only allows the reader to have a background in what they are, but also builds the reader's understanding as these factors are explored in later chapters of this study. As this research seeks to explore and understand those factors that influence project or process success, this discussion is relevant in providing definitions of critical success factors, provide the reader with an understanding of that they are, so that when they are clear to the reader as they read through the thesis.

2.14 Conclusion

This chapter discussed relevant leadership theories that underpin this research as Behavioural; Contingency; Trait, Leader Member Exchange; Path-Goal and Power and influence theories. The chapter further discussed the leadership styles examined in this research, namely, Authentic; Consultative; Laissez-Faire; Transactional; Transformational; Clinical and Medical Leadership together with the dimensions for these styles. Change management and Learning organisations literature was also explored. In this Chapter, literature review highlighting the gaps in existing literature was undertaken. These gaps were then linked to the research questions which were introduced in section 1.6 and provided justification of this research as discussed in section 1.5. Finally, this chapter outlined the research outcomes which included the contribution to theory and practice.

The next chapter, chapter 3 will review the methodology used for this research.

Chapter 3 - Methodology

3.1 Introduction

Chapter 2 reviewed the leadership theory applicable for this research study and the associated research question. It also explored the various leadership models. This chapter discusses the methodology that was utilised to conduct this study. According to Abdulghani et al. (2014) the methodology comprises of methods, processes and activities used to solve research problems and understand the issues being investigated or test some predetermined hypothesis. Gandhi (2011) suggested that, as such, researchers need to understand the assumptions underlying various research techniques and decide on appropriate techniques to problems.

Initially, an overview of the research methodology is provided in this chapter, briefly outlining how the research was conducted. In this overview, there is a discussion of the mixed methods approach undertaken in this study. Then the justification for the research methodology is provided before a summarised model of the series of steps involved in the study is outlined. Further detailed discussion is then presented, expounding on the step-by-step processes and stages of the research. This chapter also explores the multiple data collection stages undertaken together with the analysis that follows in chapter 4.

Chapter 3 also explores the data collection tools utilised, including the alternatives that were considered. A research methodology matrix is also developed in this chapter, integrating the research steps, data collection methods and research questions addressed. This matrix is presented in section 3.7 (Figure

8) of this chapter. Finally, ethical considerations of the research are discussed, including the ethical challenges faced and how they were managed.

3.2 Overview of Research Model

Kumar (2019), outlines how the research approach provides details and justifies how the research questions are answered and conclusions drawn. This research was designed to identify the nature of leadership and change management in medical imaging departments within the healthcare setting. It examined the leadership styles of medical imaging departments' leaders and their exhibited behaviours. This was undertaken through examination of successive change management processes over time, examining the most successful change management leadership styles and the behaviours leading to successful change management. Tools such as the leadership style inventory, KPls, interviews and questionnaires were used. Finally, a leadership matrix was produced and the extent to which leadership style impacted on successful change management determined. Further discussion was then undertaken on the findings, as outlined in the next chapter.

A longitudinal research study was used as an exploratory and descriptive tool, utilising a mixed methods approach. The mixed methods approach draws its rigour in corroborating multiple data sources such as interviews, data analysis and quantitative methods such as KPI computations and median turnaround times to provide a robust evidence base for drawing conclusions. Fetters and Molina-Azorin (2017) and Merriam and Grenier (2019) defined a mixed methods approach as one that uses two or more methods in undertaking a research study. Blank (2013) suggested that a mixed methods approach presents an alternative to qualitative and quantitative approaches by using the most appropriate methods to answer the research question.

3.2.1 Quantitative Approach

Kumar (2019) defined a quantitative study as one rooted in the philosophy of rationalisation, following a rigid, predetermined set of procedures to quantify variation in phenomenon, emphasising the measurement of variables and objectivity of the process.

In this study, an initial historical data collection was undertaken (utilising internal data sources) to ascertain baseline performance and service KPIs for the services to be investigated. These service KPIs were utilised to assess the success of processes. These KPIs formed one of a multiple bases from which success was assessed for that project or process.

3.2.2 Qualitative Approach

Kumar (2019) also defined qualitative research as being rooted in empiricism, that follows an open, flexible, unstructured approach to enquiry that aims to explore diversity rather than quantify. It emphasises description and feelings, perceptions, and experiences rather than their measurement.

This study is a longitudinal study that explored the leadership and change management behaviour of leaders over time in multiple change management processes in medical imaging. This research study was undertaken by analysing cycles of change within a three-year period, one retrospectively and two prospectively. Saldana, as cited in Farrall et al. (2015); supported this approach in identifying what makes a study longitudinal as the number of waves of data collected and not the length of time covered. To increase research transparency (Aguinis, Ramani & Alabduljader 2018), the research methodology, tools and rational for their use is included in the next sections.

3.3 Justification for The Paradigm and Methodology Design

The literature review undertaken by this researcher indicated that most previous studies were principally case studies in particular organisation settings. For example, a review of about 75 articles of transformational leadership in healthcare, revealed that the majority were case studies, such Daly, MP, Cleary and McCormack (2012); Marmor et al. (2013) and Nickel and Schmidt (2009) with scant studies utilising a mixed methods and longitudinal approach and found that these were regularly called for.

Wong and Spence-Laschinger (2013) recognised the limitations of their cross sectional study and suggested a longitudinal study design in how authentic leaders develop their relationship with their staff over time. Farrall et al. (2015) suggested that qualitative longitudinal research focuses on change over time and the participants' relationship with such change. They argued that it is also concerned with the exploration of individual lives as they develop. This view is supported by other authors (Hermanowicz 2013; Morrow & Crivello 2015; Patrick 2014; Thomson & McLeod 2015) who agreed that qualitative longitudinal research identifies and explores change over time.

According to Glinka and Hensel (2018), research design should specify how the researcher intends to ensure the credibility, reliability and transparency of their work (Aguinis, Ramani & Alabduljader 2018). Therefore, the mixed method approach in a longitudinal study methodology was chosen because it was the most appropriate form, to address the research questions. This methodology tracks leaders over time and assesses the success or lack thereof in change processes they lead. Figure 6 below provides a summary of the research methodology, which will now be described in detail.

FIGURE 6: SEQUENCE OF RESEARCH METHODOLOGY STEPS



Source: Developed for this research

3.4 Overall Research Procedures and Steps

Step 1. Ethics Approval

Before the research commenced ethical clearance was sought and granted from the SWSLHD Human Research Ethics Committee (HREC) (Appendix B). Following this consideration and approval by the SWSLHD HREC (HREC/15/LPOOL/360), this project was submitted for reciprocal approval under the Western Sydney University (WSU) HREC in accordance with candidature requirements for PhD thesis. Other ethical considerations were considered and are discussed in section 3.5 of this chapter.

Step 2. Participant Considerations

The research commenced with its population being that of all South Western Sydney LHD healthcare staff. Beaudry and Miller (2016) asserted that a general population is ascertained when the researcher is selecting subjects for the study. From this population, the researcher stratified the population into cardiovascular services; women and health services; cancer services and so on. From these strata, medical imaging was selected as a service, as this was the department of interest for this research. The medical imaging departments were then further stratified into; general staff, administration staff, medical staff, nursing staff, allied health staff and leadership staff. Stratification is the process of identification and separation of people or objects into specific groupings (Etikan & Bala 2017; Taherdoost 2016). It was deemed advantageous for this research in that it allowed the researcher to focus on the relevant participants for the research projects (Robinson 2014) as opposed to all medical imaging service staff, some of whom did not participate in any change processes or projects over the research period and would therefore have minimal contribution. From this stratification, medical imaging leaders were selected as they were the subject of this research.

All 31 medical imaging leaders and managers were invited to participate in the research study out of approximately 350 medical imaging staff across the LHD. These medical imaging leaders were identified utilising the researcher's experience and knowledge in medical imaging and healthcare generally, as those leaders who led other staff and change projects or processes. Of these 31 leaders, 21 agreed to participate in the research project. This purposeful invitation technique was particularly useful in that it allowed the researcher to specifically focus on the medical imaging departments and leaders who could contribute in this research.

Step 3. Population Data Review

The population data for the research project comprised of staff involved in change management projects in medical imaging departments within the LHD. For these projects, an initial identification of historical data from internal sources such as meeting minutes, strategic planning documents, the researcher's knowledge of research projects and document databases was utilised for the retrospective and prospective projects or processes to be researched. This informed the processes for steps 5 and 7 of the research methodology, as reflected in figure 6 above.

Step 4. Identification and Development of Data Collection Tools

Following a literature review, the researcher identified that several tools were required for data collection in this research. The data collection tool in this study included a combined Multifactor Leadership Questionnaire (MLQ)/Authentic Leadership Questionnaire MLQ/ALQ as a baseline leadership style inventory, and semi-structured interview questions. These tools are fully discussed in section 2.7, section 3.5 and steps 8, 9 and 10 below.

Step 5. Identification of Change Management Projects

The identification of medical imaging change management projects that occurred in the 18 months preceding the study, along with those to be undertaken throughout the next 24 months. This process was informed by the following:

- The Medical Imaging Services Strategic Plan to 2021 and the LHD's strategic plan to 2021.
- Meetings with the Medical Imaging Stream Director, Stream Clinical Manager and District PACS/RIS⁶ Manager.

⁶ Picture Archive Computer System/Radiology Information System

- Initial discussions with potential research participants to identify change management projects/processes they wanted to discuss.
- Pilot interviews, which highlighted several change management projects possible for this study.
- Review of Medical Imaging's performance KPIs to identify areas that required improvement and therefore would require a change management process.
- The researcher's experience and corporate knowledge of medical imaging services within the LHD.

From the defined and identified group of medical imaging projects or processes, stratification as discussed in step 2 above, was undertaken to establish a sample of mini projects or processes to inform the research. The selection criterion for projects was:

- i. These projects or processes took 18 months or less to complete.
- ii. Projects or processes could be either multi-staged or standalone.
- iii. Retrospective projects or processes included in the sample must have been completed within 18 months prior to the study being commenced.

As part of the interview meeting preparations of step 13 described below in this section, participants were asked to identify several projects that met the research brief and could be discussed and measured. These processes allowed participant identification of projects that the researcher was unaware of that were undertaken in the research timeframe.

Step 6. Establish a Participant Cohort

Following the participant selection process described in steps 2 and 3, a participant cohort was established. This process was undertaken through a stratified sampling technique, where health managers were divided into strata according to the service offering provided within the healthcare setting. This

technique was briefly discussed in step 2 above and is further discussed in section 3.4.4.

The actual participant cohort established consisted of 21 clinical, non- clinical and hybrid leaders in medical imaging who had led and undertaken change management processes or projects in the previous 18 months and were likely to undertake further change management projects or processes within 18 months from research commencement. These types of leaders are fully described in chapters 2 and 5 of this study. Of these 21 participants, four did not participate in the entire process and were excluded from the final study. This study followed through this cohort over the research period to answer the research questions.

Step 7. Obtain Informed Consent

Once the participants had expressed their willingness to participate in the research study, a formal consent form was sent to them for signature and return to the researcher (Appendix D). As part of this process and in fulfilment of the ethical requirements, the invitation to participants was sent out by an independent person from the research.

Step 8. Undertake Baseline Leadership Style Inventory (LSI)

A literature review (discussed in section 3.4.6 of this chapter) was undertaken to determine an appropriate leadership style inventory tool for use in this research. The combined MLQ/ALQ tool discussed in section 3.5.6, is also shown in Appendix E and Appendix F, and includes a results interpretation method and this was utilised to interpret the data from the Baseline LSI. This tool also enabled the research as it allowed the assessment or tracking of leadership style changes in participants as they led successive projects.

Upon determination that a leadership style inventory was to be utilised the writer purchased 50 licenses from Mind Garden the licensee of the MLQ/ALQ leadership style inventory tool. Invitations were sent to participants to their preferred email addresses via a website link requesting participants to undertake a registration and sign-up process. Following registrations, the joint tool was available to participants to complete online. During this process, a link was sent to the 17 research participants for a five-week period. The baseline LSI response rate was 10 out of 17 for the initial five-week period, the first due date. When a follow up reminder was sent out a couple of weeks later, a further 2 responses were received making a total of 12 in 17 responses. The remaining 5 respondents were discussing their first change management project. Following this further request for completion for the baseline LSI, no more participants completed the baseline LSI with the researcher deciding to continue with the next phase of data collection, which was the Round 1 interviews.

At the conclusion of the three interview rounds as discussed later in this section, participants were requested via an automated e-mail to complete the final LSI (step 14). The final LSI response rate was 10 out of 17 for the initial five-week period, the first due date. When a follow up reminder was sent out a couple of weeks later, a further four responses were received. At least 50% of the participants had completed their LSI, a decision was made to finalise this data collection round and devise a way to manage missing data. The researcher used average scores for each domain where there was missing data.

Step 9. Design of Interview Tools

Parallel to the baseline LSI process, the researcher designed an interview tool (Appendix J) as the main data collection tool for the initial interview round. For this purpose, a semi-structured interview was chosen as it allowed the

interviewer to obtain more in-depth understanding of the leadership processes employed and further information from participants that may not be evident in any project or process documentation. The researcher undertook face to face semi structured interviews for all the 17 participants of the study. While Steinberg (2015) argued that face to face interviews are more suited to illiterate respondents and the interviewer being able to control and manage the process as complex issues are identified, it is for this reason that face to face interviews were chosen over e-mail for example. Mangal and Mangal (2013) also argued that face to face interviews allow the researcher to clarify questions to participants to enable clarity of responses and reliability.

Semi-structured interviews (DeJonckheere & Vaughn 2019) were chosen as a data collection method for these reasons; they allowed the interviewee to elaborate on topics within a defined scope and allowed obtaining standardised quantifiable and more in-depth qualitative data (Adams 2015; Brown, A & Danaher 2019; DeJonckheere & Vaughn 2019). The design of this tool required to be in a manner that allowed the researcher to group participant responses to each question on a spreadsheet or other data analysis method. Common themes could then be explored and grouped together, and recurring concepts highlighted. This process was identified as necessary for results interpretation presented in chapter 5 and the drawing of conclusions, also in chapter 5.

Step 10. Conduct Pilot Interviews

Following the LSI process and interview tool development, the researcher randomly identified two participants from the cohort to undertake pilot interviews. This was an essential process to assess the interview questionnaire an effective tool for capturing research data to attempt to answer the research questions. Havig et al. (2020) and Hazzi and Maldaon (2015) suggested that a pilot

test enables the researcher to identify any shortcomings in the procedures to be used, to assess questions that might be left unanswered by respondents.

One interview was arranged in the participant's office, by mutual agreement, while the other one was held in the researcher's office, at the request of the participant.

Specific feedback was sought from the pilot participants regarding the interview format, length and question appropriateness and adequacy in answering the research questions. There were minor modifications required to the interview protocol identified during the pilot interviews. These modifications included the addition of some prompts from the researcher to assist in conducting the interviews, such as, "is this your first-time leading projects in medical imaging".

These two pilot interviews allowed the researcher to also assess the time required to complete each interview and adjust, if required, the research tool.

There were no major interview questionnaire adjustments identified from the pilot interviews and consequently, no major changes were required to the interview tools. Both interviews were scheduled for 40 minutes. However, they both concluded in about 25 minutes. It was noted that there was a 5-10 minute period required to explain the process to the participants including record keeping and access to the interview transcript. A 30–40 minute schedule for each subsequent interview was determined as an appropriate timeframe.

Step 11. Participant Interviews

The researcher contacted each participant and developed an interview schedule at mutually convenient times and locations. Each interview was set for 40minutes as identified in step 10 above. Each interview was transcribed by the researcher and a copy of the transcript provided to each interviewee to verify the accuracy and content of the interview. The researcher kept three password protected electronic copies of the interview transcripts in typed form. One was on the computer hard drive, another one on NVivo transcription and the final one on a USB drive. In addition, the researcher kept all audio copies of the interview until all interviews were fully transcribed. Initially interviews were undertaken using a voice recording device and a mobile phone device. After comparisons of the recording quality, it was noted during the pilot interviews that the mobile phone recordings had better voice clarity and were utilised for the remaining interviews.

Step 12. Review and Measure Project Success

At the conclusion of each interview, each project or process was evaluated for success or failure based on the pre project KPIs or success measures. Quantitative and qualitative KPIs were measured to determine project success or failure.

Step 13. Thematic Data Analysis

Once data collection had been completed, it required interpretation so that meaningful conclusions can be drawn. According to Clarke, V and Braun (2018) thematic analysis provides an accessible, systematic and rigorous approach to coding of the data and theme development. To obtain an understanding of the data received from the data collection processes, a thematic analysis was conducted through several methods for various data. All interviews were transferred from audio to verbatim written text as described by Kitchin and Lauriault (2015) and laquinto (2018), in the transcription process. This ensured that interview data could be stored and analysed systematically. NVivo 12 was utilised to code the interview data into themes as shown in Appendix M. This approach combines inductive and deductive coding, capturing specific themes while leaving flexibility for new themes to emerge particularly in relation to minority perspectives as suggested by Brooke-Sumner et al. (2019). Queries were

utilised in this electronic software to organise and draw some insights into the data. The data was initially captured on excel spreadsheets and some of it exported to NVivo. Where NVivo grouping and coding could be used, this was utilised, and manual excel analysis where NVivo could not be used. Broad headings were identified that appropriately described the characteristics being measured. These headings are described and discussed in Chapter 5.

Step 14. Undertake Final Leadership Style Inventory

A final leadership style inventory of the leaders at the conclusion of round three interviews or at the conclusion of three change management projects was undertaken to assess whether there were any changes to the baseline leadership style over time. This process was undertaken at the round 3 data collection stage. The ALQ/MLQ leadership questionnaire was utilised for this purpose as discussed in steps 4 and 8 above.

3.4 Research Methodology Further Considerations

3.4.1 Further Ethical Considerations

To satisfy the ethics requirements of this research, an independent person was utilised to send out participation invitations to identified potential participants. The establishment of the study cohort as discussed in Section 3.4, step 2, culminated from invitations send to 31 people identified as suitable participants. A standard email was sent for this purpose with a consent form and participant information sheet attached. Participants were requested to complete and return either to the sender or directly to the researcher. 21 of the 31 potential participants responded directly to the researcher, expressing their willingness to participate in the study. The researcher collated the contact details of all the participants who agreed to participate in the study so that these details could be utilised in undertaking the next steps.

3.4.2 Research Process and Multiple Data Collection Rounds

The research process comprised a series of three successive data collection rounds. According to Cooper (2016), data collection, recording and management commences when the project starts and ideas are conceptualised. Research data therefore had to be collected across a series of data collection rounds. The data collection rounds were approximately between three and five months apart. The steps described above were not applicable for all data collection rounds.

The data collection processes conducted for this research can be summarised as following a series of steps as depicted on Figure 7 below.



FIGURE 7: DATA COLLECTION PROCESSES

Source: Developed for this research

3.4.2.1 Round One Interviews

Participants were contacted via e-mail from the researcher to nominate a time and place for the interviews. These details were documented on a spreadsheet, outlining the contact date for each participant and additional information such as interview schedule times were added as they became available. This also became the tool utilised by the researcher for follow up with participants. If participants did not respond to the e-mail requesting availability, the researcher sent a follow up e-mail to secure an appointment.

Interviews were scheduled from November 2016 to May 2017, with 16 of the 17 respondents participating in the interview process. One participant was unable to participate in this initial interview round due to personal matters preventing her from attending work. The interview for this participant was scheduled following her return to work. The researcher considered that this did not impact on the research integrity because she could be interviewed at a later stage within the data collection period.

3.4.2.2 Round Two Data Collection

For round two data collection, it was not necessary to consider steps: 1, 2, 4, 6, 7, 8, 9 10 and 14 of Figure 6 above. All other processes were repeated for round 2 and round 3 data collection cycles. During the interviews for this data collection step, it was noted that three participants had role changes. These were participants 6, 7 and 10. Participants 6 and 7 were no longer in leadership roles but maintained their clinical roles. One participant no longer participated in change projects or processes and could not be interviewed for this interview round. The other participant could not be interviewed during round one interviews due to scheduling and workload challenges. However, this participant had already completed a baseline LSI. The researcher decided to conduct both round one and two interviews on the same day. Participant 10 was promoted during the data collection phase, which saw a significant change in responsibility.

It was also noted that participant 13's change project had not been concluded at the time of round 2 interviews. Through mutual agreement with this participant, the researcher agreed to defer round two interviewing for this participant until the third interview round when round 2 and 3 interviews could be completed. The researcher determined that they would not be any limitations to the research due to this accommodation and consequently did not adjust the research processes.

3.4.2.3 Round Three Data Collection

For round three data collection, it was also not necessary to consider steps 1,2,4,6,7,8,9,10 of Figure 6 above as these steps were predominantly relevant to the first round of data collection. However, step 14 was particularly relevant for this data collection round as it involved the use of the LSI to assess the leadership style at the conclusion of the data collection period. The researcher provided the 2 links for the MLQ and ALQ to participants after scheduling their round 3 interviews.

3.4.3 Researcher Ongoing Communication with Participants

The researcher communicated via e-mail and phone at least monthly with research participants informing them of the progress and timing of next steps. Ongoing communication with participants proved to a useful technique in keeping participant engagement high through the multiple data collection rounds.

3.4.4 Baseline Qualitative and Quantitative Project KPIs

The research project did not set specific KPIs to determine project success or failure. As part of the project/process selection, the researcher and the

participants determined the desired outcomes for each project. This was utilised as baseline KPIs for these projects both from a qualitative and quantitative perspective.

3.4.4.1 Quantitative Measures

Quantitative measures included KPIs utilised in medical imaging in specific processes. These KPIs were based on patient data available in the LHD's clinical information systems and other reporting systems.

Examples of these KPIs included:

- a. Examination Turnaround Times (TAT).
- b. Reporting TAT this KPI measures the time it took from image availability in PACS/RIS to report availability in eMR (refer to definitions).
- c. Number of examinations performed per hour.
- d. Number of examinations undertaken per annum.
- e. Patient transport TAT.

It was identified that the TAT measures utilised the median times and not average to account for skewness of data. Quantitative KPIs were chosen to corroborate qualitative views obtained from participant interviews on project or process success.

Medical Imaging KPIs were developed from existing data extracted from the PACS/RIS system, the Radiology Information System and Cerner, the patient administration system. The extracted data was de-identified for patient demographics, with only the examination details and other examination metrics excluding personal information utilised. Other activity data was sourced from the Ministry of Health and related agencies such as the Agency for Clinical Innovation and the Bureau for Health Information as required. Access and use of this data

was an ethical consideration of the research undertaken prior to obtaining approval from the organisation and undertaking the ethics approval process.

3.4.4.2 Qualitative Measures

Qualitative factors considered included successful changes in workflows that resulted in streamlined workflows the embracing of these changes by staff. Qualitative KPIs include patient satisfaction surveys and staff surveys pre- and post-change management process completion.

3.4.5 Sampling Techniques Utilised

The healthcare setting and background was discussed in section 1.2. it was noted that there are multiple groupings of workforce sub-specialities. Within this context, a combination of stratified and purposive sampling approaches was adopted.

3.4.5.1 Stratified Sampling

Stratified sampling (Iliyasu & Etikan 2021) is a probability sampling technique wherein the researcher divides the entire population into different subgroups or strata, then randomly selects the final subjects proportionally from the different strata (Etikan & Bala 2017; Robinson 2014).

For this research, the population was the various healthcare groups that were divided into strata, such as Cardiovascular Services; Women and Children Health; Cancer Services; which was within healthcare. In this case leaders from Medical Imaging were identified through a purposive/convenience sampling approach, as described below.

3.4.5.2 Convenience Sampling

Convenience sampling (Campbell et al. 2020) is a non-probability technique utilised to select a sample based on certain practical characteristics such as accessibility or availability at a particular time (Etikan, Musa & Alkassim 2016). This technique was utilised to select the participants for the pilot study due to their and the researcher's availability.

3.4.6 Leadership Style Inventory/Questionnaire

In the research design, consideration was made for leadership style inventory tools to measure the baseline leadership styles utilised by leaders in medical imaging departments. The next paragraphs review these considerations prior to summarising the methodology undertaken for this research.

This research sought to measure the leadership style changes of medical imaging leaders over time. To do that, a baseline leadership style needed to be identified, and then subsequently measured over time. Several approaches to measure leadership styles were available including commercial ones. It was not the intention of this project to review commercially available measures of leadership styles in turn. However, scholarly measures were reviewed in this section including the Multifactor Leadership Questionnaire (MLQ); Transformational Leadership questionnaire (TLQ); Empowering leadership questionnaire.

The sections that follow examined these questionnaires and assessed their fitness of purpose to this research.
3.4.6.1 Multifactor Leadership Questionnaire (MLQ)

This is a tool developed by Bass and Avolio (2004) as cited in Boamah and Tremblay (2019) which measures a full range of leadership styles and behaviours which comprise transformational styles. The transformational styles review attributes such as idealised influence; inspirational motivation; intellectual stimulation; individualised consideration. The other broad domain is transactional styles which measures contingent reward; management by exception (active) and management by exception (passive) and laissez-faire. Some researchers are not entirely convinced as to the best fit of this tool. Studies by Den Hartog et. al 1997 studies as cited by Hamstra (2014); Sarid (2016) and Wang, B et al. (2016) found a structure in MLQ comprising a transformational, transactional and laissez-fairer factors but not separate dimensions of transformational and transactional. Similarly, Berger et al. (2012); Stoller (2012) and Munir and Aboidullah (2018) found little evidence to justify the interpretation of the individual subscale scores. The dimensions of transformational are summarised in table 4 below, with a full example of the MLQ presented in Appendix E - Multifactor Leadership Questionnaire.

TABLE 5 – DIMENSIONS OF TRANSFORMATIONAL LEADERSHIP

Dimension	Discussion
Idealised Influence (Attributed IIA)	This refers to leaders who seek to build trust in their followers by inspiring power and pride in their followers, going beyond their own individual interests and focusing on group member interests, thus becoming role models for their followers.
Idealised Influence (Behaviour IIB)	Idealised Behaviours refers to Leaders who act with Integrity. These leaders manifest positive and highly valued behaviours such as dominance, consciousness, self-control, a high moral judgement, optimism and self-efficiency. Always considering the moral and ethical consequences for their actions, these leaders focus on a commonly shared sense of a vision or mission for the team or group.
Inspirational Motivation (IM)	The Inspirational Motivation scale identifies leaders who inspire others without the need for associates to identify with the leader. Inspirational leaders articulate in simple ways, shared goals and mutual understanding of what is right and important. These leaders provide visions of what is possible and how to attain such vision by enhancing meaning and promoting positive expectations of what needs to be done.
Intellectual Stimulation (IS)	This refers to transformational leaders supporting others to consider old challenges in new ways by encouraging them to question their own beliefs, assumptions, and values and when appropriate, also challenge those of the leader which may be outdated or inappropriate for resolving current challenges. As a result, they develop associates' capacity to resolve future challenges, unforeseen by the leader through creative and innovative means. A key measure of the leader's effectiveness is how capable their associates are, to operate in their leader's absence.
Individual Consideration (IC)	Individual consideration is another aspect of Transformational leadership which means that leaders understand and share in another's concerns and developmental needs and treating everyone uniquely. It also represents an attempt by the leaders to not only recognise their associates' current needs and expand and elevate those needs to maximise and develop their full potential. Transformational leaders also provide learning and growth opportunities for the individual.

Source: Developed for this research

As this tool was found to measure a full range of transformational leadership styles, it was utilised for this study. In this tool, the dimensions of transformational leadership are measured and compared to a comparative tool. If the scores meet are high, this is reflective of a leader practicing transformational leadership, while a lower score is indicative of its absence.

3.4.6.2 Transformational Leadership Questionnaire (TLQ)

The transformational leadership questionnaire was developed to measure transformational leadership in the UK public sector (Alvesson, Blom & Sveningsson 2017; Dugan 2017). The UK public sector setting appears to be prima facie like the research setting, the NSW public health setting. Further examination of the comparability however, suggested otherwise. The researcher, has a broad understanding of the NSW public system and setting as a result of working in the sector, found; the UK public health system appeared to be quite different from the former. For example, in the UK, the major public hospitals operate under the auspices of the NHS. This can be contrasted to the NSW public health system which is state based in a federal system in Australia. This creates differences across the states and there is no uniformity of policies and practices. Taking the employment conditions in NSW health for example, these are predominantly standardised employment award conditions. These governance arrangements impact practice and the applicability of local processes and measures making comparisons challenging between the two jurisdictions. The TLQ was therefore found not to be applicable to the NSW setting and was not utilised for this research.

3.4.6.3 Empowering Leadership Questionnaire (ELQ)

This tool was developed as a scale for measuring empowering leader behaviour in a series of 3 studies, initially consisting of in-depth interviews with external leaders and team members of three organisations, and then grouping these accordingly (Fong & Snape 2015; Foster 2019; Liu, Y 2015). This tool was administered to team members for two organisations leading to five criteria for measurement being developed. The third study sought to validate the scale in a sample of five organisations. The researchers (Arnold et al. 2000) who developed this tool also acknowledge its limitation as a tool still in development.

Another major criticism of this tool for this research is its specificity as a tool for measuring and assessing leader empowerment behaviours (Fong & Snape 2015). Consequently, this tool does not help in answering the research questions and was therefore not utilised for this research.

3.4.6.4 Authentic Leadership Questionnaire (ALQ)

The authentic leadership questionnaire (ALQ) is a multidimensional theorybased tool that measures the leaders' authentic leadership, measuring four dimensions: Self-awareness; Relational Transparency; Balanced Processing and Internalised Moral Perspective (Agote, Aramburu & Lines 2016; Erkutlu & Chafra 2013; Regan, Laschinger & Wong 2016). While this tool had been widely received in the authentic leadership field, the Neider and Schriesheim (2011) review of this tool raised some concerns with its validity. While acknowledging the comprehensive literature review and scale development of the of ALQ, Neider and Schriesheim (2011) carried out what they termed a more robust Authentic Leadership inventory (ALI) which they suggested used a more rigorous quantitative content validity assessment process.

While the ALQ could be a robust process for measuring or assessing leadership styles in medical imaging services, despite the concerns of the latter authors. The researcher's experience with clinical leaders in medical imaging departments suggests that hybrid managers tend to have strong convictions about their beliefs and strive to persuade others towards a different strategic or operational direction. However, as the leadership styles of medical imaging leaders were yet to be identified, it may not have been appropriate to measure the authentic leadership style alone if the leader's style is unknown. Therefore, this tool was utilised in conjunction with the MLQ. The ALQ utilised is presented in Appendix F of this thesis.

To conclude section 3.4.6, the LSI utilised in this research were the combined MLQ/ALQ. The TLQ and ELQ were not utilised. The next section discusses the data rationalisation process to ensure that missing data could be managed for this research.

3.4.7 Data Collection Rationalisation

At the conclusion of the five rounds of data collection, the researcher reviewed the participation rates for participants together with meaningful data from the participants that had been collected. As a result, the researcher decided to exclude all participant data for participants that had not been involved in at least three data collection rounds, that is, the preliminary LSI; the three data collection rounds and the final LSI. This reduced the initial study cohort from 21 participants to 17 participants. This exclusion criteria was used to allow the assessment of leadership change over time, whereas had few data points been utilised, this could potentially compromise the conclusions drawn from the data.

3.4.8 Discussion

While acknowledging the differences between their TLQ-LGV and the MLQ, Foster (2019) and Liu, Y (2015) suggested that there were common elements between these two approaches relating to critical thinking and challenging traditions.

The methods discussed above do not adequately cover emerging leadership styles such as medical leadership. Sebastian et al. (2014) defined medical

leadership as doctors with the ability to set direction, influence others and manage change. They further suggested that these leaders while clinically trained, operate at both strategic and operational levels. In this case they no longer provide clinical care but are in executive positions and; operational leaders, those with no formal management positions but lead operational activities utilising their clinical expertise and credibility, two sources of personal power that French, Raven and Cartwright (1959) and others (Bolman & Deal 2017; Burke 2017; Stewart & Shamdasani 2014; Wodak & Meyer 2015) discussed as expert and referent.

Chapman, Johnson and Kilner (2014); Berghout et al. (2017) and Kippist and Fitzgerald (2014) researched such leaders within the medical workforce. They suggested that contrary to the medical leadership model discussed above, doctor managers operated in a hybrid model as doctor-managers who assumed both clinical and managerial roles to influence at strategic and operational levels. In South Western Sydney LHD both types of leaders exist. MacCarrick (2014) recognised the challenges and complexity in exercising leadership in professional bureaucracies such as healthcare tasked with motivating clinical colleagues to attain organisational goals.

As the sample contains doctors in leadership positions, it was appropriate to utilise a mixed methods approach in the determination of the baseline leadership style inventory. This allowed for participants to self-report their leadership style and this would be corroborated by a leadership style inventory. A combined MLQ/ALQ emerged as the most appropriate tool to use in assessing leadership style and was therefore utilised for this project. Participants were also asked to self-assess their leadership style from a range of descriptors provided, which allowed the standardisation of data collection and responses. It was believed that this would corroborate and support the leadership style identified using the MLQ.

3.5 Summary of Tools Used for Data Collection

Several data collection tools were utilised as discussed in this chapter with a summary of these outlined in Table 6 below:

Leadership Style	ΤοοΙ	Research Method
Transformational	Multifactor Leadership Questionnaire	Quantitative
Transactional	Multifactor Leadership Questionnaire	Quantitative
Laissez Faire	Multifactor Leadership Questionnaire	Quantitative
Authentic	Authentic Leadership Questionnaire	Quantitative
Consultative	Questionnaire	Qualitative
Servant	Questionnaire	Qualitative
Situational	Questionnaire	Qualitative

TABLE 6 - SUMMARY OF TOOLS USED FOR DATA COLLECTION

Source: Developed for this research

3.6 Congruency of Methodology, Data Collection and Research Questions

The matrix below indicates the linkage between the research steps, research questions and the data collection tool in Figure 8 below as follows:

Figure 8: Research Methodology Matrix



Source: Developed for this research

3.7 Ethical considerations

Key ethical considerations for this research included:

3.7.1 The Researcher's Knowledge of Medical Imaging Departments

The researcher worked as Business Manager for medical imaging departments for SWSLHD for three years. In this time, the researcher was involved in several change management projects. This knowledge of medical imaging departments could potentially have impacted on the researcher's interpretation of the information provided by the research participants. This ethical risk was managed through the transcription of all interviews and verification by interviewees of each interview to ensure that their response had been captured accurately. As the researcher was not in a managerial role over participants and had no influence on their remuneration, there was no coercive power involved. Additionally, the researcher's supervisory panel was utilised for advice to manage this perception and the interpretation of data.

3.7.2 Selection Bias

There was the risk that the management team of the medical imaging departments would only select projects that had a successful outcome and not those that were considered unsuccessful. This would impact on the interpretation of the project results. The longitudinal nature of the study meant that through the review of a series of projects, the effect of selection bias would be negated.

3.8 Confounders

Potentially, the research could have been blighted by several confounders which included:

3.8.1 The Element of Luck on Project Success

Had the research not been a longitudinal study, correlation and the drawing of conclusions would have been difficult. The study design (longitudinal) allowed the study to be robust to eliminate the element of luck through a review of successive project cycles.

3.8.2 External Factors Impacting on Project Success

Healthcare is a dynamic environment. Projects commenced and being part of this research could be impacted by external environmental changes within the landscape. This risk was managed by selection of projects that were relatively

short to implement, that is projects with an overall duration of no longer than one year to complete.

3.9 Conclusion

The project methodology implemented for this research project was robust enough to facilitate the research objectives. The choice of the baseline leadership style questionnaire allowed for a thorough assessment of the participants' leadership style. This was then followed up by allowing the participants to provide a self-reported leadership style through the round one interview process described in section 3.4 above. The longitudinal nature of the study also allowed the research participants to reflect on their leadership styles in successive change projects as this was discussed following the completion of each change project cycle. At the conclusion of round three change cycles, all participants were afforded the opportunity to reflect again on their leadership styles and assess whether there had been any changes to their measured and self-reported baselines. Like the baseline measurement process, participants were requested to measure any changes in their leadership styles and a second layer of assessment undertaken through the final leadership style inventory questionnaire.

Data management tools such as Microsoft excel and NVivo were utilised in the classification and sorting of data. The researcher, who had no prior knowledge of NVivo, attended specific training to gain and develop knowledge in the use of this software. The researcher's prior knowledge of the medical imaging operations within the research settings allowed for an in-depth understanding of the processes and projects that were undertaken by the participants of this research project. The researcher could therefore pose follow up questions leading to richer insights into the change management processes described. The next chapter discusses the findings of this research.

Chapter 4 – Results

4.1 Introduction and Overview

Chapter 3 presented the research methodology utilised for this research including the data collection tools. The purpose of this research was to investigate how leadership in healthcare effects the outcomes of the change management process in medical imaging departments. To understand this phenomenon, the research project sought to answer the following research questions as outlined in section 1.3.

- 1. What are the dominant leadership styles for change management approaches in medical imaging departments?
- 2. What strategies are used for successful change management in medical imaging?
- 3. How does the nature of leadership and change management processes in healthcare impact on the success of change management processes?
- 4. Why do some projects/processes fail or succeed in medical imaging?
- 5. How is the healthcare leader's leadership style influenced and reshaped by learned experience in prior change management processes?
- 6. How do "learning" leaders achieve better outcomes in successive change management processes?

As earlier discussed in chapter 3 of this research, three interview rounds were undertaken as part of the research. During this study, a pre- and post-leadership style inventory questionnaire was offered to participants at the commencement and completion of the data collection process to all applicants. The previous chapter of this thesis also outlined how this study was undertaken through a mixed methods approach in a longitudinal study. In this chapter, results are presented and described using a mixture of quantitative data and qualitative data to clearly illustrate the findings. This chapter begins with a discussion of the sample characteristics. Then the findings from the pilot study are discussed, followed by the findings from the leadership style inventory. The quantitative findings are presented first and then the qualitative findings articulated. Fourthly, findings from the interview rounds are discussed before findings on the dominant leadership styles are presented. Then the reasons for project success or failure are presented prior to the strategies utilised for project success are outlined. Other strategies utilised for project success on how learned experiences impact on leadership behaviour. A chapter conclusion is then provided.

The results of this study are grouped into sections and will be presented based on the research questions for completeness and consistency. In the presentation of these results a mixture of formats will be utilised, which include tables, figures, and annotations, for comprehensiveness, to represent the findings of this research. Further analysis will be presented in chapter 5, of this research. However, this chapter's discussion will commence with some preliminary information that includes the following matters: sample selection results; sample and project demographics; pilot study; leadership style inventory; interview rounds and the final leadership style inventory.

4.2 Sample Results

4.2.1 Characteristics

As noted in section 3.4.2, this research had a 67.74% acceptance rate. This sample was a typical and ideal sample for the research study as it was representative of the population, with all leadership roles within medical imaging represented. The roles of the participants were also analysed with the results outlined in Figure 9 below.

Role Description	Population Size	Sample Size	Role Sample
			Percentage
Administration Manager	4	2	12%
Chief Radiographer	4	2	12%
Chief Technologist, Nuclear Medicine	2	1	6%
Dept			
Clinical Stream Manager	1	1	6%
Deputy Chief Radiographer	4	2	12%
Director Diagnostic Radiology	4	1	6%
Director Interventional Radiology	1	1	6%
Director Nuclear Medicine Department	2	1	6%
Director Nuclear Medicine	1	1	6%
Department/Clinical Stream Director			
Medical Physics Specialist	1	1	6%
PACS RIS Manager	1	1	6%
Senior Radiographer - Interventional	1	1	6%
Radiology			
Staff Specialist and Director of Training	5	2	12%
Total	31	17	100%

FIGURE 9: PARTICIPANT BY ROLE TYPE

Source: Developed for this research

Of the 17 participants in this research study, 8 (or 44%) were female compared to the 9 (or 56%) which were male. It was also noted that the medical imaging workforce was predominantly full-time staff, with 15 staff (or 83.33%) of staff engaged in full-time work within the service. Most of the participants were based at the largest hospital (Liverpool), with 13 (72%) participants, followed by Fairfield with 3 (17%) participants. Bankstown and Campbelltown hospital had a participant each (6%).

The occupations of the participants were also classified according to the functions they performed, and the results depicted in Figure 10 below.



FIGURE 10: PARTICIPANTS' OCCUPATIONS BY CLASSIFICATION

Source: Developed for this research

Three of the participants were drawn from the Business Support classification, which was administration type staff and non-clinicians, while the majority (13) came from the Clinician group and the remaining person being from Technical Support. Technical staff were those staff that provided technical expertise to the medical imaging departments that were non clinicians or business support staff.

While most of the participants, (10) worked at a single facility, four had a dual District wide role and facility role, one performed in a District wide role while two worked across several facilities within the same District.

4.2.2 Project Demographics

4.2.2.1 Project Duration

43 projects were researched in this study. These were part of the projects completed in medical imaging during the research period as depicted in Table 4 below. From the researcher's experience in medical imaging, most projects range in duration from about a month to less than a year. Additionally,

participants suggested that these projects are representative of the typical projects undertaken in medical imaging within South Western Sydney LHD.

Of these six projects were completed in less than three months, 11 projects took between three to six months to complete. Five projects were in the six to nine months category while seven were projects were completed within a duration of nine to 12months. Five projects took 12 to 18 months. Nine projects took between 18 months to 24 months to complete. This reflects a divergent range of projects completion time.

	Project Duration										
Project	> 3 3 < 6 6 < 9 9 < 12 12 < 18 18 < 24										
Round	months	months	months	months	months	months	Total				
Round 1	4	3	1	2	2	4	16				
Round 2	1	5	1	5	1	2	15				
Round 3	1	3	3		2	3	12				
Grand											
Total	6	11	5	7	5	9	43				

TABLE 7 – PROJECT DURATION BY ROUND

Source: Developed for this research

4.2.2.2 Stakeholder and Site Involvement

Of the 43 projects researched in this study, 30 projects involved multiple stakeholders with 13 involving internal stakeholders. This is reflected in Table 8 below. 29 projects involved a single hospital or site while 14 projects were undertaken involving multiple sites within the LHD.

TABLE 8 – PROJECTS AND STAKEHOLDERS INVOLVED

	Sta	keholders	;
Project Types	Location	Internal	Multiple
Business Process Redesign	Multi-site		3
	Single site	2	6
Clinical Procedure Redesign	Multi-site		1
	Single site	1	1
Clinical Process Redesign	Multi-site		1
	Single site	7	
Equipment Replacement	Single site		3
Infrastructure Project	Multi-site		1
	Single site		5
IT Project	Multi-site		3
New Process/System	Multi-site		5
	Single site	3	1
Grand Total		13	30

Source: Developed for this research

4.2.2.3 Project Types and Project Drivers

Table 9 below reflects that there were several key project drivers. The dominant projects were Business Process Redesign with 11 projects completed during the research period, followed by New Processes/Systems and Clinical Process Redesign with nine and eight projects, respectively. Infrastructure projects followed closely with six projects, with the following project types, with three projects each: Clinical procedure redesign; Equipment replacement; IT Projects.

TABLE 9 - PROJECT BY TYPE AND DRIVER

Project Driver	Project Types Undertaken										
	Business Process Redesign	Clinical Procedure Redesign	Clinical Process Redesign	Equipment Replacement	Infrastructure Project	IT Project	New Process/System	Grand Total			
Accreditation Requirements							1	1			
College Requirements	1						1	2			
Customer Feedback	1		2					3			
Infrastructure Upgrades	2			2	2	3		9			
Ministry requirement							3	3			
New Service	1				3		1	5			
Patient Safety	1	1						2			
Service Improvement	5	2	6	1	1		2	17			
Succession Planning							1	1			
Grand Total	11	3	8	3	6	3	9	43			

Source: Developed for this research

4.3 Pilot Study

Arising from the pilot study as discussed in section 3.3 (step 10), minor modifications were made to the interview tool as depicted in Appendix I. These modifications were prompts to interviewees at various questions, to assist less experienced leaders. For example, where the interviewee provided one-word answers, the researcher would probe further to solicit an explanation and therefore more meaningful information and insights in the change project. The interview tool was adopted for use in subsequent interviews. Accordingly, these interview results were incorporated as part of round 1 interviews.

4.4 Leadership Style Inventory (LSI)

Participants reported that the combined ALQ/MLQ tool took approximately 20 minutes to complete which they also reported as a reasonable length for this task. The researcher deemed this completion time to be appropriate, based on the participants' feedback during round one interviews, and no modifications were made to the tool. These responses were aggregated by participant and separated for results tabulation as shown in Table 10 below. Results of the LSI were separated between the ALQ and MLQ components were tabulated and are discussed separately, below, commencing with the ALQ.

4.4.1 Authentic Leadership Questionnaire

Table 10 below shows the baseline and results of the ALQ. Participants scored an average of 13 points for both the baseline and final leadership style inventories. This reflects a low level of authentic leadership across the LHD. Of all participants, Participant 6 scored the highest score of 15, which is still considered to be a low level of authentic leadership as measured by this selfrating scale. Between the baseline and final ALQ scores for each participant, there was no discernible difference in their scores with most recording a difference of two points or less. There was also no major difference in the median and modal scores for the participant cohort. Participants 12 and 17 fell outside of these benchmarks gaining three points and losing four points respectively between baseline and final scores. For participant 12, this points gain was because of the use of average scores for the final LSI, as the participant did not participate in this data collection round. For participant 17, a four points loss between the two data collection point was noteworthy. This participant was reasonably new in their role, having spent less than 12 months in this role, at the time of baseline data collection. The ALQ results as presented above are tabulated below and further discussed in chapter 5, section 5.5:

		E	Baseline /	ALQ	0	-	Final ALQ						
Participant	Transparency	Self Awareness	Ethical/ Moral	Balanced Processing	Total	Overall	Transparency	Self Awareness	Ethical/ Moral	Balanced Processing	Total	Overall	Difference over time
Participant 1	3.20	3.30	4.00	3.70	14	3.50	3.20	2.50	3.30	4.00	13	3.20	-1
Participant 2	3.20	1.80	3.50	3.00	12	2.90	3.06	3.14	3.27	3.34	13	3.20	1
Participant 3	3.17	3.13	3.25	3.28	13	3.21	3.06	3.14	3.27	3.34	13	3.20	0
Participant 4	2.80	4.00	3.30	3.00	13	3.30	3.00	4.00	3.50	3.70	14	3.50	1
Participant 5	3.00	3.30	3.80	3.00	13	3.30	2.40	3.00	3.00	2.70	11	2.80	-2
Participant 6	3.80	3.80	3.50	4.00	15	3.80	3.40	4.00	4.00	4.00	15	3.80	0
Participant 7	3.17	3.13	3.25	3.28	13	3.21	3.06	3.14	3.27	3.34	13	3.20	0
Participant 8	2.60	2.30	2.80	3.00	11	2.60	3.20	2.30	3.30	3.30	12	3.00	1
Participant 9	3.80	2.50	3.50	3.30	13	3.30	4.00	2.50	3.80	3.70	14	3.50	1
Participant 10	3.17	3.13	3.25	3.28	13	3.21	3.06	3.14	3.27	3.34	13	3.20	0
Participant 11	3.60	4.00	3.50	3.30	14	3.60	3.20	3.80	3.50	4.00	15	3.60	0
Participant 12	3.20	1.80	1.50	3.70	10	2.50	3.06	3.14	3.27	3.34	13	3.20	3
Participant 13	2.60	4.00	2.30	3.30	12	3.00	2.60	4.00	3.00	3.30	13	3.20	1
Participant 14	3.17	3.13	3.25	3.28	13	3.21	3.06	3.14	3.27	3.34	13	3.20	0
Participant 15	2.80	3.30	3.80	3.30	13	3.30	3.06	3.14	3.27	3.34	13	3.20	0
Participant 16	3.40	3.50	3.50	2.70	13	3.30	3.40	2.80	3.00	2.70	12	3.00	-1
Participant 17	3.17	3.13	3.25	3.28	13	3.21	2.20	2.50	2.30	2.00	9	2.30	-4
Median	3.17	3.13	3.30	3.28	13	3.21	3.06	3.14	3.27	3.34	13	3.20	0
Average	3.17	3.13	3.25	3.28	13	3.20	3.06	3.14	3.27	3.34	13	3.20	0
Mode	3.17	3.13	3.50	3.28	13	3.21	3.06	3.14	3.27	3.34	13	3.20	0

TABLE 10 – PARTICIPANT RESULTS FOR THE ALQ

Source: Developed for this research

4.4.2 Multifactor Leadership Questionnaire

The MLQ component of the LSI comprised of 45 questions as shown in Appendix D. To recap the discussion in chapter 3, the MLQ measures three types of leadership namely, Transformational; Transactional and Passive/Avoidant also referred to as Laissez Faire. The results of the completed MLQ were categorised as tabulated in Table 11 and discussed in the paragraphs that follow below, under their respective domains:

		Bas	seline MLQ			Final MLQ Differences Over						es Over Tim	ïme				
	Transformational	Tran	sactional	Passive	e/Avoidant	Transformational	Tra	nsactional	Passive/	Avoidant	Transformational	Transa	actional	Passive/A	voidant		
	Five I's of				Avoids							Rewards	Monitors Deviations				
	Transformatio	Rewards	Monitors	Fights	Involve	Five I's of	Rewards	Monitors	Fights	Avoids	Five I's of	Achieve	&	Fights	Avoids		
	nal	Achievement	Deviations &	Fires	ment	Transformational	Achievem	Deviations &	Fires	Involvem	Transformational	ment	Mistakes	Fires	Involvem		
Participant	Leadership	(CR)	Mistakes (MBEA)	(MBEP)	(LF)	Leadership	ent (CR)	Mistakes (MBEA)	(MBEP)	ent (LF)	Leadership	(CR)	(MBEA)	(MBEP)	ent (LF)		
Participant 1	3.30	3.00	2.80	0.30	-	3.50	3.80	1.80	0.80	1.30	0.20	0.80	- 1.00	0.50	1.30		
Participant 2	3.06	2.87	2.05	0.37	0.17	3.10	3.50	2.80	1.00	0.50	0.04	0.63	0.75	0.63	0.33		
Participant 3	3.06	2.87	2.05	0.37	0.17	3.20	2.96	2.51	0.58	0.41	0.14	0.09	0.46	0.21	0.24		
Participant 4	3.30	2.50	2.80	-	-	3.10	3.00	2.30	1.00	0.30	- 0.20	0.50	- 0.50	1.00	0.30		
Participant 5	2.90	2.50	0.50	0.30	0.30	2.90	2.80	1.50	0.30	-	-	0.30	1.00	-	- 0.30		
Participant 6	3.50	3.30	3.30	0.50	0.30	3.40	3.30	3.30	0.80	-	- 0.10	-	-	0.30	- 0.30		
Participant 7	3.06	2.87	2.05	0.37	0.17	3.20	2.96	2.51	0.58	0.41	0.14	0.09	0.46	0.21	0.24		
Participant 8	2.90	2.80	1.80	-	0.30	2.80	3.00	2.30	0.30	0.30	- 0.10	0.20	0.50	0.30	-		
Participant 9	3.10	2.30	3.00	1.30	-	3.00	1.50	3.30	0.80	-	- 0.10	- 0.80	0.30	- 0.50	-		
Participant 10	3.06	2.87	2.05	0.37	0.17	3.20	2.96	2.51	0.58	0.41	0.14	0.09	0.46	0.21	0.24		
Participant 11	3.50	3.50	3.00	0.30	-	3.80	3.80	2.30	1.00	-	0.30	0.30	- 0.70	0.70	-		
Participant 12	3.06	2.87	2.05	0.37	0.17	3.10	2.00	3.70	0.70	1.50	0.04	- 0.87	1.65	0.33	1.33		
Participant 13	3.00	3.50	1.50	0.30	-	3.10	2.80	1.30	0.30	1.00	0.10	- 0.70	- 0.20	-	1.00		
Participant 14	3.06	2.87	2.05	0.37	0.17	3.20	2.96	2.51	0.58	0.41	0.14	0.09	0.46	0.21	0.24		
Participant 15	3.06	2.87	2.05	0.37	0.17	3.40	3.50	3.00	-	-	0.34	0.63	0.95	- 0.37	- 0.17		
Participant 16	2.80	2.50	0.30	-	-	3.20	2.50	2.50	-	-	0.40	-	2.20	-	-		
Participant 17	2.30	2.80	1.50	0.70	0.80	3.20	2.96	2.51	0.58	0.41	0.90	0.16	1.01	- 0.12	- 0.39		
Benchmark	3.05	2.9	1.69	1.08	0.7	3.05	2.90	1.69	1.08	0.70							
Median	3.06	2.87	2.05	0.37	0.17	3.20	2.96	2.51	0.58	0.41	0.14	0.09	0.46	0.21	0.24		
Average	3.06	2.87	2.05	0.37	0.17	3.20	2.96	2.51	0.58	0.41	0.14	0.09	0.46	0.21	0.24		
Mode	3.06	2.87	2.05	0.37	0.17	3.20	2.96	2.51	0.58	-	0.14	0.09	0.46	0.21	- 0.17		
Median and Benchmark	0.01	- 0.03	0.36	- 0.71	- 0.53	0.15	0.06	0.82	- 0.50	- 0.29	0 14	0.09	0.46	0.21	0.24		

TABLE 11 – BASELINE AND FINAL MLQ RESULTS AND DIFFERENCES OVER TIME

Source: Developed for this Research

The results reflected in Table 11 above are interpreted using the tool shown in Appendix F and G and are discussed below under those three sub-headings.

4.4.2.1 Transformational Leadership

12 participants or 71% scored higher than the benchmark score of 3.05. The average score for transformational leadership for the cohort, was also 3.13 across the baseline and final MLQ results. This means that 29% of the participants were weaker in their Transformational Leadership on average, compared to the benchmark cohort.

A comparison of the Transformational leadership styles between the baseline MLQ and the final MLQ was undertaken, showing a very small change of 0.14 points overall in the leadership style of participants over time.

4.4.2.2 Transactional Leadership

According to Bass and Avolio (2004), transactional leadership is measured by the attributes of Rewards Achievement (CR) and Monitors Deviations & Mistakes (MBEA). 24% and 71% of participants scored higher than the benchmark score of 2.9 for the baseline and final MLQ for the CR attribute respectively. This means that reward was a motivator in medical imaging leaders. 76% and 88% of participants scored higher than the benchmark for the attribute of MBEA for the baseline and final LSI respectively. The overall average scores for transactional leadership were 2.46 against a 2.3 benchmark at baseline data collection. This score increased to 2.73 against the same benchmark for the final MLQ. Participants 12, 16 and 17 contributed to this increase with increases in scores of 1.65; 2.20 and 1.01 respectively, for the MBEA attribute. This means that these participants were active in monitoring mistakes or managing by exceptions. Participant 1's score on the other hand reduced by a score of 1, from baseline

data collection to final data collection. This corresponds to an increase in this participant in demonstrating a Laissez-Faire style in the later part of the research. Like the comparison of scoring for Transformational leadership above, participants in this research demonstrated stability progressively, for their transactional leadership style over time, with an average scoring difference of 0.27 points.

4.4.2.3 Laissez Faire or Passive/Avoidant

Only 6% of participants scored higher than the benchmark for the MBEP attribute at baseline, with this number reducing to 0% at final MLQ stage. One participant also scored higher than benchmark at baseline for the Avoids Involvement (LF) attribute, with this number increasing to 3 at the end of the research period indicating an increase in participants that tended to exhibit this attribute. The median score for the passive/avoidant leadership style at baseline was 0.27 against a benchmark of 0.89 while the score was almost double at 0.5 but well short of the same benchmark at final MLQ data collection.

These results mean that in general, participants did not exhibit a Passive/Avoidant style at any time during the research period. Of note, participant 4 showed a 1-point score increase for the MBEP attribute during the research period. However, this score was still below the benchmark of 1.08. Participants 1, 12 and 13 indicated an increase of at least 1 point at 1.3;1.33 and 1 respectively.

In conclusion, the quantitative research presented above identifies that there was a discernible difference in the participant leadership styles across the research period with a difference with a range of 0.09 to 0.46 points. For transformational leadership, a median score of 3.06, slightly higher than benchmark indicated that generally participants identified as utilising style. Examining transactional leadership showed that the median score was 2.73, slightly higher than the 2.3 benchmark. This indicates that participants sometimes used the transactional leadership style. Finally, participants scored generally low scores of 0.5 at the end of the research period indicating that they were not leaders who were inclined towards the passive/avoidant style.

The next section presents the qualitative findings of the interviews, with these findings grouped into coded themes and discussed in chapter 5.

4.5 Round One Interviews

During this round, there was a total of 17 participants who participated in the study. These participants are representative of all the roles within medical imaging and form a typical sample of medical imaging leaders in healthcare. 16 (94%) were involved in the interview process with one participant unavailable for interviews as reflected in Table 12. Of the 16 participants who completed the baseline LSI, 7 (44%) were female and 9 (56%) were males. 13 out of 16 (82%) of the leaders within this setting identified as clinical leaders; 75% as hybrid leaders and 50% as strategic leaders.

Similarly, to the participant characteristics, the number and type of projects undertaken, were typical projects undertaken in medical imaging in public hospitals as reported by participants. The number of projects were variable in length as: under 3 months (2 or 17%); 3 to 6 months (5 or 42%); 6 to 12 months (2 or 17%); and over 12 months (3 or 25%). The projects undertaken were evenly spread between Infrastructure/Equipment commissioning (5 or 42%); introduction of new processes (5 or 42%) and clinical/process redesign (5 or 42%). 58% (8) of the projects or processes involved multiple stakeholders with only 33% projects being completed as internal to the department only. 33% of the projects

were multisite projects and 67% were department specific change projects/ processes.

In undertaking these projects, the communication styles varied from project to project. All projects utilised at least two forms of communication with stakeholders. 100% of the projects required meetings to be held as a form of communication. Additionally, e-mail was the dominant mode of communication with 83 % of leaders utilising this mode of communication. 17% of the projects utilised the top down and bottom-up approaches for messages to get to the intended audiences. 83% of the leaders also utilised a collaborative communication style.

An interview transcript was prepared for each interview in Microsoft word and exported to NVivo where it was stored under the participant's name for further analysis.

4.6 Round Two Interviews

During round 1 interviews, participants were requested to identify future projects to be discussed in round 2 interviewing.

A similar process to the one adopted in round one was utilised to request for interview availabilities as described in section 4.5. Interviews for the second interview round were scheduled from January 2018 to July 2018. A total of 15 (or 88%) out of 17 participants participated in the interview process as indicated on Table 12 below. Participants were contacted through e-mail requesting them to nominate their availabilities for the research project interviews. If responses were not received within a week, a follow up e-mail was sent to the participant. During this interview round, two participants did not respond to requests for interviews,

despite several follow up requests to interview. The final two participants could not be interviewed due to their projects not being completed at the time of the interviews. The researcher and these participants agreed to schedule these interviews concurrently with the third interview round when it was anticipated that the change projects would be complete.

Like the process adopted in round one interviews as described in section 3.4.2.1. An interview transcript was utilised (Appendix K) as a case study in assessing an appropriate method of presentation of results. Categorisation of self-leadership styles was undertaken with the results reflected in Table 13 in section 4.8.

4.7 Round Three Interviews

Like the project/process identification process described in section 4.5 above, participants were requested to identify a change management project for discussion in this interview round.

A process consistent with that utilised in rounds one and two was utilised to request for interview availabilities as described in sections 4.5 and 4.6 Participants were requested to nominate a time for an interview with the researcher in mid-August 2018. The interview schedule considered the commitments and circumstances of the leaders in the LHD.

The interview window was set for August 2018 to October 2018. 12 participants (71%) had their interviews scheduled for August 2018 with the remaining 19% (or five participants) scheduled for the other two months. During round three data collection, the researcher also ascertained whether participants had received the final MLQ/ALQ links as described in section 3.4 prior to the interview commencing. All participants confirmed having received the link to the final MLQ/ALQ and therefore no resending was required.

Following each interview, a transcript was prepared, and this was utilised for the coding in NVivo. Data collection across the five processes was completed with the results summarised in Table 12 below.

	Partici	ocesses			
Participant	1st LSI	Round 1	Round 2	Round 3	Final LSI
Participant 1	Yes	Yes	Yes	Yes	Yes
Participant 2	Yes	Yes	Yes	No	No
Participant 3	No	Yes	Yes	Yes	No
Participant 4	Yes	No	Yes	Yes	Yes
Participant 5	Yes	Yes	Yes	Yes	Yes
Participant 6	Yes	Yes	Yes	Yes	Yes
Participant 7	No	Yes	No	Yes	No
Participant 8	Yes	Yes	Yes	Yes	Yes
Participant 9	Yes	Yes	Yes	Yes	Yes
Participant 10	No	Yes	Yes	No	No
Participant 11	Yes	Yes	Yes	Yes	Yes
Participant 12	Yes	Yes	Yes	No	No
Participant 13	Yes	Yes	Yes	No	Yes
Participant 14	No	Yes	Yes	No	No
Participant 15	Yes	Yes	Yes	Yes	No
Participant 16	Yes	Yes	No	Yes	Yes
Participant 17	No	Yes	Yes	Yes	Yes

 TABLE 12 – PARTICIPATION IN DATA COLLECTION PROCESSES

Source: Developed for this research

4.8 Dominant Leadership Styles in the Local Health District

Across the three interview rounds, 47 interviews where cumulatively conducted via the 17 participants who participated in this study. Of these participant projects/processes, an overwhelming majority of 45 out of 47 (96%) reported that they successfully implemented change, with 2 (4%) projects or processes being unsuccessful. These two projects were identified as projects undertaken by a participant that identified as utilising the consultative leadership style. We will examine the dominant leadership styles practiced by leaders in this research in the succeeding paragraphs.

During the participant interviews, participants were asked to self-report their leadership style and these responses were aggregated with regards to their self-reported leadership styles and tabulated as shown in Table 13 below. The research study determined that the dominant leadership style found was Consultative across the three interview rounds with 44%; 67% and 56% of participants identified as utilising this style in rounds one; two and three respectively. This differs from that reported in the existing literature, as is discussed in section 5.4.

It can be noted that consistently across the three interview rounds, a consultative leadership style was preferred by medical imaging leaders, followed by a situational leadership style, with other leadership styles featuring as described above. Participant 13 underscored the need for consultation and said:

"I think consultation is very important. From my days as being the Manager here, consultation has been a key thing. I consult my staff. Any changes that I need to do, without consultation, I never do".

For participant 2 consultative leadership means:

"If there is anything new to be introduced, we always talk to Senior (Radiographers) people first or as Seniors meeting with them. If there is something that's urgent, you have to consult, you have to ask them what do you think? That is the normal process".

Participant 15 views his leadership style as consultative. He said,

"I usually consult others. What other things have I missed out on or what other things should be considered before a project takes place? Are we doing it the right way? Is there any loopholes within the system? We look at it and spend at least two weeks to see what other things will come up. What are we going to do if this happens or if that fails?"

Participant 17, highlighted the advantages of consultative leadership when she said,

"I would say I am leader who gets input from staff as well. Sometimes you never know, you get staff who have better ideas. You take that on board, you think about it. I also delegate things and also take on board their ideas as well".

A further 31% in round 1, 27% in round two and 38% of participants in round three indicated that they were situational leaders.

Participant 12, viewed himself as a situational leader as he said,

"whilst consultation to achieve a milestone or certain deadline, has been a discussion or a strategy when it was organically grown as a group, but now given that we aim to get [x result] then what that means is a achieving deadlines. We need to be checking routinely and then start to tap shoulder required. Maybe consultative becomes slightly more forceful from time to time".

Participant 16 also viewed herself as a situational leader, when she said,

"When I think of all the things you have said, I think I am a little bit of all of them but I wouldn't say its my way or the highway, but I would like to think that I communicate with staff and also get their feedback as well".

Other leadership styles identified in this study included one person as a transformational leader, in round one interviews, two people identifying as practicing servant leadership; one person that utilises a collaborative leadership style in rounds two and three of interviews.

For participant 5 who identified herself as a transformational leader, leadership for her meant;

"You have to get people involved, but you have to get, the whole idea of a leader is not just to lead a team. The whole idea is to get people on board, have a look at their visions and your vision. Keep getting back to them and say, have I missed anything? How can I help you?" Interestingly, participant 14 who identified as a servant leader in round 1 interviews said,

"A true leader has a servant spirit". So, you can only lead if you know what it is like to be the person who is being led."

In subsequent interview rounds, this participant's style changed to consultative leadership. The self-reported leadership styles are reflected in Table 13 below.

TABLE 13 - SELF REPORTED LEADERSHIP STYLES

									Sel	f Re	epoi	rted	Lea	ders	ship	Styl	e Ca	teg	oris	atio	n							
			Round 1 Interview								Round 2 Interview						Round 3 Interview											
Participant	Gender	Transformational	Situational	Servant	Participative	Democratic	Consultative	Collaborative	Authoritarian	Authentic	Transformational	Situational	Servant	Participative	Democratic	Consultative	Collaborative	Authoritarian	Authentic	Transformational	Situational	Servant	Participative	Democratic	Consultative	Collaborative	Authoritarian	Authentic
Participant 1	М							~									✓									✓		
Participant 2	F						 ✓ 									✓									✓			
Participant 3	М						~									>									>			
Participant 4	F															>									>			
Participant 5	F	~										>									>							
Participant 6	М						~									<									<			
Participant 7	F		 Image: A set of the set of the																		<							
Participant 8	М		~									>									<				~			
Participant 9	F		✓									~									 Image: A start of the start of							
Participant 10	М			 												~												
Participant 11	F						~									>					~				~			
Participant 12	М		✓									~																
Participant 13	F						~									>												
Participant 14	Μ			 ✓ 												✓									✓			
Participant 15	Μ						~									~									~			
Participant 16	F		 ✓ 																		~							
Participant 17	F						 ✓ 									 Image: A start of the start of									 Image: A start of the start of			

Source: Developed for this research

Table 13 above outlines that consultative leadership style was progressively prominent in the 3 interview rounds, indicating a general trend in the adoption of this leadership style as a style of choice.

In the 3 interview rounds across the 17 leaders in medical imaging, situational leadership style featured in a third of the participants as the second most dominant from consultative leadership.

4.8.1 Nature of Leadership and Impact on Success

As discussed in section 3.4, step 12, for each project or process, success was measured against pre-determined KPIs or objectives. For example, if the objective was to deliver on the procurement and commissioning of a new CT scanner on budget and on time, that project was deemed to have been a successful project. It was noted that 41 out of 43 projects were successfully carried out and meeting their set objectives. More telling is that across the 3 interview rounds, the most dominant leadership style was consultative leadership style which averaged 56% compared to the other styles practiced by the leaders in this research setting.

Participant 14 lamented;

"I would like to think that I have a consultative type of approach. I don't want it to be a truly dictatorship kind of thing but I do believe that if you are in the leadership world, sometimes you do have to take those difficult decisions".

Participant 14's sentiments suggests that this participant would be a situational leadership in these settings when circumstances permit.

The findings presented in Table 13 above show that the dominant leadership style was consultative leadership for several medical imaging leaders, across

multiple data collection rounds. Figure 11 below also shows the leadership style utilised by the leader.





Source: Developed for this research

As shown in Figure 11 above, the dominant style of leadership was consultative leadership. Most leaders successfully undertook change projects utilising this leadership style. This research also showed that situational leadership resulted in failure to achieve the intended project objectives. The next section presents reasons for project success or failure in medical imaging in healthcare.

4.9 Reasons for Project Success Or Failure

Following the coding process, reasons for project success were identified. These reasons for success were stratified with the top five reasons for project success identified as shown in Figure 12 below. These results are explained in the paragraphs that follow in order of their frequency.





Source: Developed for this research

4.9.1 Stakeholder Engagement

Stakeholder engagement appeared 44 times as the most prominent reason for project success. In one project, participant 5 said,

"Definitely engagement and engagement across the board."

Also, in a different project, when asked about key reasons for project success, she said, "

"Engagement, engagement and engagement. And I think that's the big thing because you can lead a change process, but if you don't take the people with you, they don't own it".

Participants reported that stakeholder engagement was very crucial in project/process success. To effectively engage stakeholders, participant 8 noted,

"It was established very early on who would be involved. Everyone has worked extremely collaboratively throughout the whole process, so we have got internal experts, various people who are familiar with these items, that came together and make sure that there was adequate representation across the group."

They also further articulated that the involvement of stakeholders earlier in the project increased the prospect of project success, when they said,

"So, involving ED earlier on and involving eMR earlier on to say, is this something you would like? We have looked into it, we know we can do this, is this something that you would like and then if they say it's something that we like, then it is developing that plan, having key milestones to hit."

This highlighted the need for early stakeholder identification and involvement. In the project setting, stakeholders were identified by participants as both internal and external stakeholders such as staff members from the department. Participant 2 suggested who the internal stakeholders were when they said,

"There was a lot of consultation in the department between the groups involved, specifically between myself, the CT Senior Radiographer and NUM for Imaging as well."

Internal stakeholders were identified as participant groups as identified above who worked at the hospital or within the LHD, including local departmental staff and those belonging to other departments other than Imaging Departments, such as Registered Nurses or Physicians from the Emergency Department. This was aptly described by participant 8 who said,

"Engineering was heavily involved. Obviously, power issues, transport; delivery. There were quite a few different parties involved."

Some participants also identified that timely engagement of their key internal stakeholders such as staff was a key enabler to project success, such that whenever they wanted to commence a project, they would engage their staff first, when participant 2 said,

"So basically, in here, if there is anything new to be introduced, we always talk to Senior (Radiographers) people first or as Seniors, meeting with them. If there is something that's urgent, you have to consult, you have to ask them what do you think? That is the normal process."

Reflecting on the need to consult the affected stakeholders, participant 11 said,

"I should probably say that before I made the change, I put out the roster templates, showed it to the staff for 4 weeks."

This research project established that there were several stakeholders as summarised in Table 14 below:

Internal S	takeholders	External Stakeholders					
Radiographers	Radiologists	NSW Ministry of Health					
Radiology Registrars	Registered Nurses	Ambulance Service					
Radiology Support staff such as wardsman	Other Departments such as Information and Technology Department; Emergency Department)	Sydney Local Health District					
Heads of Departments of other Clinical Specialties	Nuclear Medicine Technologists	Equipment Vendors (Companies)					
Emergency Department Physicians	Intensive Care Unit Physicians	Royal Australian College of Radiology					
ReferralPartners(OtherClinicalspecialistsrequiringRadiology exams)	Local Health District Executive	Interventional Radiology Society					
Hospital Executive		Referral Partners (Other Clinical specialists requiring Radiology exams)					

TABLE 14 – STAKEHOLDER TYPES

Source: Developed for this research
External stakeholders were identified as participant groups as detailed above that worked outside of the hospital or LHD, that had interactions with the LHD. These stakeholders included NSW Health staff or divisions, specialists training colleges, equipment vendors as outlined above.

The engagement of these stakeholders was identified as being crucial to the success of projects or processes in the LHD. External stakeholders such as equipment vendors provided support to the LHD in the form of innovations in the market evidenced by remarks from participant 2,

"They do have a very nice product and we do buy gowns from them".

This participant also suggested that equipment vendors value-add in this mutually beneficial relationship, when they said,

"But putting this aside, all other gowns we had in this department were actually screened as well."

Stakeholders such as the Royal Australian and New Zealand College of Radiology and the Interventional Radiology Society are the custodians of standards of education and practice for both diagnostic and interventional radiology in NSW and Australia, As such, it was important for participants to engage with them in their projects as noted by participant 3, when they said they required,

"...support from our Interventional Society and the College this year, things have actually changed, they have gone on to sort of, the College and the Interventional Society have come together on this project."

Stakeholder consultation took several forms in these projects such as:

- Phone calls
- Meetings (both face to face and virtual meetings).
- Discussions
- Presentations

Representations

Participant 6 identified this engagement as,

"... a lot of external consultation, multiple phone calls, meetings."

For clinical projects, engagement was informed by what was considered clinical best practice. For example, participant 6 also said,

"It was based on what we thought was best practice. And the best practice is that lung scans are preferable in young women and pregnant women, particularly those who have clear x-rays and no other pre-existing conditions."

4.9.2 Planning

In this research project, planning was defined as the process for the identification of goals to be achieved; creation of interventions to achieve them; the arrangement of these interventions and implementing these interventions and monitoring that they were carried out appropriately to attain the identified goals.

Planning was identified as the second most important reason for project success with a frequency of 17 occurrences. In this project, it took the form of business planning and general planning. Participant 10 recognised planning occurred at several levels when he said,

"...the business or corporate side, we had regular meetings with the Business Manager. We also engaged all the high-level executives, the GM, Corporate, Director of Finance at the highest level, the hospital executives at District level and we had regular business meetings".

Planning was undertaken initially at the commencement of the project when the project was identified as being required within the medical imaging setting as identified by participant 15. A Project Lead would be identified and given the task of leading the project. This individual, in consultation with other key stakeholders would identify potential key stakeholders and organise meetings with them.

Project participants were then involved in identifying other key stakeholders required to participate in the project.

General planning involved the deliberate process or undertaking initiatives to ensure that projects were conducted successfully. In this research project, ongoing planning was conducted through regular meetings by project leads in consultation with the project key stakeholders for the duration of the project. These meetings had minutes produced and action lists developed. It was also noted that meeting minutes provided most project documentation for projects, with limited to no other formal project documentation. Project documents typically available for projects are: project scope; project plans with milestones; progress tracking tools such as GANNT charts; risk registers change control processes; communications plans; roles and responsibilities (Carstens & Richardson 2019; Fewings & Henjewele 2019).

Some participants identified that it was crucial for them to have sub plans for their projects in addition to the main project plan. Participant 5 identified that a communication plan was crucial to manage project risks by ensuring that all stakeholders were notified of relevant project information in a timely manner, when asked about how they managed risk, the participant said,

"Use your communication plan."

Participant 8 also identified that the development of project planning tools allowed them to effectively manage their projects. These included:

- i. Action lists to ensure that items were assigned responsibility and were marked off as actions were completed.
- ii. Document templates with concise and clear information.
- iii. Meeting schedules

One of the keyways this participant kept stakeholders informed was through the development of an e-mail template, saying,

"Then we also developed an e-mail template, where every two days, I would give a project update with key milestones to our district and some clinicians that were heavily involved such as the Head of Department for ICU and the Director for ED."

This process facilitated the clear and concise dissemination of information to key stakeholders.

The planning process of projects involved several stakeholders as identified above as well as the undertaking of certain activities prior at specific intervals prior to the other project stages. For example, participant 15 said,

"I think it was planned day of delivery; booked for the application specialist; equipment; unpacking and everything was done prior to the day of installation, so the machine was unpacked and things were put all together fully configured, software and all those things were upgraded and everything was ready."

Where projects had a knock-on effect on other hospitals and the Ambulance service, there was another level of planning that required the agreement of those later stakeholders. Participant 15 suggested,

"The main stakeholders were in the hospital, the referrers for CT scans and outside were; Liverpool Hospital where CT scans could be done after hours or in hours for all urgent CTs and Ambulance Service. So, the two main stakeholders were briefed how the project was going to be undertaken and how the project could be run smoothly. There was planning involved, in consultation with the Hospital Executive so that the project went well ahead."

Such collaboration was crucial as it resulted in added workload that required the planning and resourcing of these other stakeholders in anticipation of extra patients. Planning also involved the identification of peak activity periods and appropriate downtimes where disruption would not be as significant to the daily operations of the hospital or imaging service. One participant summed this up when they said,

I think, we made sure we allocate a day when there weren't any major studies, how busy the department was and as a result we informed the stakeholders of the project that we were going to do, which impacted less on patient services."

4.9.3 Education and Training

Education and Training appeared with a frequency of 11 times, as the third most prominent reason for project success. Participants identified that for a project to be implemented successfully, there was a requirement for staff to be educated and trained in the correct use or process, which was the subject of the change. Following the planning process identified above, the project lead was able to identify stakeholders that required education and training for change to be successfully embedded. Participant 8 said,

"We had to meet with all these people for education and information on what the requirements were, how we were going to roll it out and when we needed it completed by. There was a lot of support from the Directors and Chiefs from each department which meant that resources could be allocated, and this could actually be done. This participant, during a change project to the IT platform used in Imaging, said,

"We also had to ensure that the Radiologists and Nuclear Medicine Physicians were trained on how to use the system once it went live, with the system differences pointed out from the old system and how to navigate the system. Training was also required for Radiographers; Nuclear Medicine Technologists; Administration staff to ensure that they were across the system at crossover time."

The training provided was in the form of group and individual settings as stated by participant 8,

"We set up a training room and ran regular training sessions in both group and one on one sessions over a week period to ensure that we captured everyone."

This training was provided by utilising the subject matter experts within the department or external to the department as noted by participant 13,

"In the beginning, I think it was a training process and getting people into the small groups and getting the PACS Administrator to demonstrate to them how the process works and also the Senior Radiographers reviewing the work being performed to see if processes were being adhered to."

The use of subject matter experts was important in the embedding of change as end users obtained their information from appropriately skilled individuals fit for purpose.

Participant 2 also advised that when training could not be offered to a wider group of people, a train the trainer process was utilised, when she said,

"Key people were trained, that this is what we do, with any new product that comes to the department."

This also suggested that in this setting when there is a change in the workplace, education and training of staff is valued and a routine occurrence.

4.9.4 Resetting Expectations

The joint fifth prominent frequent reason for project success was resetting expectations, with a frequency of six. It was identified that when change was required, participants suggested that there was often need to reset expectations. Resetting expectations is the process of establishing new expected standards or behaviour or performance upon which people would be measured. For example, if people previously had an expected output of 15 units, and the new standards was 20 units, this required communication with key stakeholders. Participant 11 said,

"So, if I was a consultant that was rostered, I would know the work for the day and what is expected of me. If there are enough Radiologists to cover each of the modalities, work should be done. ... And the same applies for Registrars as well."

In resetting expectations, it was identified that equity was critical in the process. Another participant also said,

"So, everyone has their defined work, no one works more than the other. So I counted the number of sessions and the modalities and made sure that work was distributed equally for all consultants for all modalities and based on how many days they are working, so that one person does not do too many MRI and the other person does too many X-rays."

The participant further stated,

"The only reason I was able to get the support of my Consultants was because they saw the benefit in it as well, because work was distributed equitably." Engagement with the key stakeholders in resetting expectations is critical in ensuring an agreed process would be utilised with minimal resistance. This was suggested by participant 13 who said,

"I should probably say that before I made the change, I put out the roster templates, showed it to the staff for 4 weeks."

One participant also recognised that when there was a perception of a mutually beneficial outcome for stakeholders, they were more likely to be amenable to change, when they said,

"They were quite happy with that and on a weekly basis, they get a mix of everything. So that's why there was buy in from the staff as well because they saw the value in that."

4.9.5 Processes and Procedures

Processes and procedures featured as prominently as resetting expectations. It refers to the presence of processes and procedures in the change management project itself. For example, for a project to be successfully implemented, there is a blueprint to be followed by participants in undertaking the change process. This was particularly important as noted by participant 10 as the key ingredient for project success, when they said,

"We also set up our initial practice in the Cyclotron from day one to be to the TGA standard and so we have documentation of about six hundred and eighty runs with documentation of quality measures as you get process we have been able to establish a track record."

The use of stringent processes and procedures facilitated the success of this project.

It was also noted that when there was an existing precursor process in place, subsequent processes could be implemented with relatively good prospects of success. This was supported by participant 1 who said,

"The key reasons were that there was a process in place. A policy in place."

The existence of other support processes was crucial in the success of change management process. For example, during a CT installation project at one hospital, there were no significant concerns as patients could be easily transferred to neighbouring hospitals. The existence of previous similar projects laid out the blueprint for this project to be successfully implemented.

4.9.6 Other Reasons for Project Success

This study also identified other reasons for project success, which are discussed in the following paragraphs.

4.9.6.1 Resourcing

Project resourcing came up as the sixth frequent reason for project success. It refers to the resources made available for the project to implement successfully. Where key infrastructure projects were being carried out, the Capital Works and Infrastructure team was involved in these projects. While it added a layer of complexity to these projects, it also brought an increased level of rigour. This was in the form of a more robust level of documentation and in most cases, the use of external project management specialists. This resourcing tended to impact on change projects undertaken in the medical imaging arena.

When projects undertaken in medical imaging had an impact on key hospital and LHD matrices, there was a willingness of the hospital and LHD executives to provide more resources to these projects. For example, for a change project that would lead to an improvement in timeliness on patient care, for example the ETP, additional resources were made available. Participant 6 suggested,

"We got one Orderly that was actually given to us. So, it was a big plus." For this project, this participant noted that there was an additional wardsman (or orderly) that was allocated to improve timeliness of patient care.

It was also noted that most projects conducted during the research period did not have additional human resources to support the projects being conducted. The medical imaging departments were required to undertake projects in addition to the day-to-day role. Participant 3 lamented,

"Everything was done purely from your self-energy; I have relied on the doctors and staff. It's all because you have got committed and enthusiastic people otherwise you won't have it."

The lack of designated time for project related activities was identified as a constraint for effectively undertaking projects within the medical imaging setting.

4.9.6.2 External Support

External support was the seventh most frequent reason for project success with four occurrences, like Governance Arrangements: Negotiation and Subject Matter Expert Engagement. These three will be discussed in the following paragraphs.

External support refers to the support provided by external stakeholders in the carrying on of change management projects. This support may be in the form of additional training, product expertise and implementation guidance. For

example, with the implementation of a new product and process, participant 2 said,

"They provided help. They do have a very nice product and we do buy gowns from them. But putting this aside, all other gowns we had in this department were actually screened as well."

Within clinical settings such as medical imaging, external bodies that provide oversight for clinical standards, credentialing and accreditation are crucial to change management projects driven by changes in standards. While undertaking a change management project, participant 3 said,

"I had support from our Interventional Society and the College this year, things have actually changed, they have gone on to sort of, the College and the Interventional Society have come together on this project."

Such support for the accreditation and credentialing body is crucial to some of these change management projects.

4.9.6.3 Governance Arrangements

Governance in this context can be defined as the process of decision making and how the process is implemented. In a clinical setting, the NHS defines Clinical Governance, as a systematic approach to maintaining and improving the quality of patient care within the National Health Service (Smith, A, Latter & Blenkinsopp 2014; Walshe & Chambers 2017).

Within the healthcare setting, projects are usually managed by way of committees. A project committee oversaw each project. Depending on the project complexity and determined importance to the organisation, a steering committee was also established as the apex leadership body for the project. Each project also had an identified project lead who led each project.

For each project, there was a need for a structured approach in undertaking projects, with Participants 10 and 14 suggesting that there was a need defined governance arrangements and an understanding of what these roles entailed.

Where change management projects involved multiple departments; hospitals and departments, formal documents such as memorandum of understanding were identified as being crucial in the change management process. Participant 7 said,

"There is a memorandum of understanding around that. So, there is that constant information sharing."

4.9.6.4 Negotiation

In the change process there was a requirement for negotiation in achieving project outcomes. For example, if timelines were no longer achievable the Project Lead successfully renegotiated timelines with key stakeholders to ensure that the project came back on track. Participant 5 suggested the following when looking at this,

"And renegotiating those timelines if there are some issues. Having like a definite timeline for the project but also maintaining a bit of flexibility within those."

Where clinical change management projects were involved, it was also imperative that a level of flexibility was maintained despite a set of clinical criteria being set for this project. Participant 6 said,

"So, it does not all have to fit under that criteria, there is still the added criteria of Consultant to Consultant discussion, so it is up to ourselves to accept outside of these criteria." This concept of clinical decision making that is evidence informed, was found to be an important consideration for clinicians whenever there was a change process required. This was the basis upon which senior clinicians would negotiate how clinical care would be provided.

4.9.6.5 Subject Matter Expert Engagement

During the discussion in training and education above, it was identified that the use of subject matter experts was crucial in the provision of education and training. Participants identified that for the implementation of technical projects, the use of appropriately skilled and qualified staff was crucial to the success of the project. This is evidenced by participant 10 who said,

"The key reason I think is that this is a very technical project. So, we have the right facility and equipment and I think in part is because were able to pick the best people, who are capable."

During the implementation of another project, participant 12 indicated,

"For this process, [the education and training of staff] we also had GE Engineers onsite and applications specialists providing the training, and this ensured that people knew the functionality of the system well before the "Go Live".

This suggested that the appropriate use and engagement of subject matter experts was crucial to project success. Also, in another project, one participant outlined that they engaged a specific cultural grouping with the expertise in the development of some key project inputs. Participant 6 said,

"The other things that were important were that we had a lot of input from our Indian staff."

In section 4.9, it was noted that several reasons resulted in change projects or process success. This included: stakeholder engagement; planning; education

and training; resetting expectations; and the prevalence of good processes and procedures. Participants further reported that resourcing; external support; governance arrangements; negotiation and subject matter expert engagement were critical factors that facilitated project success. Section 4.10 presents the strategies utilised by medical imaging leaders to achieve project success.

4.10 Strategies Utilised for Project Success

Several strategies which had not been identified in the literature review in chapter 2 were identified that led to successfully implementing projects within the LHD as shown in Figure 13 below. While it was noted that there was some overlap between the key reasons for project success, it is important to distinguish the two concepts and discuss them separately. For strategies implemented, it was necessary to discuss the top 10 frequently occurring strategies utilised for project success. The researcher decided to highlight the top five strategies and a brief discussion of the other five strategies in the paragraphs that follow.







4.10.1 Consultation and Engagement

Consultation refers to the degree to which leaders in this research project sought the advice or communicated about the research, with potential stakeholders to improve project success prospects. Engagement, on the other hand entailed the regular involvement of stakeholders throughout the project cycle providing feedback as required. In both cases, stakeholders had the opportunity to participate in the project and provide input aligned with their interests.

Consultation and engagement was the single most prominent strategy utilised in leading change projects successfully in this research with a frequency of 38 times.

Participant 13 clearly outlined,

"I think consultation is very important. From my days as being the Manager here, consultation has been a key thing. If you consult people, they are part of the group, you will have more success because they are part of it, and they will make it a success"

Where projects predominantly involved internal staff, this participant emphasised the need by stating,

"I consult my staff. Any changes that I need to do, without consultation, I never do."

Participants 1 and 8 said that consultation and engagement involved meeting with many diverse stakeholders as identified in section 4.9.1.

Participant 5 also suggested that consultation and engagement took the form of,

"Lots and lots of meetings with the key stakeholders",

and meeting consolidation with existing ones being key to managing the everincreasing meetings required. Consultation and engagement also require effective leadership. Participant 12 found out that they had to adopt strategies to manage dominant stakeholders there were some strong characters that sought to impose their styles on the meetings, when they said,

"It was but lessons learnt very quickly that all the stakeholders would want to have their authority and their way of managing to be heard, how things are done in their way. So, we have to listen to them, but there are some project managers who just want things to be done their way".

Consultation and engagement were identified as a key strategy to achieve project success involving the early identification for key stakeholders, engaging and devising effective ways of leading them. The timing and format of the engagement facilitate seamless project implementation in that participants were able to plan project activities at appropriate timepoints, such as downtime.

4.10.2 Workload Management

In the implementation of new change projects, it was imperative that the workload of the staff be defined and appropriately resourced so that the project could be implemented successfully and sustained. Workload management featured as the second most prominent strategy, appearing 22 times.

Participant 11, indicated that they had to manage their resources appropriately and said that they designed,

"...a roster so that each of the modalities were covered for the day. And then once that was done, we created a worklist that was in the RIS for each individual Consultant. For example, if a consultant is on CT, all the CTs that are done for that day will be assigned specifically to that Consultant".

Workload management also meant staff expectations required to be reset, with staff made aware of these new expectations. Participant 9 suggested,

"So, if I was a consultant that was rostered, I would know the work for the day and what is expected of me. If there are enough Radiologists to cover each of the modalities, work should be done".

Further to resetting staff expectations, participant 14 suggested equity of workload is key in implementing change, when they said,

"they used a framework of the plan going forward which was to identify the main leader and the leaders at other hospitals and to form division of labour accordingly by dividing the workforce pool into subgroups".

When a project involved the installation of new equipment, participant 2 indicated that it was imperative to have workload organised and resources allocated so that change could be affected and embedded. The role of these resources was to, start early,

"...make sure everything's set up for everything. Also, in terms of training, with this introduction of extra staff for the equipment, staff has got more time to actually do the work more efficiently".

Participants 2, 3 and 12 also suggested that workload management including providing additional human resources for new projects was crucial for project success as it allowed for troubleshooting and issue resolution. Participant 2 suggested,

"...minimum staffing levels were required to undertake certain tasks".

These participants highlighted the importance of managing workload in change management processes or projects. While resources were identified as necessary to manage workload, other key components included the equity considerations in the distribution of workload.

4.10.3 Education and Training

Education and training appeared a joint third prominent strategy utilised, together with communication/explained communication, appearing 18 times.

Education and training in this context is the process of imparting new knowledge to enable participants to achieve the desired results.

When new innovations were implemented within the service, education and training was a key strategy utilised to impart knowledge to staff. Participant 8 noted,

"We also had to ensure that the Radiologists and Nuclear Medicine Physicians were trained on how to use the system once it went live, with the system differences pointed out from the old system and how to navigate the system"

Underlining the important role of education and training and to reinforce new processes or systems, participant 7 also suggested she had to,

"...get back to the staff members and just re-educating them as to this is the correct financial class that we now need to use because when people go through and do it automatically it becomes quite a habit".

Education and training took various forms which included an assessment of knowledge retention, demonstration and review of staff understanding of new knowledge, through the adherence to the new training methods as taught. For example, participant 13 suggested,

"In the beginning, I think it was a training process and getting people into the small groups and getting the PACS Administrator to demonstrate to them how the process works and also the Senior Radiographers reviewing the work being performed to see if processes were being adhered to".

Online training was ongoing support was also used by Participant 17 who said,

"We had an online training as well. And then we also had the recruitment team that looks after Radiology. So, again we go back to that person for any help".

As pointed out by participant 8 in a new system implementation, training scheduling is very important to ensure maximum participation. For example,

sessions were organised in the mornings and afternoons across the week. The participant said,

"We setup up a training room and ran regular training sessions in both group and one on one sessions over a week period to ensure that we captured everyone".

Participant 15 used a slightly different approach of external training and simulators for training in new equipment rollout. The participant said,

"So initial training was set up elsewhere with a core group of radiographers that was trained outside and then we had the radiographers trained in the hospital with a demo tool for them to use, or a simulator".

This approach was also utilised by Participant 3's project where the participant, said,

"we had international speakers come talk to us".

For new systems and processes impacting on several staff groupings the leaders ensured that they received training. This ensured that all staff understood the new system or process. Another participant also suggested,

"Training was also required for Radiographers; Nuclear Medicine Technologists; Administration staff to ensure that they were across the system at crossover time".

As outlined above, education and training were identified as a key strategy for successfully implementing change in projects. Leaders were successfully able to conduct projects by ensuring that they undertook several considerations for their staff, including mode and delivery of training so that they addressed the diverse staff groupings and requirements.

4.10.4 Explained Communication/Communication

Communication or explained communication in this project was the process used by participants or the project in disseminating and sharing information about the project, in a way that was clear and easy for key stakeholders to understand.

Underlining the importance of communication, four participants discussed why communication was imperative. Participant 8 said,

"I can add that Communication was key in this project. Ensuring that the Hospital was appraised of what was happening, the LHD was updated of the progress; IM&TD were involved in the process as they were part of the tests for various scenarios, Sydney LHD were also consulted as a copy of the PACS was being made".

When asked what strategy they used for project success, participant 3 said, *"Well I think it was a communication and learning the processes".*

Responding to the same question, participant 4 said,

"Yeah a lot of communication with staff involved like our booking/clerical staff and sonographers".

When asked about the key learning from their project, participant 16 said,

"Definitely Communication; engaging with different departments and talking with different people about how to set up things"

Participant 5 stated, "*Use your communication plan*". This was a formal communication plan which outlined the way in which communication would be carried out throughout the project duration. This included mode of communication, duration of communication, frequency of communication and stratification of shareholders according to communication mode.

Communication took different forms such as e-mail, face to face (one on one), general meetings and verbal. Participant 2 said,

"So, I think I send them e-mails (and explained) that this is what we are trying to do and everyone was pretty much happy because it is easier to manage".

Participant 8 also said,

"So, it was mainly email at the beginning".

Participant 14 also suggested highlighted the use of a similar communication mode when he said,

"The strategies we used were Strong Communication by verbal means or in writing through e-mail".

Participant 7 explained that their communication involved a lot of discussion with staff and senior management while participant 17 also found that having a faceto-face meeting with each staff member was an effective method of communication.

Meetings were particularly an important mode of communication with participants 7 and 15 recognising the shortcomings of other communication modes such as e-mail and the ability to allay fears and therefore a transparent platform for effective message delivery.

While meetings were an important tool for communication, Participant 5 also indicated that, regular communication regarding the project was instrumental in its success when they said,

"...bringing it up in every ward meeting and you know local meetings as well".

It was important for communication to be customised. Where staff could not access e-mail or attend project meetings, participants 7 and 12 adopted different

strategies such as printing documents and innovative ideas such using social media. In conclusion, communication was noted as a key strategy in project success. Different participants utilised this strategy slightly differently. Of note was the use of a formal communication plan to guide the process.

4.10.5 Planning

The strategy of planning appeared with a frequency of 10 times, as the 5th most important strategy in implanting change projects effectively. Planning is the process of undertaking activities in a deliberate, structured and logical approach after considering relevant factors that affect the project. As participant 15 noted,

"I think the planning for the project was more important without impacting on patient flow".

In the LHD, it was noted that it meant different things for people. For participant 3, planning meant that there was an implementation plan for the project where certain activities would be undertaken prior to other activities being undertaken. Participant 3 said,

"We had a plan. We went and educated ourselves. We had mentors". For participant 10, it meant goal setting as they said, *"We had different goals as well".* The participant underscored the importance of planning when they went on further to say,

"And then we also planned and wrote the right procedures/protocols. So, there's very careful planning and making sure that it's implemented correctly. So, from my point of view I think we follow a lot of good production principles, the GMP⁷ principles".

Where additional financial resources were required, a specific process was required to be followed for projects within the LHD. Participant 15 explained this formalised process of seeking hospital and LHD approval. During the

⁷ Good Manufacturing Practices

implementation of a project, participant 15 highlighted key points on selecting downtime dates when they said,

"I think, we made sure we allocate a day when there weren't any major studies, the busyness of the place and as a result we informed the stakeholders of the project that we were going to do, which less impacted on patient services".

The participant further suggested the completion of prerequisite steps before downtime to facilitate a smooth transition. Even with the best of plans, participants noted that project schedules might have some challenges. Participant 5 suggested that plans could be renegotiated with stakeholders, when she said,

"And renegotiating those timelines if there is some issues. Having like a definite timeline for the project but also maintaining a bit of flexibility within those".

Participant 12 also recognised that planning is not a static process but an iterative one when he said,

"And then the other thing is not necessary every project you can hit a milestone as per schedule. So constantly reviewing, revising a schedule".

In summary, planning played a key role in in successful project implementation. While pre-project planning was important for projects to take off, planning during the project implementation was also very important to consider changes that were observed as necessary as projects progressed.

4.10.6 Other Key Strategies Utilised

The following paragraphs briefly discuss the other top 10 strategies utilised in achieving project success.

4.10.6.1 Evidence Informed Practice

Evidence informed practice is a key consideration for clinicians in clinical decision making and was described by participant 6 as clinical best practice. The participant in implementing a new clinical intervention, said,

"It was based on what we thought was best practice."

As noted in the project demographics section of this chapter, 78% of the participants in this research were clinicians. Unsurprisingly, evidence informed practice emerged as the joint sixth most prominent strategy (together with Governance Arrangements) utilised in effectively managing change processes. Clinicians are more responsive to practice driven by clinical best practice.

Evidence informed practice was evident in several scenarios. Participant 15 utilised it for additional resource requests for senior leaders so that the project could be implemented. She said,

"When the staff actually brought it to our attention that they needed more support in the evenings. So, I spoke to the two PACS Administrators and said can you collect this data for me".

Participant 7 utilised evidence to assess whether training had been effective and proper practices were being implemented when she said,

"We had staff X go into RIS and build reports so that I could capture the daily hospital billing of what patients were coming through to ensure that the correct financial class was being used".

Participant 11 used evidence as a two-pronged tool, to support a project justification and drive project traction when using CT activity data to demonstrate the need for change and support a request for resources.

It was noted that the use of evidence informed practice played a pivotal role to the success of a project. Clinicians were noted to be receptive to projects that referenced clinical best practice. Administration and support services used the strategy to both support education requirements and project justification both to sponsors and on an ongoing basis.

4.10.6.2 Governance Arrangements

Governance arrangements in this LHD referred to how accountability was distributed across the various levels of the project. Participants 8, 11 and 14 noted that for most projects including high-risk projects, more stringent requirements applied to them, including a risk management approach and consultation with all key stakeholders. This consultation included the clarification of roles involved in the project.

The LHD structure was discussed in chapter 1, section 1.1.4. When participants refer to "District" they will be referencing the Senior Leadership team of the SWSLHD, the Chief Executive and her direct reports.

For projects involving external stakeholders a similarly more formal and complex reporting structure including a communication plan was adopted as pointed out by participants 5 and 8.

Participant 10 alluded to having good processes and formalised ways of managing changes within the project. He said,

"So basically, we defined the vision and also the process. So, the process is for example, they need to have good procedures, training and everything with change control"

Participant 14 also suggested,

"There is a memorandum of understanding around that [the delineation of roles and responsibilities]. So, there is that constant information sharing". In implementing projects, it was noted that having clearly defined reporting and accountability structures ensured that projects were approved at the appropriate levels of the organisation and could therefore be implemented successfully with minimal challenges. Any challenges faced could be escalated to the appropriate resolution authority as noted by participants 8, 10 and 11 above.

4.10.6.3 Trials

The strategy of using a trial period was the eighth most prominent strategy utilised by participants in this research with a frequency of seven times. A trial period in this research meant that participants undertook a trial period prior to a full-blown implementation of the change.

Participants 4, 11 and 13 were major proponents of the trial strategy which have been used in their departments in several situations. The trial process was utilised when there was no consensus, or the leaders were unsure of whether the change would result in the desired changes. Participant 11, supported this strategy, when she said,

"But if there is not a unanimous decision, and some people are not sure of it then the only thing I do is have a trial period".

When additional resources were required to embed changes, participant 13 suggested using this trial strategy as it builds the evidence base to support implementation. Trialling thus provides a setting within which evidence can be gathered in a real time setting and assumptions made previously can be approved or disproved.

In conclusion, it was noted that trialling was a strategy that worked for participants in several situations. It was used a strategy to get consensus on a course of action. When evidence was required to support a decision or a request for additional resources, trialling was also utilised successfully for this purpose. Finally, trialling was utilised effectively as an evidence informed practice tool in real time but temporary settings.

4.10.6.4 Senior Clinician Negotiation

Senior clinician negotiation was the 9th most prominent strategy in this research with a joint frequency of 6 times, with the strategy of Leader Approach as discussed in section 4.10.6.5 below.

Senior clinician negotiation in this research referred to the process by which senior clinicians negotiated on a case-by-case basis for a treatment intervention that fell outside of established protocols as understood by Participant 6, who said,

"So, it does not all have to fit under that criteria, there is still the added criteria of Consultant to Consultant discussion, so it is up to us to accept outside of these criteria".

As an example, Participant 6 highlighted a patient case when he said,

"That's the one that XX [Nuclear Medicine Physician] accepted. This patient already had a CTPA and a confused patient. And the CTPA people said that they could not get this patient to breath hold. So, we ended up doing a lung scan for a demented patient"

For senior clinician negotiation to work effectively, there is need to be a robust framework as suggested by participant 6 who said,

"Also, it was on a Consultant-to-Consultant basis and we did set up a lot of criteria and made it very tight"

Participant 6 also said, there is another decision-making tool when they said,

"So, the checklist that we use for our purpose here is this one, the VQ criteria, and now from Dr X [Director Medical services] is happy with this"

When diagnostic testing could be delayed and interim clinical interventions implemented, Participant 6 also outlined that negotiation could occur when he said,

[Physician AF] had another view that most patients can be on anticoagulants for at least 24hrs. As long as they get a diagnostic test within that timeframe, he was very happy with that".

The rescheduling of diagnostic testing therefore meant workload could be managed more appropriately and the proposed clinical process changes being sustainably managed.

In conclusion senior clinician negotiation was a key strategy utilised in effectively leading change management approaches. It was used on a case-by-case basis to manage scenarios that fell outside of established clinical protocols. Senior clinician engagement also facilitated the prioritisation of workload if interim clinical interventions could be affected.

4.10.6.5 Leader Approach

The leader's approach, like senior clinician negotiation was the 9th most prominent strategy in this research with a frequency of 6 times. Leader Approach in this research referred to how the leader's approach towards both the project and staff influenced the outcome of the change project or process.

Participant 5, 9, 10 and 11 suggested they are hands on leaders and lead by example, from the front. They suggest that, leading by example demonstrated to

the group the importance of the project and how seriously the leader considered it.

Being hands on allowed projects to managed according to the project schedule as outlined by Participant 10 who said,

"We used to meet weekly and don't meet as often now because things are on track. I think this kind of hands on management has to be implemented for a big project".

Participant 9 attempted to be proactive in her leadership approach and said,

"Well because I had had time to think about it and it was a forum for discussion. When you are presented with a case and it's wrong, it's too late or you can't go back or it's too hard to go back and fix it. So, it was trying to be proactive".

Being proactive for procedures was crucial from a patient safety and best practice perspective and the department usually had one opportunity to get the study right the first time.

The leader's approach was crucial as a strategy in effectively implementing change. Some leaders suggest leading from the trenches with a hands-on approach was an effective way of effectively leading change. Other leaders preferred to be more proactive by getting tests completed well the first time every time.

The first part of 4.10 presented the top 5 strategies utilised for project or process change management strategies. These included consultation and engagement; workload management; education and training; explained communication/communication and planning. Through the consultation and engagement process, participants noted that stakeholder engagement throughout the project life cycle was crucial to project success. Similarly, participants found that robust communication process through fit for purpose or communication modes preferred by participants resulted in project success.

In the second part of 4.10, it was noted that other strategies utilised were also important in successfully leading change within medical imaging. For example, SME engagement allowed the research participants to draw on the skillsets of external experts to the department. Smart goals allowed the leaders to have tangible outcomes in their projects and a meaningful way of measuring such outcomes. It was also discussed that the strategy of having process improvement as an outcome, resulting from the change project was important as it led to traction in project. This was because participants could clearly see where improvements would be made to daily workflows and therefore their work output. For example, where new equipment was being installed, participants identified that choosing the right equipment had a significant impact in that project succeeding as there were compatibility considerations with the existing fleet to be considered.

To effectively lead change, participants identified that there was a need to have regular meetings to discuss project matters and resolve issues that arose. Within the healthcare settings, one department could not operate in a silo and where systemic changes were required a piecemeal approach was adopted. The next section, section 4.11 presents the findings from the learned experience of leaders in successive change projects.

4.11 Leadership Style Influence from Learned Experience

The medical imaging workforce was found to be relatively stable over the research period. All participants that commenced the research project were still employed by the LHD for at least the entire period of data collection. Only one participant retired from the LHD about two months following the completion of

data collection. This finding is crucial in that this stable workforce retained corporate knowledge of leadership of change processes. Over time, it was noted that the leaders would repeat change processes and build on the prior processes and leadership styles they had previously implemented.

It was noted that within SWSLHD, there was limited formal project management documentation. Most leaders relied on their experience in leading projects and processes. Mostly these were not formally documented except for three projects that had the involvement of the external project management support. Table 13 (section 4.8) above identified that 2 participants, participant 5 and participant 14 where the only participants with a change in leadership style over time. Participant 5 identified mostly as a transformational leader earlier in the research. With experience in leadership of change processes, this participant adjusted their leadership style towards a situational style over time with a focus on collaboration. When asked whether their leadership style had changed and in what way, Participant 5 said,

" Collaboration, for sure. And I think that was in one of my earlier projects an issue. I didn't collaborate well, and boy did I learn! Because everybody just went no!"

Participant 5 suggested that not only was collaboration crucial, but also communication and feedback to project participants was important for future project success when they said,

"So, I think it's not only just the collaboration, but it is also the feedback. That continual feedback loop from people, getting their ideas, getting their visions and the taking them with the group and embedding them in the way forward. Making sure everybody is kind of involved." Participant 14 initially identified as utilising a servant leadership style. Over time, this participant's leadership style was, *"enhanced"* and adopted a more prominent situation leadership style over time. This participant suggested,

"Yes, consultation is important, but ultimately there are decisions that need to be made at a higher level and that requires stepping aside from the general population and making the hard decision when you have to."

Over time, some leaders therefore recognised that their leadership style required adjustment to meet the leadership needs of the day.

Leaders in medical imaging highlighted that meeting minutes provided the primary documentation basis for most projects conducted. They also held regular meetings with their teams and wider organisation, where project success was shared. The sharing of project successes and processes though these meeting allowed participants to be aware of what projects were being conducted within the LHD and by which leaders. The information sharing among leaders allowed leaders to share techniques around project management. As most projects were duplicable across hospitals, lessons learnt by some hospitals were found to provide a template upon which future projects could be undertaken.

4.12 Conclusion

This chapter presented the quantitative and qualitative results of the research project. In doing so, it answered the 6 research questions. For research question 1, the research found that the dominant leadership style for change management in medical imaging departments was consultative leadership. Other prominent leadership styles were situational; transformational leadership and transactional leadership, in that order, with some participants practicing situational leadership. Most participants highlighted that consultation was crucial in medical imaging due to the amount of interdependencies for projects to be successfully carried out. Consultative and transformational leadership styles were dominant across both the 78% clinician and 22% non-clinician workforces.

It was noted that the majority, 45 of the 47 reported projects, had the leaders practicing consultative leadership which positively impacted on the change project outcomes. The remaining two projects that were unsuccessful, were also led by a consultative leader.

The second research question also examined the strategies utilised in leading change projects. Consultation and engagement with multiple stakeholders were identified as a key enabler for successfully implementing change projects. When new projects or processes were being implemented, most participants identified that additional interventions such as resetting staff expectations was required to ensure that staff were aware of what output was required from them. Also, ensuring that the staff workload was manageable, meant change could be sustainably implemented. Communication/explained communication was also identified as one of the most important strategies for leading change effectively because changes were able to be articulated prior to them occurring. This led to robust discussions of the process and the merits and demerits of any proposed changes. Face to face and written communication was identified by most participants as the most common modes.

The third research question examined the nature of leadership and change management and the impact on the outcomes of change management processes/projects. Leaders in medical imaging were found to be consultative in their approach which led to successful change management projects. Such an approach was consistently used by medical imaging leaders. The fourth research question also examined the reasons for project success or failure. It was established that consultative; transformational and transactional leadership styles were adopted by most leaders in medical imaging and resulted in the success of change management projects. Several reasons were proffered for the success of change projects. The three most prominent reasons were stakeholder engagement, planning and education and training. Most participants recognised the need for early identification of key internal and external stakeholders and engagement with them as being crucial to project success. Planning throughout the project life cycle was also recognised by most participants as an important ingredient in project success as it also facilitated clear and concise information to stakeholders. Finally, most participants highlighted education and training as a key requirement for change to be effectively implemented. Thus, training usually run by SMEs ensured that leaders and staff were able to acquire new knowledge which they utilised in effecting and embedding change.

The fifth question was in relation to how the leader's style is influenced by learned experience in prior change management process and whether "learning" leaders achieved better outcomes in successive change management processes. The relative stability of the medical imaging workforce meant that the research did not lose participants during data collection and the research could be successfully conducted with the original participant cohort. This meant that the research questions could be answered. It was noted that in the absence of formal project management documentation, most leaders relied on their experience in leading change management projects. Most leaders' leadership style remained stable over the research period with minor modifications from learned experience. Leaders in medical imaging shared knowledge within their own teams and externally which allowed them to improve of their leadership styles.

Leaders in SWSLHD highlighted that meeting minutes provided the primary documentation basis for most projects conducted. They also held regular meetings with their teams and wider organisation, where project success was shared. The sharing of project successes and processes though these meeting allowed participants to be aware of what projects were being conducted within the LHD and by which leaders. The information sharing among leaders allowed leaders to share techniques around project management. As most projects were duplicable across hospitals, lessons learnt by some hospitals were found to provide a basis upon which future projects could be undertaken.

Chapter 5 - Discussion

5.1 Introduction

Chapter 4 outlined the research findings of this project, from a quantitative and a qualitative perspective. These results will be discussed in this chapter. In the discussion, the researcher uses previous similar comparable studies or relevant articles, which according to Bell, J (2014), will assist with the understanding of the topic because they are either relevant to the study or build on existing scholarship. Firstly, a summary of the general findings is outlined, and then a discussion of the results conducted. This discussion chapter is organised according to the research questions and the discussion is structured on this basis, in part with research questions as sub-headings. Secondly, the discussion that follows will contrast the study findings in this research to existing literature. Where there are discrepancies between findings of this study and existing literature, these will be highlighted.

Finally, this research sought to examine the leadership styles for leadership and change management processes in medical imaging departments. In doing so, it aims to add to the scholarly literature and practice on leadership styles in medical imaging. To introduce the discussion, a framework has been developed for this research summarising the research process and this is discussed first in section 5.2 below.

5.2 Research Framework and Recap

According to Msweli (2011), a conceptual framework shows the interrelationships between the concepts in a study. Accordingly, a conceptual
framework has been designed for this study as shown in Figure 14 below. To recap, this research commenced with 31 medical imaging leaders who were the subject of this research. The leadership styles of the medical imaging leaders were measured using the data collection tools as discussed in section 3.4.2.

5.2.1 Research Conceptual Framework

FIGURE 14: CONCEPTUAL/RESEARCH FRAMEWORK



Source: Developed for this research

The data collection tools used were the Authentic Leadership Questionnaire, Multifactor Leadership Questionnaire, and semi structured interviews. The data collection tools were critical in ascertaining the leadership styles utilised in medical imaging including the dominant ones. Ascertaining the dominant leadership styles ensured that the outcomes of the leader and the organisation could be explored. This study showed that leaders who learnt from prior projects improved the outcomes of future projects and shared the knowledge with their colleagues and thereby created and developed the organisation's capacity for undertaking change. In the next section, an overview of the findings is undertaken before a full discussion of the results is presented in subsequent sections.

5.3 Overview of Findings

This research found, while situational, transformational, and transactional leadership were successful styles utilised in medical imaging, consultative leadership was the most dominant style used. This finding contributes to theoretical literature in medical imaging. It appears that most projects undertaken in the medical imaging settings are successful, suggesting that the leadership styles practiced matches the types of projects usually conducted within medical imaging. Findings by Cameron and Green (2019); Latham (2013) and Thomson III et al. (2016) suggested that 70% of projects fail. Burke (2017); Cameron and Green (2019) and Kuipers et al. (2014) found that only 20-25% of change programs actually succeed. While the findings of this research support previous findings by Hornstein (2015); Müller, R (2017) and Serrador and Turner (2015), these are new findings for medical imaging in healthcare as there appears to be little research in this area. The authors listed suggest that matching leadership styles to project types positively influences project outcomes. This is because as identified in this research, medical imaging is hierarchical in nature with medical dominance prevalent. For example, medical practitioners in medical imaging need to develop strong referral networks with other physicians.

As a result of the nature of their roles, they undertake consultation prior to implementing change that affects other clinical disciplines. In project implementation, their leadership approach tends to permeate to other members of their leadership teams such as senior radiographers, administration managers and business managers.

All senior roles were represented in this study, such as clinical, business support and technical support staff somewhat with different perceptions towards change. For example, participant 6 believed that constant change is detrimental to the organisation to the extent that those contemplating retirement might bring it forward. Leaders, experience powerlessness; apathy and exhaustion with constant change (Brown, R, Wey & Foland 2018; Camilleri, Cope & Murray 2019; McMillan & Perron 2020). Medical imaging and healthcare experience constant internally and externally driven change (Braithwaite, Glasziou & Westbrook 2020). Internally, this may be because of clinical practice improvement; new models of care; new equipment procurement and other factors organisational factors such as changes in organisational structure. External drivers of change include the socio-political factors as discussed in section 1.2 such as changes in government and healthcare reform.

This study further found that at least a quarter of the projects were discrete projects with a duration of three to six months. Three to six-month duration were found to be appropriate project lengths for participants to retain learning for the next projects. While there were several projects spanning over 12 months, the modal time for undertaking projects in the research setting was found to be three to six months. This was indicative that the projects in this research were short turnaround projects or processes. Of the 16 participants who completed the baseline LSI, 7 (44%) were female and 9 (56%) were males. Similarly, 17 (55%) participants of the target population (31) were male compared to 14 (45%) females. This represents a disproportionate representation of participants skewed against females, in leadership roles (Wehner et al. 2015) within the South Western Sydney LHD, which is different from the generalised healthcare setting. The gender gap has been the subject of several studies (Bismark et al. 2015; Cheryan 2012; Holman, Stuart-Fox & Hauser 2018; Kalaitzi et al. 2017; Mann, A & DiPrete 2013; Qamar et al. 2020; Shen 2013), and there appears to be hardly a definitive reason to attribute this disparity. Shen (2013) found that unconscious biases; pay disparity between genders and family values contributed to this gender disparity. Pay disparity between genders would not be an applicable factor in the NSW healthcare context due to the labour laws, which prevent this disparity, particularly, the unionised nature of public healthcare where awards determine the remuneration and other terms and conditions of employment for healthcare workers.

While the skewness in favour of males was prevalent in the UAE, according to Hallward and Bekdash Muellers (2020), and in Pakistan (Sarwar & Imran 2019), this could not be said of the NSW healthcare context. According to Australian data from the Workplace Gender Equality Agency (2019) there is a large female workforce of approximately 79% compared to males, that dominates healthcare and social services industries. This is contrary to representation in the banking sector which is male dominated according to Asrar-ul-Haq and Anjum (2020). Data from Workplace Gender Equality Agency (2019) pointed out that females are underrepresented at manager and above level across the healthcare and social services industries, at approximately 36.3% in 2018. This study, undertaken in medical imaging, was inconsistent with regards to general workforce numbers in healthcare, with the WGEA⁸ data, which is a new finding in medical imaging.

⁸ Workplace Gender Equality Agency

However, the data from this research supports the WGEA data, and studies by Burton (2019); Hempenstall, Tomlinson and Bismark (2019); Mousa et al. (2021); See, Lussier and Jones (2021); and Teede (2019) that also found significant under-representation of women in clinical leadership roles in healthcare. This study found that there was female underrepresentation at the leadership level within medical imaging, an area with scant previous studies. The underrepresentation of women leaders within medical imaging may be down to two main reasons in SWSLHD: family values and lifestyle choices and organisational support. These are discussed in the following sections.

5.3.1 Family Values and Lifestyle Choices

Hempenstall, Tomlinson and Bismark (2019); Ovseiko et al. (2017) and Wang, Mt and Degol (2017) found the number of women in science and engineering are growing, yet men continue to outnumber women, especially at the upper levels of these professions. Family values played a pivotal role with women more willing to make occupational sacrifices than men for family reasons (Botella et al. 2019; Hedley 2021). Leadership roles by their nature have significant responsibilities which are quite demanding in terms of time, that may become challenging for some women who intend to balance family commitments and their professional responsibilities (Mousa et al. 2021; Weale, Wells & Oakman 2019).

Women were more likely to consider more senior roles once their children were older and more independent (Alqahtani 2019; Athanasopoulou et al. 2018). On the other hand, men are able to utilise the support provided by their spouses to further their careers. Kena et al. (2015) and Laver et al. (2018) found that in academia, more women were vacating tenure tracking positions in favour of flexible part-time position, which contributed to their underrepresentation at more senior levels within the organisation. It may be concluded that these disparities may be addressed by more interventions and support for women to take up leadership roles in medicine (Holliday et al. 2018; Lewiss et al. 2020; Shaikh et al. 2019). The next section discusses the implications of organisation support on women in leadership roles in medical imaging in SWSLHD.

5.3.2 Organisational Support

Some research participants expressed organisational support was crucial to them undertaking leadership roles within SWSLHD. Such support included active encouragement and opportunities targeted at women from the LHD and deliberate actions such as mentoring which increased the capacity of women to take on more leadership roles (Laver et al. 2018; Stephens et al. 2020; Weigel, Kubik-Huch & Gebhard 2020). Other available support within the award structure, for example, parental leave provisions in healthcare with particularly flexible working arrangements until the youngest child turned five years, were helpful in increasing the participation of women at senior organisational levels. Flexible working arrangements such as remote reporting, were particularly valuable for radiologists. However, such flexible working arrangements were not always practical for all workgroups. For example, clinical managers were sometimes required to be hands on to resolve clinical issues in real time and onsite. Similarly, administration managers were also required to manage issues as they emerged, such as daily staffing.

Organisational support can also be in the form of flexible working arrangements such as working from home. More recently, since 2020, during the Covid 19 outbreak SWSLHD has introduced for some workers. Where these are in place, it is crucial in healthcare that adequate backup onsite cover arrangements are in place to manage contingencies. For example, in medical imaging, there are offsite remote reporting arrangements that accommodated work afterhours or remote health services. However, this is not always practical for some teaching hospitals where clinical leaders may be required to be available onsite for supervision of junior doctors. Where women have chosen to exercise remote reporting options, they may be disadvantaged as they are unable to participate in decision making in real time. This is alluded to, during afterhours or for faceto-face clinical services, which are still dominant. Therefore, the organisational support provided, needs to consider the factors outlined above, such as onsite clinical arrangements through registrars, to ensure that any support offered to women to promote their participation in leadership roles is meaningful. The next sections discuss the types of projects found in medical imaging and the leadership styles found in medical imaging.

5.3.3 Project Types Found in Medical Imaging

While over a quarter of the projects were Business Process Redesign projects, more than 69 % of the projects involved multiple stakeholders. Siegal, Stratchko and DeRoo (2017) found that radiology played a significant part in the diagnosis of disease. This reinforces the researcher's observations that medical imaging affects clinical service delivery in several ways. For such a setting there is a requirement for a robust system for knowledge retention within the department which can then be shared by similar entities within the same district.

This research found that there are consistent leadership styles utilised by medical imaging leaders, which is a new finding in medical imaging. The homogeneity on one hand allows for alignment in the attainment of organisational goals, and on the other hand, fosters a culture where ideas and processes are not challenged. This perpetuates a system where improvements cannot be undertaken, and leaders do not challenge each other or each other's findings. Such findings are unsurprising however, as the department heads in medical imaging are medical doctors. Given their dominance in the clinical settings, as discussed in section 2.5, it may be expected that this would be the case. This research also found that the leadership styles adopted by medical imaging leaders remained relatively constant over the research period, again a new finding in medical imaging. This finding does not support the study by Goleman (2017) who suggested that successful leaders practice at least four leadership styles.

This research found that 75% of the participants viewed themselves as hybrid leaders. While the concept of hybrid leaders was explored in section 2.3, Frich and Spehar (2018) ; Maile, McKimm and Till (2019) and Shaw, G et al. (2019), noted that hybrid leaders existed in the form of doctor managers. This research extends the work of Mak et al. (2019) and proposes that there are also hybrid leaders for other clinical specialties such as medical imaging and other allied health services, such as social work, physiotherapy or occupational therapy.

5.3.4 Communication Styles That Work

This research found a combination of communication styles were effective for disseminating information to team members with the use of e-mail; meetings; face to face communication and telephone calls being used to good use in projects. Such findings represent new knowledge in project management and change management in medical imaging.

Newer social media techniques such as SMSs, WeChat, WhatsApp and Facebook were not generally employed. Perhaps this is due to the reluctance of healthcare leaders and the industry, to adopt these technologies widely due to confidentiality of medical records and the management of patient data. Meetings emerged as the dominant communication tool during projects, utilised in all projects undertaken. E-mail was employed as a communication tool by most leaders, to reinforce discussions held in meetings. It also emerged that the leaders interviewed adopted a collaborative communication style to engage their audiences. For example, participants engaged with their audiences and determined their preferred mode of communication. They checked in regularly with their audiences, to determine whether those preferences were still valid during the currency of the project. If communication preferences were no longer valid, other mutually agreed communication modes were utilised such as e-mail or text messaging. This collaborative communication was crucial in maintaining engagement with participants and stakeholders throughout the project.

The researcher also found that there is usually a disconnect in views regarding operations between the different organisational levels that may be caused by the lack of a shared vision or miscommunication. For example, operations staff were clear about the need for additional staff to improve workflow or safety. However, the LHD and hospital executives had too many other competing priorities and budgetary constraints to prioritise the requests. This was found to be frustrating for operations staff who suggested that action could only be taken when an adverse event occurred. This finding is important in medical imaging and healthcare from an organisational alignment perspective. To achieve organisational success, there is a need for organisational goal alignment. When groups of staff perceive that senior leadership teams are apathetic to the issues being raised, then they are likely to disengage in the change processes and other organisation wide initiatives they do not perceive to be value adding to their work or patients.

In summary, section 5.3, presented and discussed an overview of the research findings. It was noted that there was underrepresentation of female medical leaders despite women outnumbering men in other specialist health roles, due to family values and lack of organisational support. It was also found that the

majority of the projects in medical imaging were business process redesign projects which involved multiple stakeholders. Finally, this research found that the use of multiple communication techniques as determined by the stakeholders were crucial to project success. The next section presents and discusses the dominant leadership styles found in medical imaging.

5.4 Dominant Leadership Styles in Medical Imaging Departments

A new finding of this research in medical imaging, was that the dominant leadership style for leading change management was consultative leadership. Situational, transformational, and transactional leadership were the other styles used successfully to implement change in medical imaging.

While this researcher was expecting authentic leadership, to feature as a dominant leadership style, this was interestingly, not the case in medical imaging. Similarly, Perez (2021) found that clinical leaders could succeed in healthcare if they adopted transformational leadership when promoted from clinicians to clinical leaders. Within professions such as healthcare that involve social welfare interventions, it may be argued that people who work within these settings do so from a caring perspective, such that they undertake healthcare professions mostly from a perspective of wanting to make a difference in others' lives. Most participants interviewed expressed through the interviews that their personal values, "to treat people with respect", influenced their interactions with patients and colleagues. Therefore, leaning more on their strong moral compass would lend more leaders to be authentic.

Consultative leadership on the other hand has had little research as an alternative style that could be utilised effectively in a healthcare setting, particularly with medical imaging leaders. Most leaders saw themselves as practicing a consultative leadership style in leading change as opposed to or in addition to transformational leaders during the semi-structured interviews. They viewed themselves as consultative leaders. Some researchers (Cheng et al. 2016; Perez 2021; Specchia et al. 2021) advocate for transformational leadership which they suggest increased motivation to positively contribute to organisational success and authentic leadership which they argued supported healthcare leaders in developing more confidence in effectively discharging their roles.

The consultation process can be noted as being crucial in medical imaging from several fronts. Firstly, the medical imaging subspecialty is referral driven in both private and public sector settings. When radiologists need to implement changes in their departments, they must consult with their referrers (or referral partners) who are other medical doctors or clinicians ordering tests), on the implications of the proposed changes prior to implementation. This then builds a culture of consultation which radiology requires as it has evolved over the past 20 or so years from the days of x-rays to CTs and Interventional Radiology (Brant & CA 2012). The British Society of Interventional Radiology (2020) outlined the advantages of interventional radiology as being minimally invasive, requiring small incisions to provide care; ability to undertake most procedures under local anaesthetic and targeted intervention which positively impact on recovery times and therefore minimal lost work time for recovery. Such prevalence of interventional radiology has seen the cases performed by the LHD grow from approximately 100 cases per annum in 2010 to over 650 cases per year in 2020 (Lin 2021).

Secondly, radiology as a subspecialty is driven by Radiologists and Nuclear Medicine Physicians as experts in their fields. This expertise ensures that they develop power and authority within their work setting and influence the culture and hierarchical nature of their practice, like other medical disciplines (Willis 2020). As the culture of consultation develops, it may be suggested that this influences the organisational culture and consequently the leadership style adopted by leaders within this setting. Darling (2017) described organisational culture as a set of beliefs, shared values and norms affecting the way employees feel, think and behave in the workplace. Within the organisational culture subcultures exist. Bordenave (2017) and Coleman (2017) suggested that an organisational subculture results from groups of organisational members sharing distinct values, beliefs and assumptions that differ from the dominant culture of the organisation yet are compatible. The reason that there are medical imaging staff in SWSLHD's subculture is because of their recognition that certain practices yield results in their settings and utilise these practices unconsciously.

Some participants suggested that their leadership style is impacted by other factors, such as the LMX theory as discussed in section 2.6.5. For example, participant 7 who is a non-clinician manager suggested that her leadership style is predominantly situational. She also suggested that her leadership style is influenced by the senior leadership team within organisation. For example, if the LHD or hospital was led by a directive leader or one with an authoritarian approach, such an approach tended to cascade to other leaders within this participant's team. This, as suggested by participant 7, would have implications on how the rest of the leaders will lead the organisation.

The experience of participant 7 as a middle manager is she is often faced with the dichotomy for middle managers, where they get caught in between the directions of senior staff and the staff they lead, having to balance the needs of their bosses and the staff they lead. Participant 7 suggested that as a middle manager, she manages both the expectations of senior leaders and her team through interpretation of the messaging from the former and delivering this message to the latter in an appropriate manner. While this is common within complex workplaces, particularly matrix organisations, skillsets are required for this process. A less experienced manager would find it challenging to manage upwards and downwards effectively, a skill that is done better by more experienced leaders. Participant interviews and anecdotal evidence has shown that reasons for the challenges faced by less experienced leaders may include; an inability to have difficult conversations; their inability to grasp the key deliverables and engage with their teams to deliver this, and inexperience in managing difficult team members.

Several studies, (Fletcher, Friedman & Piedimonte 2019; Głód 2018; Zuraik & Kelly 2018) reviewed transformational leadership in healthcare and viewed it as the most appropriate type of leadership. Boutcher (2019); deJong (2018); Krepia et al. (2018) and Mafora (2020) reviewed transformational change in a general health care setting and discussed its importance in improving health care quality and safety. This study supports these research findings, that transformational leadership was a common leadership style in healthcare. Also in partial support of these findings, Emere, Aigbavboa and Thwala (2018) found that democratic, transformational and transactional leadership styles were the leading styles utilised in the construction industry. Key differences between the construction industry and healthcare, are the nature of the service provided and the professionalism in the former sector. In the construction industry professionals such as architects design and plan the service to be provided, building and project managers are tasked with delivering these projects on time and within budget. On the contrary, medical doctors and other clinicians, are primarily concerned with delivering safe and quality healthcare to patients, while cost is a consideration, as discussed in section 1.2, quality and safety of care is considered equally important. The level of professionalism in healthcare, is significantly higher, in healthcare than construction, with healthcare providers belonging to a professional organisation, such as Australian Medical Association, or training colleges such as the Royal Australian and New Zealand College of Radiology (RANZCR).

Asrar-ul-Haq and Kuchinke (2016) also found that transformational leaders help individuals to adopt organisational change. In academia at university libraries, Mayowa-Adebara (2018) and Okere and Olorunfemi (2018) found that transformational leadership was prevalent with some laissez-fairer style evident. Academia and healthcare are somewhat similar in terms of professionalism, with most employees within both professions being highly skilled and professional.

Other confounders to leadership styles were found to exist within this study, for example, participant 15 suggested that the skill level of subordinates has an impact on the leadership style of the leader. An example for this scenario would be where a Chief Radiographer in a radiology department leads recent graduates, who are relatively inexperienced in radiology practices. For this leader, the performance expectations for recent graduates would be far lower than that of more experienced radiographers who require less monitoring, coaching and mentoring. Consequently, the Chief Radiographer's leadership approach will require adjustment, with a skewness towards mentoring, coaching and nurturing staff required. On the other end of the spectrum, the Chief Radiographer would expect a far superior performance level from experienced or knowledgeable staff.

Thus far, a discussion on the dominant leadership styles has been undertaken. One of the questions this research sought to answer was how the leadership styles employed in healthcare impact on change management outcomes in medical imaging departments. A discussion on this, follows.

5.5 Leadership Styles and Change Management Outcomes

Zhao, Hwang and Lee (2016) found that the project managers' leadership style was key success factor for projects. This research found that the leadership style utilised by the leader positively influences the outcome of change management processes. Success was reported by 96% of participants who utilised a consultative leadership style; 82% who used transformational leadership and by 82% who utilised transactional leadership. This research supports Goleman (2017) who suggested that use of multiple styles, at least four leadership styles, resulted in more positive outcomes than the use of a single style. This study also supports Aga, Noorderhaven and Vallejo (2016) and Larsson et al. (2015) who found that leadership style is a critical success factor influencing project performance in a civil engineering setting.

In this research a sizeable number of projects undertaken were business process redesign projects which impacted on multiple stakeholders. Crucially, in this clinical setting it was also identified that consultation was critical, more so in a hierarchical healthcare system with clinicians, higher up in the hierarchy. For example, medical imaging departments have a physician as a department head, as opposed to other clinical specialities. The use of consultation is supported by the General Medical Council (2022) who suggested in its guide that all doctors be willing to work with others to maintain and improve performance and respect the leadership and management roles of others.

It appears change that has not followed an appropriate consultation process can be fraught with failure risk due to apathy. In clinical settings, participant 9 noted that usually the lead clinician dictates the processes/procedures that they would like their patients to receive, even if the lead clinician is not the expert for a procedure. This is problematic where the referral system works. For example, an intensivist may dictate to a radiologist what sort of test a patient may require. If there is no good communication, or a lack of respect between the specialists, then the treatment provided may not be the appropriate one for that patient as requested by the lead clinician.

While the consultation style was shown to be effective in medical imaging, in clinical practice, radiologists often work with the available information at the time, usually on the paper referral form for some outpatients, or electronic medical systems on the imaging request form for inpatients. As clinicians, they rely on other clinicians to provide an adequate clinical history for them to have a complete understanding of the patient's condition and therefore consider all relevant facts to provide an imaging report. This report and images can then be utilised by the referring specialists or others in the provision of appropriate clinical care. If radiologists are provided with suboptimal information, then they also produce suboptimal reports which impact on the quality of care provided to patients, which may lead to adverse patient outcomes.

The transformational and transactional styles were also found to be effective leadership styles in leading change in medical imaging settings, this research has considered the possibility that the assessment tool used to measure the presence or absence of this style may be limiting the outcomes of the leadership styles. For example, the measurement tools only measure the former styles and not a wider range of leadership styles. The semi structured interviews provided participants with an opportunity to self-nominate a leadership style they believe is the best representation of their leadership style. However, this research supports studies by Aga, Noorderhaven and Vallejo (2016); Breevaart and Zacher (2019); Fletcher, Friedman and Piedimonte (2019) and Perez (2021) and with respect to transformational leadership as a style suitable for leading change in healthcare. Furthermore, this research supports Ghani et al. (2018) who found a correlation between transformational and transactional leadership and

employee engagement and thus effective in leading change within the healthcare setting.

Section 5.5 discussed that leadership styles influenced the outcomes in change management projects in medical imaging, with consultative leadership being dominant, and transformation and transactional leadership styles also positively influencing change projects' outcomes. This is crucial for healthcare leaders particularly in medical imaging because medical imaging leaders seek to optimise change management success. As consultative leadership style was found to be a critical factor in maximising change project success, as found in this research, medical imaging leaders would need to adopt leadership approaches such as consultative leadership, that improves the chances of project change management success. The next section discusses the strategies used for successful change management in medical imaging.

5.6 Strategies Used for Successful Change Management

This research found that several strategies were utilised and led to successful change management. These strategies include consultation and engagement; workload management and explained communication and are discussed in the next sections.

5.6.1 Consultation and Engagement

This research found that consultation and engagement was the most prominent strategy utilised in leading change projects. These findings are consistent with the work of Geerts, Goodall and Agius (2020); Savage et al. (2020) and Spurgeon et al. (2017) that concluded that the engagement of clinicians was necessary to bring about changes within hospitals and primary care practices. This was also supported by participants in this research who stated that, they found consultation being very important to succeed as their followers felt included in the decision-making process.

Consultation and engagement within the healthcare setting is challenging but important as different specialties have different competing interests. An example of this may be that of an interesting case which can be managed by either a vascular surgeon or interventional radiologist. If there is a lack of consultation between these two specialities, there may be competition between both specialists that may result in negative consequences to patient care, safety and quality.

Also, Specialists; Visiting Medical Officers (VMOs); Fellows and Registrars all have competing interests and priorities, which may impact on meeting organisational goals. In medical imaging for example, Fellows and Registrars who are still undergoing their training, are primarily interested in meeting their curriculum requirements in terms of both the number of cases performed and number of hours in service, among other requirements (Sydney Local Health District 2017).⁹ Visiting Medical Officers and Specialists on the other hand, are qualified and usually experienced, may be interested in perfecting their technique, achieving better outcomes for their patients and consequently building their reputation as experts in this field. VMOs and (Staff) Specialists are also remunerated differently, with the former paid on either a fee for service or sessional basis and the latter on a salary. These interests and priorities impact on perceptions to consultation and engagement. For example, in metropolitan radiology departments, a VMO would typically be assigned a predefined number of cases to report on and is typically not a permanent staff member. A Staff Specialist on the other hand, is usually permanently contracted and available for additional initiatives such as registrar training, multi-disciplinary team meetings and morbidity and mortality meetings. However, these medical roles are critical

⁹ Sydney Local Health District

and collectively play a critical role in the provision of clinical care as they allow medical coverage to be provided in different healthcare settings. Like every other profession, people undertaking these roles need to be managed by appropriate governance structures.

Consultation and engagement is important as even though organisations are not entirely project-based per se, temporary organisations within them exist in the form of assignments, task forces, programs, and so forth (Hassan, Bashir & Abbas 2017). Also, according to Hass (2011), leaders of complex teams require an understanding of the team development dynamics, how teams work so that they develop specialised skills for building and sustaining high performance.

5.6.2 Effective Leadership

Within medical imaging, consultation and engagement meant meeting with diverse stakeholders. This may require effective leadership to manage the divergent views and making compromises where possible to ensure that processes/projects could be implemented. As discussed in section 2.3, effective leadership has been highlighted in the literature as a key requirement for change management. It is therefore important to explore these concepts from this perspective. Effective leadership has had little research to understand what it is and how it is enacted in a medical imaging setting. In this research, effective leadership is a concept that includes several facets or activities that result in the ability to successfully influence the outcomes of change projects or processes. Strong leadership can be encapsulated as a systematic process through which leadership activities are undertaken. Such leadership can influence the attainment of organisational goals through the management of competing demands and divergent views within a complex organisation. These activities include but are not limited to; stakeholder engagement and consultation; conflict resolution; use of simple and concise communication. This process is underpinned by a consistent, evidence informed, contemporary approach in planning and governance. As an example, this research found that strong leadership is leadership that exhibits the following characteristics.

- i. Is systematic (in considering the planning and governance framework)
- ii. Contemporary in nature and is evidence informed
- iii. Resolves conflict and
- iv. Achieves organisational goals.

Strong leadership as discussed here is different from general leadership, firstly through definition. Strong leadership was viewed by Colla et al. (2014) as being similar to effective leadership. This research found that effective leadership is defined by specific behaviour, such as consultation and engagement of stakeholders to consistently achieve successful outcomes. Secondly, effective leadership is exhibited through specific characteristics or traits such as being systematic, considering the latest evidence informed practice in resolving conflict and or achievement of organisational goals. Some leadership styles, such as laissez faire leadership do not necessarily attempt to resolve conflict. In conclusion, effective leadership was found to be leadership that results in successful change management. This research has identified the description of what strong leadership is and is new knowledge in healthcare generally and medical imaging specifically. It clearly explains that strong leadership is leadership that yields results in successful change management. The next section discusses workload management as another crucial factor in achieving project success.

5.6.3 Workload Management

This research also found that workload management was crucial in ensuring project success. This finding is analogous to Williams, Perillo and Brown (2015) who found that workload was a barrier extensively reported on with regards to the implementation of evidence based practice. This was both on the part of

participants and the followers in these projects. The participants within this research reviewed the workloads of their staff to ensure that any additional activities could be completed within the time constraints. Where output was previously unsatisfactory, either through high workload, high expectations or a combination of factors, participants found that resetting expectations resulted in a new understanding and clarity on the required output.

Workload was cited as a key consideration in the medical imaging setting. If not managed well, it has a detrimental effect on the safety, quality and care provided. Medical imaging directly influences the patient care provided. For example, the pathology from medical imaging studies, will inform the care plan for each patient. Inaccurate studies or reporting therefore has a direct impact on patient care and safety. One of the management functions within the medical imaging department are quality control activities (Siegal, Stratchko & DeRoo 2017). These are conducted through a peer review of a sample of studies reported by the Radiologist or Nuclear Medicine specialist. The error rate is required to be within a certain threshold to satisfy good quality care requirements. Deviant performance such as missed pathology or inaccurate interpretation of studies can thus be managed appropriately. This research found that physicians working within SWSLHD generally agree that a high workload environment exerts pressure on them to produce timely reports. This time constraint therefore negatively affects the report quality, where the reports are found to contain errors, such as missing some pathology. Where reporting errors exist, such as a missed diagnosis, negatively impacting on patient safety and quality possibly including quite catastrophic consequences for patients.

Within an interventional radiology setting, procedures are guided by imaging and conducted in real time, that is, the interventional radiologist will be simultaneously undertaking the procedure. The imaging guides the intervention to be provided. In this case, errors like those found in diagnostic radiology are less prevalent. However, different risks emerge with interventional radiology, mostly related to equipment, skill level of the personnel involved, workload and the site of the intervention. For example, if the equipment is older or is of inferior quality, the images seen by the interventional radiologist are not ideal for optimal patient outcomes. Similarly, this research also found that less experienced interventional radiologists took longer to perform procedures, and outcomes varied between patients.

In administration or back-office functions, high workload also negatively impacts on revenue or the financial health of the department. While the LHD has an automated billing process, there is also some significant human intervention required. For example, the researcher's observations were that exceptions reporting occurs regularly on the automated interphase. This reporting was a daily automated report outlining discrepancies between Medicare accepted claims and those rejected. Administrators are then required to review and rectify these errors daily to ensure billing can proceed as normal to improve cashflow. If administration staff are time poor to perform these key tasks, then this is at the detriment of the organisation.

Workload management also involved the planning of tasks so that activities could be commenced earlier in the day for example. This could then mean that shift times could be managed differently. During this research, the LHD was addressing challenges to the service delivery model for some modalities. For example, services such as ultrasound, were offered during business hours only by some hospitals. However, the organisation had identified that a more generous service offering round the clock was critical to ensure patient safety and quality as Ultrasound was the only suitable imaging study for some patient cohorts, such as pregnant women. To manage the demand for these services various models were explored including extended operation hours or an on-call service.

5.6.4 Explained Communication

The research also found that explained communication/communication was another key strategy in leading change projects effectively. This finding parallels literature by Morgan (2020); Ojo (2019) and O'Connor (2013) who discussed the criticality of implementing a good communication strategy for change. Soomro et al. (2017) also found that effective communication improved acceptance levels and reduced resistance to change.

Communication clarity ensured that the diverse stakeholders understood their roles and responsibilities and could effectively discharge their tasks. Effective communication within medical imaging was found to be communication that had clarity; was simple; precise and done regularly ensuring that all parties were kept abreast of developments as they occurred. For example, participant 8 developed an e-mail template that met the criteria discussed above which was regularly dispatched to all key stakeholders to great effect.

Section 5.6 discussed the strategies utilised for successful change management within medical imaging. Consultation and engagement; workload management and explained communication were found to be key strategies that led to successful change management processes. In this section, effective leadership was also defined and discussed as leadership that led to successful change management. Typical characteristics for effective leadership were outlined as; systematic; contemporary; conflict resolution focused and achievement of organisational goals. This definition of strong leadership in medical imaging is one of the key scholarship contributions to be discussed further in chapter 6. The next section explores the reasons for why some projects succeed and others fail.

5.7 Why Do Some Projects/Processes Fail or Succeed?

Alvarenga et al. (2019); Gharouni Jafari and Noorzai (2021) and Vrchota et al. (2021) postulated that if different leadership styles are appropriate in organisational change projects, then they should also be appropriate for other non-change management projects as well. This research found that there are key reasons for project success or failure, in medical imaging, other than the leadership style employed and this section will discuss the top five reasons found. Participants discussed some of the complexity within the healthcare setting when they highlighted examples such as, the vested interests and egos in healthcare, where if everyone was not provided with an opportunity to provide their input, they would not be engaged with the process.

5.7.1 Stakeholder Engagement

These participants' sentiments are reflective of stakeholder engagement emerging as the prominent reason why change projects fail or succeed in medical imaging. This research supports the General Medical Council (2022) suggesting doctors have a responsibility to engage with their colleagues in undertaking their roles. Engagement involves the consultation of key internal and external stakeholders who could be working within a single site, multiple sites or external to the organisation. An example of this was a business process change involving the Emergency Department. In this project, it was established that another LHD was going to be involved in the change required, together with multiple other stakeholders such as the Information Services Team, Senior Clinicians of other departments, Clinical Leadership Teams of the imaging departments, among others. These are fully discussed in section 4.9.1 in the previous chapter. Seelen et al. (2020) and Weiss and Weissman (2020) supported early stakeholder engagement in the early planning and decision making by key stakeholders which is correspondent to this research findings in medical imaging. In this project there were multiple confounding factors which increased this project's complexity.

Firstly, there were different levels of hierarchy including the heads of medical imaging across the two LHDs, the two LHD information technology teams and other stakeholders already outlined. Importantly, the two LHDs had different approaches in their IT and clinical strategies on how to manage the proposed change, such as whether to wait for a state-wide PACS/RIS system or to implement a new system now, to address the current challenges. Additionally, from the Area Health Service era, discussed in section 1.2, there were legacy issues regarding the management of services still serving the two new LHDs that needed to be managed. A solution was therefore required that considered the requirements of these two distinct entities. A governance framework was established for this project that included escalation processes for the resolution of matters that were at a stalemate.

Secondly, medical imaging clinicians had significant power and influence in this project and were both keen to accommodate the requests from their peers, who saw the value of the proposed changes as it was evidence informed. The governance arrangements implemented outlined key issues such as the modes and frequency of communications; meeting arrangements; escalation processes; due dates for decisions and consequences of not meeting the due dates. Such a framework made this complexity manageable. During this project, a concept map was drawn up by the medical imaging team outlining the project scope and platform formed for the engagement of these stakeholders to effectively implement change. This project was a complex change project with significant

implications for clinicians across the LHD on how they ordered imaging studies and was implemented successfully.

Participants also highlighted the importance of engagement when asked about key learning from prior projects, with some suggesting change as a journey where one needed to take their followers with them to increase engagement.

While early engagement of key stakeholders was identified in round 1 interviews as a key success factor in project/process change success. Some participants suggested that when dealing with multiple stakeholders, there is often frustration that occurs due to the non-responsiveness of stakeholders. It is important for leaders to develop tools and strategies to manage this as they emerge. Tools that could be used by leaders include e-mail templates with options from which stakeholders can choose from. Strategies could include also include the following:

- i. Researching on stakeholders to understand how to end\gage with them effectively.
- ii. Correspondence with an end date for providing feedback. Stakeholders are advised that should information not be received by that date, the consensus responses provided to date could be used to inform change.

The research also found that stakeholder engagement was not a discrete activity conducted at the commencement of the project but was a regular ongoing process with the stakeholders involved. For example, some participants captured this when they suggested that they developed tools for use, such as document templates like action lists, disseminated regularly throughout the project. Such tools provided progress updates and key milestones to the LHD Executive, Heads of Departments of departments involved, such as Emergency and Intensive Care Unit. Stakeholder engagement also involved lobbying, a key part in leadership in hospitals, for example, getting many clinicians to agree with one that a project or service was key is crucial. Lobbying resulted in getting projects off the ground. This supported the work by Kelly, Tumblety and Sheron (2016) who found that medical practitioners who sought improvements in public health needed to recruit influential supporters, willing to listen to and weigh up the research available.

5.7.2 Planning

This research also found that planning was another prominent reason for project success in medical imaging. Planning ensures that activities can be undertaken in a systematic approach that leads to consideration of multiple factors in project implementation. This ultimately leads to project activities being undertaken in a structured approach, with potential bottlenecks identified and managed appropriately. This finding parallels Eder and Register (2014); Seelen et al. (2020) and Weiss and Weissman (2020) who found that transitioning to a new model required enough consideration on care improvements and advance planning to ease the transition, identify gaps and enhance workflows.

In this study context, planning spans from strategic planning for projects extending over a more than five years from conceptualisation to implementation; to business planning for pre, intra and post project implementation phases. Preproject planning involved several activities such as stakeholder identification, development of other sub-project plans like the communications plan; while intra-project planning involved activities such as; the scheduling of meetings, education and training activities; schedule of project facilitation activities. Finally, post-project planning involved activities to embed the change project into clinical practice through activities monitoring this. This research identified that project planning in medical imaging involved the development of tools that would be used during project implementation, such as weekly e-mail templates for regular information dissemination to all stakeholders.

5.7.3 Education and Training

This research found that education and training was another prominent reason why projects succeeded in medical imaging. The results of this study mirrors the work by Soomro et al. (2017) who found that education training and support of end-users in the implementation of a new system resulted in successful implementation and attainment of project objectives. Education and training in this context referred to the impartation of knowledge to staff as part of the change management process to ensure that staff had detailed knowledge of the proposed change, and implications. The education and training offered in the LHD considered several factors including staff availability; relevance of training for staff groupings; use of subject matter experts to conduct training, both from within and external to the organisation and training scheduling to ensure that those targeted by the training would be captured.

Education and Training offered does not need to be classroom type education and can be undertaken via several pathways including virtually. This study supports the view by Thomson III et al. (2016) that a training course on leadership change management could be offered virtually and was offered over a five-week duration with at least 1,500 participants enrolling from over 40 countries. More recently, others (Chatziralli et al. 2021; De Ponti et al. 2020; Ehrlich, McKenney & Elkbuli 2020; Vigersky et al. 2021) found that virtual training was as effective in delivering training as traditional tools.

5.7.4 Resetting Expectations

This research identified that resetting expectations was a further key reason for project success in medical imaging. A literature review conducted using the terms, "resetting expectations; expectations; resetting", found no existing literature to support this research finding. This finding is a new finding in medical imaging. When staff had been set in their ways, participants found that it was crucial to draw a line in the sand and outline the new expected standards of behaviour or outcomes. Where the expectations required resetting, participants also found that engagement with those affected was crucial in managing the change process as staff would have an opportunity to be heard and offer other alternatives that could be better than those proposed. Where participants could not find common ground with their stakeholders, participants found they could offer a platform to trial changes before they could be fully implemented. Some participants discussed this as an effective strategy when they suggested that before they made changes to existing processes, they provided staff with an adequate opportunity to offer feedback.

5.7.5 Processes and Procedures

This research found that the presence of processes and procedures to undertake the change process was a further crucial component to ensure success in medical imaging. For example, a communication plan was identified in section 4.9 as a tool used for the management of risk in communicating project related information. It is a document that outlines all the key stakeholders in the project and the mode of communication, type, and frequency of communication with key stakeholders. It augments the project plan itself or can be used as the main project planning document with the addition of further information. The processes and procedures provided a blueprint which could be used to run change projects and provided a template for this process. Scant prior research has been found in similar studies that outline that this is a contributor to project success. It was noted in section 4.10 that the key formal project documentation was meeting minutes with the absence of other formal documentation. This is a shortcoming in the documentation of projects in the LHD as knowledge retention is by the individual rather than the organisation. The data suggests that there is limited formal project management expertise in SWSLHD, as none of the participants indicated they undertook any formal project management training. This finding supports Aberese-Ako, Agyepong and van Dijk (2018); Brooke-Sumner et al. (2019); Cleary et al. (2018) and Nzinga, McGivern and English (2018) who found that although clinical leaders are experienced clinicians, they appear to have little management training or mentoring to increase their development as managers.

The LHD offers an in-house project management program known as Accelerated Implementation Methodology (AIM), which is a project management methodology adopted for the LHD for public health entities, by the Agency for Clinical Innovation. This short course [3-days] provides a structured and methodical approach in undertaking project management within the public healthcare settings. The LHD, part of NSW Health, has access to the centralised function of the Project Management Office (PMO). According to NSW Health (2018), the PMO supports strategic change and sustainability within the health system by the provision of expertise; connecting NSW Health organisations and sharing scales and enhancement of successful programs throughout NSW Health. SWSLHD could therefore access PMO support in undertaking projects or building capacity to do so. This would facilitate a more consistent approach in project management in the LHD and improve change project outcomes. This is further discussed in section 6.4.

Some entities within the private sector, such as healthcare consumable manufacturers, successfully undertake initiatives to streamline processes such as LEAN (Abuhejleh, Dulaimi & Ellahham 2016; Anderson, NC & Kovach 2014),

Kaizen (Shatrov et al. 2021; Suárez-Barraza & Miguel-Davila 2020) among others. Collaborations can be developed with the private sector in the development and use of other techniques in project management such as LEAN methodology. Synergies would be derived for both participants. For example, the private sector may have superior project delivery principles and techniques, while public healthcare organisations provide a complex platform for these techniques to be refined. Additionally, private sector organisations benefit by obtaining knowledge about the use and function of their products in public healthcare, which provides them with an opportunity to better meet the product needs of consumers.

5.7.6 Project Resourcing

This research found that project resourcing impacted on the successful project change management process in medical imaging. Pohjola et al. (2016) also found that no additional time and resources were afforded to healthcare leaders to undertake projects. Instead, leaders were expected to undertake projects in addition to their daily roles.

In this study, most leaders were not provided with specific designated project management time. They were expected to implement change projects within the existing resources. In this study, the lack of project resourcing did not negatively impact on project outcomes. However, participants reported that they often undertook projects outside work hours, with their passion for the projects being the main drivers. Two opposing views emanate from this. Medical imaging clinicians in the public health settings reported they are inherently busy with clinical work due to the increasing workload. As a result, they are time poor to carry out change projects well. Such change projects include the procurement and installation of major equipment. Where these projects become increasingly complex and time consuming, change projects led by clinicians may become challenging to successfully implement. This may be because of lack of enthusiasm from the clinical project lead; being time poor or a lack of project management skills among other reasons. This research supports Goldstein et al. (2021) findings that frontline clinicians are increasingly busy to undertake research.

An alternate view is that clinicians are keen to be involved in change management projects or processes that impact on their work. This is particularly important because most change management projects impact on the clinical work carried out by the clinicians and therefore their involvement as key stakeholders would be crucial. Kippist (2012) and Belrhiti, Van Belle and Criel (2021) found that medical doctors maintain their dominance over other clinical professions by excluding them and controlling their work. It is therefore in the interest of medical doctors to control, direct and be involved in change projects that impact on their work environment.

Several participants suggested heavy workloads seem to propose that people's leadership styles are impacted. Leaders tend to be more dictatorial, perhaps as coping mechanism to manage their time. This research showed where change projects were mandated by NSW Health, the leadership approach was more directive. Ghazzawi, Shoughari, and Osta (2017) suggest situational leaders were more directive when there was a lack of commitment or organisational goals were not being met. Anecdotal evidence suggests frontline clinicians are usually aware of the concerns they have with most existing processes, and it would be in their best interests to align change projects to their wants, hence their engagement is easier.

Section 5.7 above discussed the five prominent reasons why projects succeed or fail in medical imaging. Firstly, stakeholder engagement was found to be the most prominent reason for project success. Secondly planning was also found to be a key reason for project success. Thirdly, education and training were crucial in project success followed by resetting expectations as a key reason for project success. Finally, the presence of processes and procedures in medical imaging departments contributed to project success, negating the effects of lack of formal project documentation. The next section discusses how the leader's leadership style in medical imaging is influenced by prior or learned experience.

5.8 Leadership Style and Learned Experience

This study found that there were no significant changes to the leadership styles of the participants in medical imaging. This research supports work by Allen et al. (2018); Bell, BS et al. (2017); Keiser and Arthur Jr (2021) and Vogel et al. (2020) who suggest individuals do not automatically learn from experiences, and experiences do not automatically improve leader development. Homan et al. (2020); Larson and DeChurch (2020) and Uhl-Bien and Arena (2018); also suggest that learning from experience requires deliberate practice for behavioural reinforcement which has often been overlooked.

As leaders learn, they collaborate with their peers influencing both their individual learning and organisational learning. Barley, Treem and Kuhn (2018); Barros et al. (2020); Edmonstone (2017) and Martínez-Costa, Jiménez-Jiménez and Dine Rabeh (2019) found lessons learnt evolve to organisation learning for a far reaching impact in an organisation's capacity. In healthcare the concept of double loop learning is particularly important. Davies and Nutley (2018) and Abdullah et al. (2018) discussed double loop learning as occurring when organisations rethink basic goals, norms and paradigms. While SWSLHD attempts the process of double loop learning, it can maximise learning capacity initially by documenting individual learning and then utilising this to develop double loop and meta learning.

A new and interesting finding in medical imaging in SWSLHD was the research participants concurred the organisation does not document project methodology well. This research found the only form of documentation available was meeting minutes, action lists and e-mail correspondence regarding the project. It is an interesting finding in that, despite the lack of formal project documentation, projects carried out, were usually successful. This seems to suggest the leadership style and leader experience, played a key role in project success in medical imaging.

Project documentation could be added to include documents on project; scope, (what is in and what is out); timelines and implementation methodology. It could also include pre and post project implementation evaluation and improvement opportunities. This can then facilitate organisational learning as discussed in the preceding paragraph. This finding supports those of Day, DV and Thornton (2018) who discussed the 70/20/10 rule that 70% of leader development comes from onthe job experiences, 20% from developmental relationships and 10% from formal programs.

The concept of organisational learning starts with the individual and then cascades to the organisation through collaboration as outlined by David and Golan (2017) and Park and Kim (2018). However, this deficiency of project documentation was partly offset by the stability of the medical imaging workforce. During the research period, only one participant retired from the organisation reflecting the longevity of staff within this organisation. Staff longevity was found to be crucial in the retention of corporate knowledge in medical imaging, particularly with regards to future projects. The importance of documentation is clear, with some participants suggesting that people

remember history differently and will raise the same issues that were previously addressed, when it suits them.

This research also found that the nature of projects undertaken was homogenous and knowledge could be shared with practitioners undertaking similar projects in a relatively informal setting. There seems to be scant studies in Australia in medical imaging, on project homogeneity as a contributor to project success. As outlined earlier in section 4.11, medical imaging in SWSLHD enjoyed a relatively stable workforce that individually retained project knowledge. If similar projects were undertaken again in future, leaders would possess the necessary and relevant skillset to successfully lead projects. Medical imaging leaders could share their successes in leading change successfully through a variety of LHD wide mediums such as the newsletter; their intranet pages and other platforms such as hospital wide forums.

This research also found that participants modified their leadership style in a way they referred to as enhancements from previous change projects. This supports Brown, R and Abuatiq (2020) and Li, L et al. (2022) who conclude that leadership development is a continuous process. Participants found it challenging to find the right balance in documentation and suggested future projects could be easier as they now had templates and a formula that worked. This finding is not dissimilar to Abdulla, Alhashimi and Hamdan (2019); Irfan et al. (2021) and Khan et al. (2020) who suggested a project management approach at the beginning of the project and concluded that the right project approach was a critical success factor.

When participants were asked what key insights they had learnt from prior projects, some people highlighted collaboration and pointed out that in prior
projects where they collaborated poorly, it was a steep learning curve for them as they found that projects were challenging to implement. This demonstrates that leaders in medical imaging who modified their style and reshaped their style are more likely going to be effective leaders. For example, those participants that enhanced their leadership style to include more collaboration found that they could successfully undertake change projects in medical imaging.

Some research participants in this project also found being a new leader in a new place presents a good opportunity to lead for change. New leaders have a clean slate to implement new innovations, to which staff were more receptive, compared to an existing leader implementing new processes. For example, a participant pointed out that making roster changes and implementing reporting new workload management techniques was viewed more favourably by staff. This was because staff expected new changes to occur with the commencement of a new leader.

Section 5.8 above discussed how learned experience influences a leader's leadership style in medical imaging. This study found medical imaging leaders adjusted their leadership style because of learned experience. While participants reported that the organisation documented projects poorly, the workforce stability of the department had assisted the department in knowledge retention. The next section discusses how learning leaders fare in future projects.

5.9 How Do "Learning" Leaders Fare in Future Projects?

Buchanan and Badham (2020); Denhardt et al. (2018) and Stouten, Rousseau and De Cremer (2018) found that the emergent approach to change emphasises that change be perceived as a continuous, open-ended process of adaptation to changing circumstances and conditions rather than as a series of linear events within a given period. This therefore requires individuals and organisations to learn from prior projects. This research found that "learning" leaders achieve better outcomes in successive change management processes. This is a new finding in medical imaging as there is scant literature on the topic of learning leaders and change management in medical imaging. For example, participants 2; 5 and 14 found that experience from prior projects resulted in them modifying aspects of their leadership styles resulting in better outcomes for their projects. These participants highlighted that they found that wide consultation and early engagement were crucial in project success and modified their leadership styles. These findings support studies from Busari et al. (2020); Cameron and Green (2019); and Saira, Mansoor and Ali (2020) who wrote that a key determinant of implementation success is learned ability for individuals to navigate change successfully. Haski-Leventhal (2020) and Trivedi (2021) postulated that change is so hard because change is required at an individual behaviour level, organisational structural level, to make transformation successful. Therefore, in medical imaging, learning leaders who embrace improvement opportunities from prior projects are more likely to succeed in subsequent similar projects.

Medical imaging leaders who adjusted their leadership styles from the first project to subsequent projects indicated prior learning was crucial for future projects. When asked what their key learning was, from previously leading a change project, some participants suggested engagement while others suggested collaboration. Such participant sentiments support the view that learning leaders achieve more positive outcomes than their opposite counterparts, as they both successfully led subsequent change projects, following their learning. Helm et al. (2020) and Van Rossum et al. (2016) found that organisational change is vulnerable to numerous mistakes and change managers should use explicit models with all the elements to enable them to succeed.

While the concept of learning leaders was explored in section 5.6, noteworthy was the fact that in medical imaging, change processes in most projects reviewed were not formally documented with project documentation. Corporate knowledge could not be appropriately retained by the organisation as this knowledge was leader based rather institutional knowledge. Experience or learning should occur at any time rather than leaders considering themselves as being set in their ways and not consider making any changes. Smith, RH (2015) postulated that organisational learning could begin at any time so that organisational benefits are realised. Bogers, Chesbrough and Moedas (2018); Müller, JM, Buliga and Voigt (2021) and Papa et al. (2018) found that employees needed to harness external and internal knowledge to improve processes and products, which according to Campanella et al. (2017) and Vrontis et al. (2017) led to competitive advantage to an organisation.

All participants interviewed for this research indicated that they had not undertaken any formal project management or leadership training with the training offered discussed in section 5.4. This finding supports Lexa and Fessell (2019) who recognised that leadership is best learned by a combination of experiential learning (learning by doing) and some formal instruction. However, Lexa and Fessell (2019) suggested that what most Radiologists need was to take on projects or roles during their career where they could do the work of leading, such as opportunities to work on quality and safety projects. This could also be said of non-radiologists as this research found that this group also lacked formal leadership training and acquired their skills through experience.

However, noteworthy is that research participants have been utilising their experience, albeit successfully, in the LHD to conduct projects. The participants in this study were employed by the LHD for the duration of the study. This stability in the medical imaging workforce has contributed to the retention of skills and experience in project management in individuals in medical imaging. The organisation could improve on the current individual-centric skills retention and put in place mechanisms to retain this corporate knowledge. Skills retention in a stable workforce not only improve project success in the LHD and lead to better patient outcomes through the retention of project management and clinical skills in medical imaging. This reinforces the earlier finding that good communication was critical to project success. The lack of mechanisms for organisation wide knowledge retention becomes even more critical, so that SWSLHD relies on multiple strategies for project success rather than over reliance on communication.

Section 5.3 described the dominant leadership styles in medical imaging and how they are enacted. In this research, it was found that for earlier projects (round 1), participants self-reported a higher level of consultative leadership. However, a small number of participants commented on minor adaptation of their natural leadership styles to be more effective leaders. The next section provides a conclusion to this chapter.

5.10 Conclusion

This chapter discussed the dominant leadership styles of consultative; situational; transformational and transactional as appropriate for leading change in healthcare, particularly in medical imaging services due to it being the focus of this research. The study found that consultation and engagement; workload management; education and training; explained communication and planning were effective strategies in leading change effectively in healthcare.

The study also found that other than the leadership style employed by the leaders, stakeholder engagement; planning; education and training; resetting expectations and the presence of processes and procedures were crucial reasons why change projects were successfully conducted.

This chapter discussed how the leader's leadership style is influenced and reshaped by the leader's learned experience from prior projects. Medical imaging leaders were found to be learning leaders who adopted their leadership style from prior projects. This led to medical imaging leaders achieving more positive outcomes in their change projects compared to their peers, in other disciplines, who did not carry forward learning from prior projects.

The following chapter, Chapter 6, will conclude on the research findings, scholarly contribution, implications for theory and practice, as well as propose further research areas.

Chapter 6 – Contribution, Conclusions, and Implications

6.1 Introduction

Chapter 5 discussed this research study's findings. In this discussion, it highlighted the new findings of this research and outlined comparisons to other studies where appropriate. This study investigated how leadership in healthcare affects the outcomes of the change management process in medical imaging departments. This was conducted through addressing the six research questions as outlined in section 1.3.

Building on chapter 5, this chapter, summarises the key research findings. It also outlines articulates the key contributions to scholarly leadership theory. One of the key aims of this research was to understand and inform policy and practice in medical imaging. A section of this chapter is dedicated for this purpose recommending leadership practices in successfully implementing change management processes.

This chapter will also outline the assumptions, limitations, and weaknesses of this research and offer potential future research areas.

6.2 Research Conclusions

6.2.1 Dominant Leadership Styles

This study found that the dominant leadership style in medical imaging was consultative with situational, transformational, and transactional leadership

styles being other prominent leadership styles leaders displayed. Participant 13 summed this by suggesting that they consult their staff for any changes in their department. Consultative leadership was identified as being a critical success factor in building and maintaining ongoing relationships with various stakeholders. In the LHD studied, the medical imaging departments were found to have a stable workforce, demonstrated by the longevity of staff within their departments and participation throughout the longitudinal project. It could be argued that these relationships fostered the uptake of new ideas, made colleagues more willing to engage in change projects which resulted in a better contextual understanding of the intended change process, making the case for change stronger.

The consultative leadership style was found to be the most dominant style for several reasons. The researcher noted several observations during the research process which include:

- i. Medical imaging is a subset of healthcare where referrers are crucial to the service. Radiologists are legislatively and professionally not permitted to self-refer patients for studies. They rely on other referring partners for ongoing inpatient and outpatient work. As discussed in section 1.1.6, inpatients are funded through the ABF process while outpatient work is funded by Medicare. Radiologists therefore must develop relationships with their referrers and hone their consultation skills, so that they receive referrals from other clinicians, since they cannot self-refer patients. Such relationship building could be in the form of offering education to their referring partners on medical imaging procedures, their common indications and appropriate imaging studies ordering.
- ii. Consultative leadership was found to achieve the most successful outcomes in medical imaging and should be utilised where possible to

optimise change project success. Consultative leaders demonstrated early stakeholder engagement; collaboration and engagement; planning and good communication skills which made them effective leaders of change management projects. The effectiveness of this leadership style in change management, means that change projects can be successfully implemented in a healthcare setting, with the aim of reducing the 70% failure rate of general change management projects, (as was identified in chapter 2). As consultative leadership is a key leadership style for successful change management, it is suggested that core training for medical imaging healthcare professionals such as radiographers and radiologists include leadership training, particularly training in consultative leadership. This training could be included in their basic profession training.

This study also found how consultative leadership was enacted by medical imaging leaders to influence the outcomes of change management processes. The next section discusses how leadership was undertaken in change management projects in medical imaging.

6.2.2 Strategies Used for Successful Change Management

Medical imaging leaders used several key strategies to successfully lead change in their departments. The top three strategies were: consultation and engagement; workload management; and education and training, which are now discussed.

6.2.2.1 Consultation and Engagement

Consultation and engagement were a deliberate process undertaken by medical imaging leaders in the change process. This was out of necessity, as discussed

in section 6.2.1, or as a key tool in successfully leading change due to the selfreferral prohibition. As identified in section 4.2.2.2, there were several stakeholders directly and indirectly involved in the provision of patient care, such as wardsman: paramedics and information technology teams. The involvement of such stakeholders was crucial to project success hence consultation became a prerequisite for project success.

To effectively succeed in change management, medical imaging leaders adopted robust communication strategies as fully discussed in section 5.6.4. Not only did they send e-mails, but they also took time to provide detailed explanations, which informed the communication with their key stakeholders. For example, in a project involving the replacement of a hospital's only CT machine, there was a great deal of planning required upfront and a communications plan outlining the nature and preference of communication to the various stakeholders was implemented. Some internal stakeholders required regular short, sharp e-mails, SMS and huddles. One participant developed an e-mail template that was utilised for regular stakeholder updates. This was both effective and satisfied the information exchange process requirements with these key stakeholders.

6.2.2.2 Workload Management

As identified in sections 1.1 and 1.2.7, medical imaging plays an increasingly important role in the diagnosis of diseases and meeting key performance metrics because it facilitates faster clinician decision making. Downtime of services is to be avoided or kept to an absolute minimum as it negatively impacts safe patient care, as patients cannot access imaging in a timely manner. For this to occur, medical imaging leaders utilised robust planning processes to ensure that minimum service disruptions were encountered ensuring that patients could continue accessing the healthcare they needed. For example, these planning processes included a risk-based assessment framework for effecting downtime in the installation of key infrastructure such as CT machines. This process included the timing and duration of work and alternative imaging solutions for hospital patients when equipment was being installed.

Medical imaging leaders highlighted a few key reasons why they were successful in their change management projects and processes. The researcher will now discuss the three main reasons of: stakeholder engagement, planning and education and training, in the following paragraphs.

Stakeholder engagement was found to be the single most important reason for change project success in medical imaging. It was recognised that key stakeholders, such as referring clinicians and other subject matter experts, were crucial to the change process. These key stakeholders were value adding to the change process because their participation impacted on project success and keeping them engaged and were crucial in achieving success in change projects.

To ensure that medical imaging services had very little interruptions meticulous planning was required. This ensured that processes were thought through including the change methodology along with a risk assessment of potential challenges and solutions. An example of this planning was seen in the implementation of IT type projects in medical imaging, such as the implementation of a new PACS/RIS system for the district. In IT projects, stakeholder engagement was one component. Another aspect was the end-to-end testing of possible scenarios, as a proof of concept in a training environment. Following success in a dummy environment, further tests were conducted in the production environment and finally in the full deployment process. Similar actions need to be undertaken for non-IT projects, as potential scenarios can be explored and potential outcomes evaluated prior to full project implementation.

6.2.2.3 Education and Training

Finally, the research found that education and training played an integral part in the success of change management projects in the LHD. Where new services were introduced in medical imaging in the LHD, all staff impacted by the change would undergo education and training in the use of the new products or in the implementation of the new process. For example, in a project involving the identification of ED CTs and their prioritisation in the PACS/RIS, stakeholders such as ED physicians; radiologists, radiographers and nursing staff had to undergo training in the CT ordering process and how to escalate studies to ensure that key performance benchmarks were met. This education process ensured the successful change management of projects within medical imaging.

Having investigated the strategies and reasons for project success, the next section discusses the nature of leadership and change management, the reasons why some projects fail and the learned behaviour of medical imaging leaders in successive change management processes.

6.2.3 Nature of Leadership and Change Management Processes

As discussed in sections 5.4 and 6.2, this research found that the dominant leadership style practiced in medical imaging was a consultative leadership style. Leaders who practiced a consultative leadership style tended to achieve success in their change projects because they identified and engaged with their key stakeholders early and did so well throughout their projects. This style was characterised using 3 key strategies as outlined in section 6.2.2 above of consultation and engagement; workload management; and education and training. This study also explored the reasons why projects failed or succeed in sections 5.7 and 6.2.4 below. Utilising these key strategies and incorporating key

reasons for project success may result in project success in medical imaging and other healthcare settings.

6.2.4 Why Do Some Projects/Processes Fail or Succeed?

Section 5.7 discussed this research question and summarised the key reasons for 3 key reasons project success as stakeholder engagement: planning and education and training in medical imaging. Project practitioners in medical imaging or healthcare, who adopt these enablers to project success are likely to achieve project success.

6.2.5 Learning Leaders, Organisations, and Project Outcomes

The stability of the medical imaging workforce over the research period allowed for the retention of corporate knowledge of change management within the LHD, despite the lack of formal documentation of project processes. As identified in section 3.2, this study was a longitudinal study which tracked leadership style changes over time. During this period, there were no significant changes to the workforce, with only a single retirement noted within the leadership ranks. This meant that the same leaders remained within the organisation during the research period and had the benefit of having participated in previous change projects or having access to others willing and available to share their knowledge in change management.

Such stability suggests knowledge retention among the medical imaging leaders and within the LHD was crucial to project success. It led to learning departments and a learning organisation as leaders could leverage off each other. Staff retention in medical imaging can be attributed to several reasons including the highly specialised nature of the various areas of radiology, nuclear medicine, and interventional radiology. Staff who work in these areas undergo specialist qualifications to practise in these areas. Following years of study, pathways exist for experienced employees and/or further education, to navigate the corporate ladder.

Section 5.3 of this study also discussed the demographic nature of there being more males than females in leadership ranks, consistent with Weigel, Kubik-Huch and Gebhard (2020); Kubik-Huch et al. (2020) and Qamar et al. (2020). Based on South Western Sydney Local Health District (2019), the region is multicultural with a third of the population speaking a language other than English at home. From the researcher's knowledge of the region, these demographics are represented in the general healthcare workforce, and it suggests working close to home may be a key factor in employee stability in the absence of a stressful work environment. As identified in chapter 1, wages and conditions of service are generally uniform across the NSW public health system. Working from home could therefore be an attractive proposition for this cohort of employees, which is becoming more prevalent in most work settings where onsite training or supervision requirements prevent this from occurring.

On the one hand the stability of the workforce is an asset to knowledge retention. However, on the other hand it may be viewed as a handicap to the medical imaging department. This is so as the leaders in medical imaging lack a heterogeneity in leadership styles, which may not promote divergent views and leadership approaches in the management of issues, processes, and projects.

The homogeneity of projects undertaken also allowed different leaders across different departments to share and transfer change management knowledge to their peers leading to successful change management in medical imaging. For example, section 1.2.8 identified that medical imaging is a capital and technological intensive service. This means that high value equipment is required to be maintained and replaced regularly to retain attractive Medicare rebates. For this reason, leaders must regularly undertake complex equipment procurement and replacement change projects. The stable workforce means leaders can call on their colleagues to assist and support them in change processes. The research data shows that leaders share learnings in change projects implementation.

Workforce stability and homogeneity of projects are key enablers of project success. Leaders who learnt from prior change management projects found that they achieved better outcomes in subsequent projects. Paradoxically, the workforce stability can be an inhibitor to knowledge and leadership renewal, which may be required by medical imaging to develop new ways of leading change projects. This is discussed further in section 6.4 below.

6.3 Scholarly Contribution

This study contributes to literature in a several areas. Firstly, according to the literature search conducted in chapter 2, for change management within the healthcare sector and medical imaging specifically, there appears to be scant literature, that explores the leadership styles in medical imaging departments in the public healthcare domain in Australia. Section 2.7 provided a discussion of leadership styles utilised in healthcare while section 5.4 outlines the dominant leadership styles in medical imaging and illustrating this contribution. This is unique in extending the leadership practices in healthcare generally to a more specific setting, medical imaging, which faces challenges in being a capital-intensive service and dependent on a multitude of stakeholders to effect change successfully. This study found that the dominant leadership style in medical imaging was consultative with situational, transformation and transactional leadership being the other prominent styles. Where these styles are utilised

consistently, they resulted in successful change management in healthcare settings in medical imaging.

While Boaz et al. (2015); Spurgeon et al. (2017); Tsai et al. (2015) and Denis and Van Gestel (2016) found that clinical engagement promotes a greater connection between the medical profession and the organisational environment. However, it may be argued that the consultative leadership in medical imaging leaders helps facilitate this engagement. It does so through the engagement process that occurs between senior healthcare leaders and medical leaders. For example, medical leaders often require executive project sponsorship for certain projects. As part of obtaining this approval, medical leaders need to canvas support from their colleagues through a consultative process before seeking executive approval.

Extending Spurgeon et al. (2017) who focused on medical leaders overall, this research suggests that medical imaging leaders can be viewed from the organisational environment context. As part of this organisational environment, medical imaging leaders are building stronger relationships with other clinicians through consultative leadership as a preferred leadership style in successful change management processes.

Secondly, previous literature has not adequately discussed how such effective leadership is enacted in medical imaging. The International Institute of Directors and Managers (2020) referred to strong leadership in terms of their five main results of higher employee retention; higher percentage of top performers; higher productivity; consistent growth and results. This research extends this knowledge by defining strong leadership by characterisation within medical imaging and how it is enacted. This research defines strong leadership as the ability to successfully implement change within an organisation. Such leaders achieve this through the mechanism of a systematic process though which leadership activities are undertaken. They influence the attainment of organisational goals through the management of competing demands and divergent views within a complex organisation. These activities included but were not limited to; stakeholder engagement and consultation; conflict resolution; use of concise and simple communication. This process is underpinned by a consistent, evidence informed, contemporary approach in planning and governance. Consistency is required in the leadership approach as it demonstrates that a process is repeatable, while evidence informed practice is crucial in clinical practice (Lavoie-Tremblay et al. 2012; Peirson et al. 2012; Strauss 2011; West et al. 2015). This appears to be because of clinicians preferring the "show and tell" approach which could be by way of case presentation in multidisciplinary team meetings, morbidity and mortality meetings or other teaching platforms.

Contemporary practices require the most current interventions. Effective leadership for example, can be demonstrated, in an equipment procurement and installation project, by the successful purchase of a piece equipment, that satisfies the requirements of the project brief, and has been procured within budget. It is installed with minimal service delays and disruptions to end users and the equipment has been tested, its use and fitness for purpose verified by the relevant project stakeholders. Following deployment, the equipment is functioning as expected. Leading such a project successfully requires leaders to demonstrate effective leadership skills such as planning and navigating potential conflict and therefore successfully effecting change.

Thirdly, the methodology and its application in this study was effective particularly the longitudinal nature of the study as it allowed the assessment of leadership in medical imaging of participants over a period, rather than at one point in time. There is scant literature demonstrating that this had been done in previous studies in medical imaging in NSW and Australia. The answers generated were more complete as participants had the opportunity to reflect upon the findings of data collection tools used. Following these reflections, participants appeared to be more prepared in successive data collection rounds.

Fourthly sections 5.5; 5.6 and 5.7 of this study extended change management literature by exploring the critical success factors for leading change in medical imaging departments. Practitioner literature such as Kruskal et al. (2012); Itri and Lawson (2016); Matalon et al. (2018); Phalak, Gerlach and Parikh (2020) & Shah et al. (2013) has not previously discussed management or leadership theories or critical success factors for change management in medical imaging departments. Thomson III et al. (2016) pointed out that medical imaging managers lack the skills to undertake change management. This study explored the critical success factors for leading change within medical imaging. These critical success factors include the strategies for change management, as discussed in section 5.6 and utilising the consultative leadership style as identified and discussed earlier in this chapter. If practitioners understand these factors and implement strategies suggested by this research, such as consultation and engagement; workload management and explained communication; they may be able to successfully implement change. This study has extended leadership knowledge by other researchers (Berghout et al. 2017; Lexa & Fessell 2019; Matalon et al. 2018; Phalak, Gerlach & Parikh 2020) by including healthcare, specifically medical imaging, an area with scant literature despite the significant regular change that occurs in Australia and New South Wales specifically.

Fifthly, this research extended the theory on learning leaders and organisations. Studies by Ayres (2012) and Kumaraswamy and Chitale (2012) in non-healthcare settings suggested that organisations and leaders must be learning organisations and leaders to successfully manage change. This study extends their work by defining their work within healthcare, specifically medical imaging as those that create or acquire knowledge from past change management processes and apply such learning to future settings, to improve the outcomes of later change management processes. As this study found medical imaging leaders who modified their leadership styles from learned experience at an organisational and individual level, leaders were likely to achieve greater success that those that did not learn from their experience.

Finally, this research provides new, practical applications for medical imaging leaders to successfully lead change within their settings. While previous studies have discussed leadership styles in other healthcare disciplines such as nursing and medical doctors, there is scant previous research that has discussed this for medical imaging leaders in Australia.

This study therefore advances leadership knowledge for a healthcare focused area of medical imaging which is important for practitioners which they may utilise to successfully lead change processes and projects.

6.4 Implications for Policy and Practice in Medical Imaging

6.4.1 Stakeholder Mapping

This research identified one of the key strategies for project success was stakeholder mapping, a critical planning process for leading projects as it facilitates the subsequent processes of consultation and engagement. It is therefore crucial for change managers to undertake a stakeholder mapping process as part of their project planning, so that early stakeholder consultation and engagement could be undertaken to optimise project outcomes. This stakeholder mapping process could be undertaken by engaging subject matter experts in the area where change is required. Discussions with such subject matter experts about intended projects will assist in the identification of those stakeholders that may require consultation for the project in question.

6.4.2 Project Documentation

The LHD and similar organisations who develop a standardised process for documenting projects will ensure that corporate knowledge is stored in a repository where others within the organisation can access and utilise it. This process could be done through the preparation of a suite of "exemplar" project documents to be maintained for specific projects. Additionally, as part of the project approval process, the approving executive can request for information on the project documentation to be kept, with those projects not conforming to those requirements not approved. Another way this could be achieved is the identification of project champions within departments who are trained in project documentation to support departments in undertaking projects. It is therefore suggested that leaders develop formal documentation processes for knowledge transfer by attrition from one leader to another.

The LHD utilises an electronic system called HPRM or Trim for corporate record keeping. As this system is NSW Government wide, its use could be harnessed to encourage system wide information sharing, akin to other NSW wide electronic systems. This system could be extended initially to project champions as discussed above and this could be extended to other users as appropriate.

While there is retention of knowledge within medical imaging, this is not necessarily shared with people outside of medical imaging who may want to adopt similar strategies as those utilised in medical imaging. A more formalised knowledge exchange process would allow medical imaging to have a more meaningful contribution and influence practices in other departments or specialties. The target audiences of medical imaging's knowledge could be the rest of the hospital or within healthcare generally, on how to effectively lead change.

6.4.3 Change Management and Project Management Training

This research found there was a high change project success rate within medical imaging utilising the dominant consultative leadership style. This style could be utilised in organisational change in healthcare to improve the success outcomes lifting it from 30% as discussed in chapters 2; 4 and 5.

Formal change management and project management training modules relevant to medical imaging could be developed and offered at intervals to ensure uptake by medical imaging practitioners who are already time poor. Because of this, clinicians tend to prioritise treating patients than spending significant time in what they may term as non-core activities.

While the organisation offers a Ministry of Health endorsed AIM course for project and change management, none of the leaders interviewed had attended this course despite being aware of its existence. There is apathy in undertaking this course perhaps due to its length (3 days at the time of research project); the perceived lack of value in the course; feedback from prior participants or participants' perception that project management is not a core function of their roles. The findings of this study suggest that organisations could undertake a study to establish the reasons for such apathy and take appropriate remedial action to increase engagement in training. Another alternative to this would be to bring in an external change agent to support clinicians to undertake change management projects.

6.4.4 Private Sector Managers

Most medical imaging practices are generally transferable between private and public sector establishments. The complexity of cases seen in smaller medical imaging practices will be clearly different from those seen in NSW public facilities. NSW public hospitals cater for sicker and more complex patients such as trauma or stroke patients (Royal Australia and New Zealand College of Radiology 2019). With this process, NSW public hospitals sometimes outsource certain types of studies to private providers under contractual arrangements, achieving successful change management outcomes utilising the leadership approaches suggested in this research becomes quite important for the sustainability reasons. Leaders who are or work for private providers, are encouraged to employ a consultative leadership style to successfully lead change. The finding from this research as discussed in Chapter 5 could therefore be applicable not only to private sector imaging departments, but also to other service departments such as pathology or other private sector departments operating in a similar manner to NSW Imaging practices. Leaders in private practice could adopt the leadership styles found in this research to be effective in successfully leading change.

6.4.5 Public Sector Policy Analysts and Managers

Public sector policy analysts and managers play a critical role in charting healthcare policy. This is so as they generally discuss with various stakeholders, including LHDs, academia, practitioners, equipment manufacturers as part of their consultation process before drafting health policy. An understanding of leadership styles that positively impact change management in healthcare will assist them in the design of public policy so that practitioners can implement it. These implications for the public sector have been subdivided into subtopics to allow for focused discussion topics in the following paragraphs.

6.4.5.1 Capital and Infrastructure Procurement

Participants reported that capital procurement processes within the LHD and within NSW Health appear disjointed at best or haphazard at worst. Research

participants in this study suggest decision making is not timely and the process quite convoluted. There is a need for investigation of the efficacy of an actionable Capital Planning Program within NSW Health rather than lists of equipment held centrally by the NSW Health Ministry of Health with no meaningful feedback to department managers. Participants lamented that often, managers are advised by the Ministry of Health just before the financial year end, that they need to procure usually large pieces of equipment such as CT scanners or MRI machines under very tight turnaround times, without a proper due diligence process having occurred.

6.4.5.2 Enhanced Role of the Project Management Office (PMO)

It was noted that the LHD had a limited number of project officers, particularly in the capital works and infrastructure. Projects in this domain are facilitated and led by the capital works and infrastructure team which have more clearly defined processes and documentation in project management. The NSW Ministry of Health does have a Project Management Office (PMO). However, there appears to be very little collaboration between this centralised office and the LHD. It is recommended that the PMO undertakes a review of its collaboration with LHDs, particularly services that conduct change projects regularly. This may build and enhance capacity in these services to ensure that projects undertaken within the health system are value for money services. This may also provide an opportunity for information sharing on the key project management documentation required by medical imaging leaders. In SWSLHD, strategic projects may be managed by the Clinical Innovation and Business Unit or similar service such as the Procurement Service.

6.4.5.3 Organisational Learning for Different Staffing Groups

This research noted that organisational learning was reactive in some instances in medical imaging. While operations staff were aware that the lack of a second scanner at one hospital would exacerbate CT scanning demand, the executive appeared not to have on a higher priority, the replacement of this machine. A more responsive equipment procurement process as suggested in section 6.4.5.2 such as consultative processes in designing equipment specifications with endusers and the procurement function would assist in managing this process. One could also argue that there was a lack of alignment of performance KPIs against which executive performance was measured. This could be achieved by a closer alignment of clinical performance indicators, imaging performance and executive performance indicators.

6.4.5.4 Human Resources and Staff Development

While the leadership styles employed by medical imaging leaders may lead to successful change management within medical imaging, it raises the concern of a lack of alternative views and skills in leading change as evidenced by 92% of leaders using a consultative leadership style (section 5.5). These divergent views strengthen the decision-making processes as different scenarios are considered prior to any decisions being made. Medical imaging departments may need to consider the implications of this finding in the development of alternative skills or hiring practices to ensure a balance in the staff member competencies and skills within the departments. This could be done through personality testing or psychometric testing of aspiring leaders in medical imaging. Alternatively, the curricula for medical imaging educational programs could include leadership tuition to assist in the leadership skill development of future medical imaging leaders.

6.4.5.5 Innovations in Hard Built Greenfield and Brownfield Buildings

This research found that configurations or building layout impacted on the quality of care provided to patients. New innovations such as the integration of diagnostic testing at the point of care, i.e., in emergency departments are critical considerations for greenfield sites or those seeking to achieve efficiencies in workflows. This is so because it eliminates the porter, reducing patient transport challenges faced by hospitals with large footprints and which are lacking in contemporary patient transport models. The same can also be said of brownfield sites. From time to time, public hospitals undergo renewal. As suggested in section 6.4.5 policy analysts may promote the co-location or integration of EDs and medical imaging departments.

6.4.5.6 Financial Considerations in Decision Making

The increasing healthcare costs identified in section 1.2.5 and the misalignment of KPIs discussed in section 6.4.5.3 make it a challenging proposition to manage healthcare budgets. The researcher's experience has shown that there may not be a willingness to invest in some human capital-intensive initiatives. Participants reported that well-known problems appear not to be resolved for extended periods of time due to other financial constraints. For example, this research found that in larger hospitals the issue of porters (wardsman) is widely recognised as a challenge in internal transport of patients between wards or for studies. Patient transport turnaround time as identified in section 3.5.4.1 was the basis upon which leaders assessed the effectiveness of patient transport processes. Perhaps innovative solutions need to be considered, such as including relatives as partners in care. They can be partners through assisting in the transportation of appropriate patients for imaging studies, which would free up porters to those patients who require their assistance. All research is not without limitations. The next section discusses the limitations of this research.

6.5 Limitations

The researcher worked as Business Manager for medical imaging services for SWSLHD for three years. In this time, the researcher was involved in several change management projects. This knowledge of medical imaging services could potentially have impacted on the researcher's interpretation of the information provided by the research participants. The potential bias was managed through the transcription of all interviews and verification by interviewees of each interview to ensure that their response had been captured accurately. Additionally, these results were discussed with the supervisory panel.

Prior to ethics approval, the researcher transferred to another role outside of the medical imaging service to an arm's length service, though still within the same LHD. As a result, the influence held by the researcher was deemed to be sufficiently managed to pose any ethical risk.

6.6 Further Research

Undertaking this research project has answered the questions that motivated the research, such as the causes of change project or process failure. As a previous leader within medical imaging this understanding will enable present leaders to take active steps to address knowledge or skills deficits so that they can lead projects successfully in the future and contribute to the scant scholarly literature in the area. The findings have indicated potential areas to explore for future research.

This study found that there is limited structured planning for equipment procurement within healthcare. Studies investigating the planning models utilised within the public healthcare sector settings in Australia could be completed, with the findings also adding to this research. There is also a need for further research into newer equipment acquisition models that consider the capital sensitivity rules for medical imaging equipment as well as the sustainability of such models. This could be in the form of equipment funding products or schemes.

There is generally limited leadership research within medical imaging in healthcare, indicating that there is scope for further research in this area. Personal observations have led to the understanding that there are competing interests between the private practice arrangements of physicians in medical imaging and their employment arrangements within LHDs. Anecdotal reports not covered in this thesis include the four-day working week for medical imaging physicians, its impact on patient care and workload management within public medical imaging departments. These could be further investigated.

During this research, there was a major critical incident resulting in the death of an infant in the LHD. There were significant implications for the way that projects were generally conducted within the LHD, particularly governance concerns to manage risks. A future research area could be in the areas of governance and risk management and how they affect the leadership style employed. Such a study would add to this research and literature on leadership generally.

6.7 Conclusion

In conclusion, this study found consultative leadership as the dominant leadership style in leading change management in medical imaging in healthcare. Leaders should therefore utilise this leadership style in change management projects to achieve superior outcomes to those expected from other leadership styles.

The definition of effective leadership as leadership that follows several characteristics and results in successful change management projects is a key contribution of this research. Previous literature had not adequately discussed the concept of strong leadership, which other scholars should now adopt.

The study also defined and extended literature on learning organisations by identifying that medical imaging leaders who learnt from prior change processes; achieved more positive outcomes in change management projects than non-learning leaders. This scholarly contribution is crucial for the modernday medical imaging leader who is required to embrace emerging challenges in healthcare provision.

Finally, this research's findings, methodology, extensions to existing healthcare literature and recommendations to practice, have thus made a scholarly contribution to the area of medical imaging in healthcare.

References Cited

Aarons, GA, Hurlburt, M & Horwitz, SM 2011, 'Advancing a conceptual model of evidence-based practice implementation in public service sectors', *Administration and Policy in Mental Health and Mental Health Services Research*, vol. 38, no. 1, pp. 4-23.

Abdulghani, HM, Shaik, SA, Khamis, N, Al-Drees, AA, Irshad, M, Khalil, MS et al. 2014, 'Research methodology workshops evaluation using the Kirkpatrick's model: translating theory into practice', *Medical Teacher*, vol. 36, no. sup1, pp. S24-S9.

Abdulla, H, Alhashimi, M & Hamdan, AM 2019, 'The Impact of Project Management Methodologies on Project Success: A Case Study of the Oil and Gas Industry in the Kingdom of Bahrain', in *Handbook of Research on Implementing Knowledge Management Strategy in the Public Sector*, IGI Global, pp. 418-37.

Abdullah, NNB, Clancey, WJ, Raj, A, Zain, AZM, Khalid, KF & Ooi, A 2018, 'Application of a double-loop learning approach for healthcare systems design in an emerging market', pp. 10-3.

Aberese-Ako, M, Agyepong, IA & van Dijk, H 2018, 'Leadership styles in two Ghanaian hospitals in a challenging environment', *Health Policy and Planning*, vol. 33, no. suppl_2, pp. ii16-ii26.

Abuhejleh, Aa, Dulaimi, M & Ellahham, S 2016, 'Using Lean management to leverage innovation in healthcare projects: case study of a public hospital in the UAE', *BMJ Innovations*, vol. 2, no. 1, pp. 22-32.

Adams, WC 2015, 'Conducting semi-structured interviews', *Handbook of practical program evaluation*, vol. 4, pp. 492-505.

Aga, DA, Noorderhaven, N & Vallejo, B 2016, 'Transformational leadership and project success: The mediating role of team-building', *International Journal of Project Management*, vol. 34, no. 5, pp. 806-18.

Aggarwal, R & Swanwick, T 2015, 'Clinical leadership development in postgraduate medical education and training: policy, strategy, and delivery in the UK National Health Service', *Journal of healthcare leadership*, vol. 7, p. 109.

Agote, L, Aramburu, N & Lines, R 2016, 'Authentic leadership perception, trust in the leader, and followers' emotions in organizational change processes', *The Journal of Applied Behavioral Science*, vol. 52, no. 1, pp. 35-63.

Aguinis, H, Ramani, RS & Alabduljader, N 2018, 'What you see is what you get? Enhancing methodological transparency in management research', *Academy of Management Annals*, vol. 12, no. 1, pp. 83-110.

Aguirre, D, von Post, R & Alpern, M 2013, 'Culture's Role in Enabling Organizational Change: Survey Ties Transformation Success to Deft Handling of Cultural Issues', *Strategy and Global Culture and Change Management Survey 2013*.

Ahmady, GA, Mehrpour, M & Nikooravesh, A 2016, 'Organizational structure', *Procedia-Social and Behavioral Sciences*, vol. 230, pp. 455-62.

Ahn, S, Kim, WY, Lim, KS, Ryoo, SM, Sohn, CH, Seo, DW et al. 2014, 'Advanced radiology utilization in a tertiary care emergency department from 2001 to 2010', *PloS one*, vol. 9, no. 11, p. e112650.

Ajayi, O 2002, *Leading Change: Leading 08.06*, Wiley.

Ajmal, S, Farooq, MZ, Sajid, N & Awan, S 2012, 'Role of leadership in change management process', *Abasyn University Journal of Social Sciences*, vol. 5, no. 2, pp. 111-24.

Allen, JA, Reiter-Palmon, R, Crowe, J & Scott, C 2018, 'Debriefs: Teams learning from doing in context', *American Psychologist*, vol. 73, no. 4, p. 504.

Alqahtani, T 2019, 'Barriers to Women's leadership', *Granite Journal: A Postgraduate Interdisciplinary Journal*, vol. 3, no. 2, pp. 34-41.

Alvarenga, JC, Branco, RR, Guedes, ALA & Soares, CAP 2019, 'The project manager core competencies to project success', *International Journal of Managing Projects in Business*, vol. 13, no. 2, pp. 277-92.

Alvesson, M, Blom, M & Sveningsson, S 2017, *Reflexive leadership: Organising in an imperfect world*, Sage, London United Kindom.

Amanchukwu, RN, Stanley, GJ & Ololube, NP 2015, 'A review of leadership theories, principles and styles and their relevance to educational management', *Management*, vol. 5, no. 1, pp. 6-14.

Anderson, MH & Sun, PY 2017, 'Reviewing leadership styles: Overlaps and the need for a new 'full-range'theory', *International Journal of Management Reviews*, vol. 19, no. 1, pp. 76-96.

Anderson, NC & Kovach, JV 2014, 'Reducing Welding Defects in Turnaround Projects: A Lean Six Sigma Case Study', *Quality Engineering*, vol. 26, no. 2, pp. 168-81, DOI 10.1080/08982112.2013.801492, <<u>https://doi.org/10.1080/08982112.2013.801492</u>>. Andersson, T 2015, 'The medical leadership challenge in healthcare is an identity challenge', *Leadership in Health Services*, vol. 28, no. 2, pp. 83-99.

Anitha, J 2014, 'Determinants of employee engagement and their impact on employee performance', *International Journal of productivity and performance management*, vol. 63, no. 3, pp. 308-23.

Antonakis, J & Day, DV 2017, 'Leadership: Past, present, and future', in *The nature of leadership*, 3rd edn, Sage, United States, pp. 3-26.

Armstrong-Stassen, M, Freeman, M, Cameron, S & Rajacic, D 2015, 'Nurse managers' role in older nurses' intention to stay', *Journal of health organization and management*, vol. 29, no. 1, pp. 55-74.

Arnold, JA, Arad, S, Rhoades, JA & Drasgow, F 2000, 'The empowering leadership questionnaire: The construction and validation of a new scale for measuring leader behaviors', *Journal of Organizational Behavior*, vol. 21, no. 3, pp. 249-69, ProQuest Central; Social Science Premium Collection, DOI <u>http://dx.doi.org/10.1002/(SICI)1099-1379(200005)21:3</u><249::AID-JOB10>3.0.CO;2-#.

Asrar-ul-Haq, M & Anjum, T 2020, 'Impact of narcissistic leadership on employee work outcomes in banking sector of Pakistan', *Future Business Journal*, vol. 6, no. 1, pp. 1-9.

Asrar-ul-Haq, M & Kuchinke, KP 2016, 'Impact of leadership styles on employees' attitude towards their leader and performance: Empirical evidence from Pakistani banks', *Future Business Journal*, vol. 2, no. 1, pp. 54-64, DOI <u>https://doi.org/10.1016/j.fbj.2016.05.002</u>, <<u>http://www.sciencedirect.com/science/article/pii/S2314721016300032</u>>.

Athanasopoulou, A, Moss-Cowan, A, Smets, M & Morris, T 2018, 'Claiming the corner office: Female CEO careers and implications for leadership development', *Human Resource Management*, vol. 57, no. 2, pp. 617-39.

Australian Health Practitioner Regulation Agency 2021, *Accreditation functions*, Australian Health Practitioner Regulation Agency, viewed 16 September 2021, <<u>https://www.ahpra.gov.au/Accreditation/Accreditation-functions.aspx</u>>.

Australian Insititute of Health and Welfare 2021, *Deaths in Australia*, AIHW, Canberra, viewed 21 June 2021, <<u>https://www.aihw.gov.au/reports/life-expectancy-death/deaths-in-australia</u>>.

Australian Institute of Health and Welfare 2009a, *Australian Hospital Statistics 2007-08*, Australian Institute of Health and Welfare, Canberra.

Australian Institute of Health and Welfare 2009b, *Health Expenditure Australia 2007-08,*, Australian Institute of Health and Welfare, Canberra.

Australian Institute of Health and Welfare 2014, *Health expenditure Australia 2012–13*, Australian Institute of Health and Welfare, Canberra, viewed 20 January 2021.

Australian Institute of Health and Welfare 2019, *Health expenditure Australia 2017–18*, Australian Institute of Health and Welfare, Canberra, viewed 18 Sept 2020.

Australian Institute of Health and Welfare 2021, *Health system overview*, Australian Institute of Health and Welfare, viewed 15 October 2021, <<u>https://www.aihw.gov.au/reports/australias-health/health-system-overview</u>>.

Avolio, BJ, Wernsing, T & Gardner, WL 2018, 'Revisiting the development and validation of the authentic leadership questionnaire: Analytical clarifications', *Journal of management*, vol. 44, no. 2, pp. 399-411.

Ayres, J 2012, 'Power and love: a theory and practice of social change', *Community Development*, vol. 43, no. 2, pp. 278-9.

Barley, WC, Treem, JW & Kuhn, T 2018, 'Valuing multiple trajectories of knowledge: A critical review and agenda for knowledge management research', *Academy of Management Annals*, vol. 12, no. 1, pp. 278-317.

Barnard, H 2020, 'Another Pandemic in Africa: Weak Healthcare, Strong Leadership, and Collective Action in Africa's COVID-19 Response', *Management and Organization Review*, vol. 16, no. 4, pp. 753-9, ProQuest Central, DOI <u>https://doi.org/10.1017/mor.2020.47</u>.

Barros, MV, Ferreira, MB, do Prado, GF, Piekarski, CM & Picinin, CT 2020, 'The interaction between knowledge management and technology transfer: a current literature review between 2013 and 2018', *The Journal of Technology Transfer*, vol. 45, no. 5, pp. 1585-606.

Bass, BM & Avolio, BJ 1990, *Transformational leadership development: Manual for the multifactor leadership questionnaire*, Consulting Psychologists Press, Palo Alto, CA.

Bass, BM & Avolio, BJ 1996, 'Multifactor leadership questionnaire', *European Journal of Psychological Assessment*.

Bass, BM & Avolio, BJ 2004, 'Multifactor Leadership Questionnaire: MLQ; manual and sampler set ', Mind Garden, Redwood City, CA.

Beaudry, JS & Miller, L 2016, *Research literacy: A primer for understanding and using research*, Guilford Publications, New York, USA.

Bell, BS, Tannenbaum, SI, Ford, JK, Noe, RA & Kraiger, K 2017, '100 years of training and development research: What we know and where we should go', *Journal of Applied Psychology*, vol. 102, no. 3, p. 305.

Bell, J 2014, *Doing Your Research Project: A guide for first-time researchers*, McGraw-Hill Education (UK), Berkshire, England.

Belrhiti, Z, Van Belle, S & Criel, B 2021, 'How medical dominance and interprofessional conflicts undermine patient-centred care in hospitals: historical analysis and multiple embedded case study in Morocco', *BMJ Global Health*, vol. 6, no. 7, p. e006140.

Berger, R, Romeo, M, Guardia, J, Yepes, M & Soria, MA 2012, 'Psychometric properties of the spanish human system audit short-scale of transformational leadership', *The Spanish journal of psychology*, vol. 15, no. 1, pp. 367-76.

Berghout, MA, Fabbricotti, IN, Buljac-Samardžić, M & Hilders, CG 2017, 'Medical leaders or masters?—A systematic review of medical leadership in hospital settings', *PloS one*, vol. 12, no. 9, p. e0184522.

Better Health Victoria 2015, *CT Scan*, Better Health Victoria, viewed 04 April 2015, <<u>http://www.betterhealth.vic.gov.au/bhcv2/bhcarticles.nsf/pages/CT_scan</u>>.

Bevan, D, Kipka, C, Kumaraswamy, KSN & Chitale, C 2012, 'Collaborative knowledge sharing strategy to enhance organizational learning', *Journal of Management Development*.

Bhattacharyya, O, Blumenthal, D, Stoddard, R, Mansell, L, Mossman, K & Schneider, EC 2019, 'Redesigning care: adapting new improvement methods to achieve person-centred care', *BMJ quality & safety*, vol. 28, no. 3, pp. 242-8.

Bhuasiri, W, Xaymoungkhoun, O, Zo, H, Rho, JJ & Ciganek, AP 2012, 'Critical success factors for e-learning in developing countries: A comparative analysis between ICT experts and faculty', *Computers & Education*, vol. 58, no. 2, pp. 843-55.

Bismark, M, Morris, J, Thomas, L, Loh, E, Phelps, G & Dickinson, H 2015, 'Reasons and remedies for under-representation of women in medical leadership roles: a qualitative study from Australia', *BMJ open*, vol. 5, no. 11, p. e009384.

Blake, N 2020, 'Authentic Leadership: How Does It Differ From Transformational Leadership?', *AACN Advanced Critical Care*, vol. 31, no. 2, pp. 196-7.

Blank, CA 2013, 'SAGE handbook of mixed methods in social & behavioral research', *Journal of music therapy*, vol. 50, no. 4, p. 321.

Boamah, SA & Tremblay, P 2019, 'Examining the factor structure of the MLO transactional and transformational leadership dimensions in nursing context', *Western journal of nursing research*, vol. 41, no. 5, pp. 743-61.

Boaz, A, Hanney, S, Jones, T & Soper, B 2015, 'Does the engagement of clinicians and organisations in research improve healthcare performance: a three-stage review', *BMJ open*, vol. 5, no. 12, p. e009415.

Bogers, M, Chesbrough, H & Moedas, C 2018, 'Open innovation: Research, practices, and policies', *California management review*, vol. 60, no. 2, pp. 5-16.

Bolman, LG & Deal, TE 2017, *Reframing organizations: Artistry, choice, and leadership*, John Wiley & Sons.

Boonyachai, Y 2011, 'An investigation of the leadership styles of middle managers in the Thai hotel industry using the MLQ (5X-Short Form) and Hofstede's Cultural Dimensions'.

Bordenave, E 2017, *A Case Study of the Perception of Succession by Audiologists in Private Practice*, Northcentral University.

Botella, C, Rueda, S, López-Iñesta, E & Marzal, P 2019, 'Gender diversity in STEM disciplines: A multiple factor problem', *Entropy*, vol. 21, no. 1, p. 30.

Boutcher, FD 2019, *A multi-methods study of the role of middle managers in brokering knowledge in hospitals*, University of Toronto (Canada).

Brady, Z, Cain, TM & Johnston, PN 2011, 'Paediatric CT imaging trends in Australia', *Journal of Medical Imaging and Radiation Oncology*, vol. 55, no. 2, pp. 132-42.

Braithwaite, J 2018, 'Changing how we think about healthcare improvement', *BMJ*, vol. 361.

Braithwaite, J, Glasziou, P & Westbrook, J 2020, 'The three numbers you need to know about healthcare: the 60-30-10 Challenge', *BMC Medicine*, vol. 18, no. 1, p. 102, DOI 10.1186/s12916-020-01563-4, <<u>https://doi.org/10.1186/s12916-020-01563-4</u>>.

Braithwaite, J, Westbrook, J, Coiera, E, Runciman, WB, Day, R, Hillman, K et al. 2017, 'A systems science perspective on the capacity for change in public hospitals', *Israel Journal of Health Policy Research*, vol. 6, no. 1, pp. 16-, DOI 10.1186/s13584-017-0143-6.

Brant, W & CA, H 2012, *Chest, Thyroid, Parathyroid, and Neonatal Brain Ultrasound*, 4th edn, Lippincott Williams & Wilkins, Philadelphia, USA.

Bratton, J & Gold, J 2017, *Human resource management: theory and practice*, Palgrave.

Breevaart, K & Zacher, H 2019, 'Main and interactive effects of weekly transformational and laissez-faire leadership on followers' trust in the leader and leader effectiveness', *Journal of Occupational and Organizational Psychology*, vol. 92, no. 2, pp. 384-409.

Briggs, DS 2017, 'Challenges for Health Systems: Australian Perspectives'.

Briggs, DS & Isouard, G 2016, 'The language of health reform and health management: critical issues in the management of health systems', *Asia Pacific Journal of Health Management*, vol. 11, no. 3, pp. 38-44.

British Society of Interventional Radiology 2020, *For patients - What is Interventional Radiology*, viewed 22 October 2020, <<u>http://www.bsir-qi.org/what-is-ir/</u>>.

Brooke-Sumner, C, Petersen-Williams, P, Kruger, J, Mahomed, H & Myers, B 2019, ''Doing more with less': a qualitative investigation of perceptions of South African health service managers on implementation of health innovations', *Health Policy and Planning*, vol. 34, no. 2, pp. 132-40, viewed 10/27/2020, DOI 10.1093/heapol/czz017, <<u>https://doi.org/10.1093/heapol/czz017</u>>.

Brown, A & Danaher, PA 2019, 'CHE principles: Facilitating authentic and dialogical semi-structured interviews in educational research', *International Journal of Research & Method in Education*, vol. 42, no. 1, pp. 76-90.

Brown, R & Abuatiq, A 2020, 'Resilience as a strategy to survive organizational change', *Nursing Management*, vol. 51, no. 2, pp. 16-21.

Brown, R, Wey, H & Foland, K 2018, 'The relationship among change fatigue, resilience, and job satisfaction of hospital staff nurses', *Journal of Nursing Scholarship*, vol. 50, no. 3, pp. 306-13.

Buchanan, D & Badham, R 2020, *Power, politics, and organizational change*, 3rd edn, Sage, London, UK.

Bui, HT 2020, 'From the fifth discipline to the new revolution: what we have learnt from Senge's ideas over the last three decades', *The Learning Organization*.

Burke, WW 2017, *Organization change: Theory and practice*, 5th edn, Sage publications, Carlifonia, USA.

Burns, JM 1978, *Leadership*, Harper & Row, New York.

Burrell, G & Morgan, G 2017, *Sociological paradigms and organisational analysis: Elements of the sociology of corporate life*, Routledge.

Burton, LJ 2019, 'Under-representation of women in leadership roles in women's sport', in *Routledge handbook of the business of women's sport*, Routledge, pp. 255-68.

Busari, AH, Khan, SN, Abdullah, SM & Mughal, YH 2020, 'Transformational leadership style, followership, and factors of employees' reactions towards organizational change', *Journal of Asia Business Studies*, vol. 14, no. 2, pp. 181-209.

Callander, E, Sidebotham, M, Lindsay, D & Gamble, J 2021, 'The future of the Australian midwifery workforce – impacts of ageing and workforce exit on the number of registered midwives', *Women and Birth*, vol. 34, no. 1, pp. 56-60, DOI <u>https://doi.org/10.1016/j.wombi.2020.02.023</u>,

<<u>https://www.sciencedirect.com/science/article/pii/S187151921930900X</u>>.

Cameron, E & Green, M 2019, *Making sense of change management: A complete guide to the models, tools and techniques of organizational change*, Kogan Page Publishers.

Camilleri, J, Cope, V & Murray, M 2019, 'Change fatigue: The frontline nursing experience of large-scale organisational change and the influence of teamwork', *J Nurs Manag*, vol. 27, no. 3, pp. 655-60.

Campanella, F, Della Peruta, MR, Bresciani, S & Dezi, L 2017, 'Quadruple Helix and firms' performance: an empirical verification in Europe', *The Journal of Technology Transfer*, vol. 42, no. 2, pp. 267-84.

Campbell, S, Greenwood, M, Prior, S, Shearer, T, Walkem, K, Young, S et al. 2020, 'Purposive sampling: complex or simple? Research case examples', *Journal of Research in Nursing*, vol. 25, no. 8, pp. 652-61.

Carstens, DS & Richardson, GL 2019, *Project management tools and techniques: A practical guide*, 2nd ed. edn, CRC Press, Boca Raton USA.

Casali, G & Day, G 2015, 'A taxonomic approach to understanding managerial ethical decision making approaches of clinically and non-clinically trained healthcare managers in Australia', *Asia Pacific Journal of Health Management*, vol. 10, no. 3, pp. SI8-SI17.

Chaplin, S 2020, 'Protecting parliamentary democracy in "plague" times: institutions Accountability and democratic during the pandemic', Commonwealth Law Bulletin, vol. 46, 110-23, DOI no. 1, pp.

10.1080/03050718.2020.1762235, <<u>https://doi.org/10.1080/03050718.2020.1762235</u>>.

Chapman, AL, Johnson, D & Kilner, K 2014, 'Leadership styles used by senior medical leaders: patterns, influences and implications for leadership development', *Leadership in Health Services*, vol. 27, no. 4, pp. 283-98.

Chatziralli, I, Ventura, CV, Touhami, S, Reynolds, R, Nassisi, M, Weinberg, T et al. 2021, 'Transforming ophthalmic education into virtual learning during COVID-19 pandemic: a global perspective', *Eye*, vol. 35, no. 5, pp. 1459-66.

Chen, X-P, Eberly, MB, Chiang, T-J, Farh, J-L & Cheng, B-S 2014, 'Affective trust in Chinese leaders: Linking paternalistic leadership to employee performance', *Journal of management*, vol. 40, no. 3, pp. 796-819.

Cheng, C, Bartram, T, Karimi, L & Leggat, S 2016, 'Transformational leadership and social identity as predictors of team climate, perceived quality of care, burnout and turnover intention among nurses', *Personnel review*, vol. 45, no. 6, pp. 1200-16, viewed 2021/10/10, DOI 10.1108/PR-05-2015-0118, <<u>https://doi.org/10.1108/PR-05-2015-0118</u>>.

Cherry, K 2012, *Leadership theories*, Sage, viewed 18 November 2021, <<u>https://www.verywellmind.com/leadership-theories-2795323</u>>.

Cheryan, S 2012, 'Understanding the Paradox in Math-Related Fields: Why Do Some Gender Gaps Remain While Others Do Not?', *Sex Roles*, vol. 66, no. 3-4, pp. 184-90, GenderWatch; ProQuest Central; Social Science Premium Collection, DOI <u>http://dx.doi.org/10.1007/s11199-011-0060-z</u>.

Chew, C, Cannon, P & O'Dwyer, PJ 2020, 'Radiology for medical students (1925–2018): an overview', *BJR/ Open*, vol. 2, no. 1, p. 20190050.

Clarke, A, Watt, I, Sheard, L, Wright, J & Adamson, J 2017, 'Implementing electronic records in NHS secondary care organizations in England: policy and progress since 1998', *British medical bulletin*, vol. 121, no. 1, pp. 95-106.

Clarke, V & Braun, V 2018, 'Using thematic analysis in counselling and psychotherapy research: A critical reflection', *Counselling and psychotherapy research*, vol. 18, no. 2, pp. 107-10.

Cleary, S, Toit, Ad, Scott, V & Gilson, L 2018, 'Enabling relational leadership in primary healthcare settings: lessons from the DIALHS collaboration', *Health Policy and Planning*, vol. 33, no. suppl_2, pp. ii65-ii74.

Coleman, DS 2017, 'The anaesthetist as a leader and manager: Priorities, dilemmas and solutions', *Australasian Anaesthesia*, no. 2017, pp. 343-52.
Colla, CH, Lewis, VA, Shortell, SM & Fisher, ES 2014, 'First national survey of ACOs finds that physicians are playing strong leadership and ownership roles', *Health Affairs*, vol. 33, no. 6, pp. 964-71.

Cook, JM, O'Donnell, C, Dinnen, S, Coyne, JC, Ruzek, JI & Schnurr, PP 2012, 'Measurement of a model of implementation for health care: toward a testable theory', *Implementation Science*, vol. 7, no. 1, p. 59.

Cooper, H 2016, *Ethical choices in research: Managing data, writing reports, and publishing results in the social sciences*, illustrated edn, American Psychological Association.

Council of Australian Governments 2011, National Health Reform Agreement(2011),CommonwealthofAustralia,Canberra<https://federalfinancialrelations.gov.au/sites/federalfinancialrelations.gov.au/sites/federalfinancialrelations.gov.au/files/2021-08/national-agreement.pdf

Cousins, B 2018, 'Design thinking: Organizational learning in VUCA environments', *Academy of Strategic Management Journal*, vol. 17, no. 2, pp. 1-18.

Cox, CW & Lynch, DA 2015, 'Medical imaging in occupational and environmental lung disease', *Current opinion in pulmonary medicine*, vol. 21, no. 2, pp. 163-70.

Coyle, B, Degeling, P & Gray, A 2017, 'Clinical and Managerial Cultures of Evaluative Activity in the Provision of Acute Health Care', in *Evaluation Cultures*, Routledge, pp. 139-58.

Cresswell, K & Sheikh, A 2013, 'Organizational issues in the implementation and adoption of health information technology innovations: an interpretative review', *International Journal of Medical Informatics*, vol. 82, no. 5, pp. e73-e86.

Cummings, GG, Tate, K, Lee, S, Wong, CA, Paananen, T, Micaroni, SPM et al. 2018, 'Leadership styles and outcome patterns for the nursing workforce and work environment: A systematic review', *International Journal of Nursing Studies*, vol. 85, pp. 19-60, DOI <u>https://doi.org/10.1016/j.ijnurstu.2018.04.016</u>, <<u>https://www.sciencedirect.com/science/article/pii/S0020748918301019</u>>.

Daft, RL & Marcic, D 2016, *Understanding management*, 10th edn, Cengage Learning, Boston USA.

Daire, J & Gilson, L 2014, 'Does identity shape leadership and management practice? Experiences of PHC facility managers in Cape Town, South Africa', *Health Policy and Planning*, vol. 29, no. suppl_2, pp. ii82-ii97.

Daly, J, Jackson, D, Rumsey, M, Patterson, K & Davidson, PM 2015, 'Building nursing leadership capacity: An australian snapshot', *Nurse Leader*, vol. 13, no. 5, pp. 36-9.

Daly, MP, Cleary, MI & McCormack, LJ 2012, 'The Bundaberg emergency response team', *Australian Health Review*, vol. 36, no. 4, pp. 384-9.

Darling, J 2017, 'A conversation with Edgar Schein: Aligning strategy, culture, and leadership', *People & Strategy*, vol. 40, no. 2, pp. 64-8.

David, K & Golan, C 2017, 'The mediating effect of interpersonal trust on virtual team's collaboration', *International Journal of Knowledge Management (IJKM)*, vol. 13, no. 3, pp. 20-37.

Davies, H & Nutley, S 2018, 'Organizations as learning systems', in *Complexity and healthcare organization*, CRC Press, pp. 59-68.

Day, DV & Thornton, AMA 2018, 'The nature of leadership', in Antonakis J. & DV Day (eds), *Leadership development*, 3rd edn, Sage, CA.

Day, G & Casali, G 2015, 'Do health inquiries lead to health system change?: What have we learnt from recent inquiries and will the same mistakes happen again?', *Asia Pacific Journal of Health Management*, vol. 10, no. 3.

de Graaf, FJ 2019, 'Ethics and behavioural theory: how do professionals assess their mental models?', *Journal of Business Ethics*, vol. 157, no. 4, pp. 933-47.

De Ponti, R, Marazzato, J, Maresca, AM, Rovera, F, Carcano, G & Ferrario, MM 2020, 'Pre-graduation medical training including virtual reality during COVID-19 pandemic: a report on students' perception', *BMC medical education*, vol. 20, no. 1, pp. 1-7.

DeJonckheere, M & Vaughn, LM 2019, 'Semistructured interviewing in primary care research: a balance of relationship and rigour', *Family medicine and community health*, vol. 7, no. 2, pp. e000057-e, PubMed, DOI 10.1136/fmch-2018-000057, <<u>https://pubmed.ncbi.nlm.nih.gov/32148704</u> https://www.ncbi.nlm.nih.gov/pmc/articles/PMC6910737/>.

deJong, LR 2018, *A comparative study of leadership practices of nurse leaders and patient satisfaction ratings in magnet and non-magnet hospitals*, Grand Canyon University.

Delaney, L 2015, 'The challenges of an integrated governance process in healthcare', *International Journal*, vol. 20, no. 2, pp. 74-81.

Démeh, W & Rosengren, K 2015, 'The visualisation of clinical leadership in the content of nursing education—a qualitative study of nursing students' experiences', *Nurse education today*, vol. 35, no. 7, pp. 888-93.

Denhardt, RB, Denhardt, JV, Aristigueta, MP & Rawlings, KC 2018, *Managing human behavior in public and nonprofit organizations*, 5th edn, CQ Press, Carlifonia USA.

Denis, J-L & Van Gestel, N 2016, 'Medical doctors in healthcare leadership: theoretical and practical challenges', *BMC Health Services Research*, vol. 16, no. 2, pp. 45-56.

Department of Health 2016, *Diagnostic Imaging Quality Programs*, Department of Health,, viewed 20 September 2021, <<u>https://www1.health.gov.au/internet/main/publishing.nsf/Content/di-qualprog#:~:text=Diagnostic%20Imaging%20Quality%20Practice%20Program%20In %202004%20the,medicine%20industries%20to%20the%20reallocation%20of%2 Ounspent%20funds.>.</u>

Department of Health 2020a, *About capital sensitivity and guaranteeing quality diagnostic imaging services.*, The Department of Health, viewed 30 Nov 2020 2020,

<<u>https://www1.health.gov.au/internet/main/publishing.nsf/Content/capsensdi</u>>.

Department of Health 2020b, *Chronic conditions in Australia*, Department of Health, Australian Government, viewed 8 February 2021, <<u>https://www.health.gov.au/health-topics/chronic-conditions/chronic-conditions-in-australia</u>>.

Department of Health 2020c, *The Diagnostic Imaging Accreditation Scheme*, Department of Health, viewed 20 September 2021, <<u>https://www1.health.gov.au/internet/main/publishing.nsf/Content/di-quality</u>>.

Deshwal, V & Ashraf Ali, M 2020, 'A systematic review of various leadership theories', *Shanlax International Journal of Commerce*, vol. 8, pp. 38-43.

Di Somma, S, Paladino, L, Vaughan, L, Lalle, I, Magrini, L & Magnanti, M 2015, 'Overcrowding in emergency department: an international issue', *Internal and emergency medicine*, vol. 10, no. 2, pp. 171-5.

Dienesch, RM & Liden, RC 1986, 'Leader-member exchange model of leadership: A critique and further development', *Academy of management review*, vol. 11, no. 3, pp. 618-34.

Dinh, JE, Lord, RG, Gardner, WL, Meuser, JD, Liden, RC & Hu, J 2014, 'Leadership theory and research in the new millennium: Current theoretical trends and changing perspectives', *The Leadership Quarterly*, vol. 25, no. 1, pp. 36-62.

Doherty, J 2013, 'Strengthening clinical leadership in hospitals', *Municipal Services Project*, Centre for Rural Health, University of the Witwatersrand, South Africa, Cape Town, South Africa.

Donaldson, L 2018, 'Clinical governance: a quality concept', in *Clinical governance in primary care*, CRC Press, pp. 1-16.

Doody, O & Doody, CM 2012, 'Transformational leadership in nursing practice', *British Journal of Nursing*, vol. 21, no. 20, pp. 1212-8, DOI doi:10.12968/bjon.2012.21.20.1212, <<u>http://www.magonlinelibrary.com/doi/abs/10.12968/bjon.2012.21.20.1212</u>>.

Drenkard, K 2013, 'Transformational leadership: unleashing the potential', *Journal of Nursing Administration*, vol. 43, no. 2, pp. 57-8.

DuBrin, AJ 2015, *Leadership: Research findings, practice, and skills*, Cengage Learning.

Duckett, S & Willcox, S 2015, *The Australian health care system*, Oxford University Press.

Dugan, JP 2017, *Leadership theory: Cultivating critical perspectives*, John Wiley & Sons.

Dwyer, PA & Revell, SMH 2016, 'Multilevel influences on new graduate nurse transition: A literature review', *Journal for Nurses in Professional Development*, vol. 32, no. 3, pp. 112-21.

Eder, SP & Register, JL 2014, '10 management considerations for implementing an endovascular hybrid OR', *AORN journal*, vol. 100, no. 3, pp. 260-70.

Edmondson, AC 2018, *The fearless organization: Creating psychological safety in the workplace for learning, innovation, and growth*, John Wiley & Sons.

Edmonstone, JD 2017, 'Organisational learning', in *Why Hospitals Fail*, Springer, pp. 129-35.

Edwards, MS, Lawrence, SA & Ashkanasy, NM 2016, 'How Perceptions and Emotions Shaped Employee Silence in the Case of "Dr. Death" at Bundaberg Hospital', in *Emotions and Organizational Governance*, vol. 12, Emerald Group Publishing Limited, Bingley UK, pp. 341-79.

Edwards, N & Saltman, RB 2017, 'Re-thinking barriers to organizational change in public hospitals', *Israel Journal of Health Policy Research*, vol. 6, no. 1, pp. 1-11.

Ehrlich, H, McKenney, M & Elkbuli, A 2020, 'We asked the experts: virtual learning in surgical education during the COVID-19 pandemic—shaping the future of surgical education and training', *World Journal of Surgery*, vol. 44, no. 7, pp. 2053-5.

Emere, C, Aigbavboa, C & Thwala, W 2018, 'An investigation of leadership styles of construction professionals in the South African construction industry', *Budapest University of Technology and Economics*, pp. 290-5.

Emergency Care Institute 2014, *NEAT - The basics*, Emergency Care Institute NSW, viewed 10 April 2015, <<u>http://www.ecinsw.com.au/NEAT-the-basics</u>>.

Enzmann, DR 2012, 'Radiology's value chain', *Radiology*, vol. 263, no. 1, pp. 243-52.

Erasmus, E, Orgill, M, Schneider, H & Gilson, L 2014, 'Mapping the existing body of health policy implementation research in lower income settings: what is covered and what are the gaps?', *Health Policy and Planning*, vol. 29, no. suppl_3, pp. iii35-iii50.

Erdogan, B, Kraimer, ML & Liden, RC 2002, 'Person-organization fit and work attitudes: The moderating role of leader-member exchange', *Academy of Management Briarcliff Manor, NY 10510*, vol. 2002, no. 1, pp. F1-F6.

Erkutlu, H & Chafra, J 2013, 'Effects of trust and psychological contract violation on authentic leadership and organizational deviance', *Management Research Review*, vol. 36, no. 9, pp. 828-48.

Erol, R, Upton, P, Mackenzie, H, Donnelly, P & Upton, D 2015, 'Delivering the future: an evaluation of a multidisciplinary clinical leadership programme', *British Journal of Healthcare Management*, vol. 21, no. 1, pp. 29-35.

Etikan, I & Bala, K 2017, 'Sampling and sampling methods', *Biom Biostat Int J*, vol. 5, no. 6, pp. 215-7.

Etikan, I, Musa, SA & Alkassim, RS 2016, 'Comparison of convenience sampling and purposive sampling', *American journal of theoretical and applied statistics*, vol. 5, no. 1, pp. 1-4.

Farrall, S, Hunter, B, Sharpe, G & Calverley, A 2015, 'What 'works' when retracing sample members in a qualitative longitudinal study?', *International Journal of Social Research Methodology*, pp. 1-14, viewed 2015/03/30, DOI 10.1080/13645579.2014.993839,

<<u>http://dx.doi.org/10.1080/13645579.2014.993839</u>>.

Fayol, H 2016, *General and industrial management*, Ravenio Books, United Kingdom.

Febres, GE 2017, *Relationship between the Transactional Leadership Styles, the Transformational Leadership Style and Subordinates' Job Satisfaction*, Northcentral University.

Fetters, MD & Molina-Azorin, JF 2017, 'The Journal of Mixed Methods Research starts a new decade: The mixed methods research integration trilogy and its dimensions', *Journal of Mixed Methods Research*, vol. 11, no. 3, pp. 291-307.

Fewings, P & Henjewele, C 2019, *Construction project management: an integrated approach*, 3rd edn, Routledge, Oxon UK and New York USA.

Fiedler, FE 1967, 'A theory of leadership effectiveness. Mcgraw-Hill series in management'.

Fletcher, KA, Friedman, A & Piedimonte, G 2019, 'Transformational and transactional leadership in healthcare seen through the lens of pediatrics', *The Journal of pediatrics*, vol. 204, pp. 7-9. e1.

Fong, KH & Snape, E 2015, 'Empowering leadership, psychological empowerment and employee Outcomes: Testing a multi-level mediating model', *British Journal of Management*, vol. 26, no. 1, pp. 126-38.

Foster, B 2019, 'Empowering Leadership Questionnaire: A Validation Study', *Jurnal Computech & Bisnis*, vol. 13, no. 2, pp. 144-50.

Franklin, BD, Panesar, SS, Vincent, C & Donaldson, LJ 2014, 'Identifying systems failures in the pathway to a catastrophic event: an analysis of national incident report data relating to vinca alkaloids', *BMJ quality & safety*, vol. 23, no. 9, pp. 765-72.

French, JR, Raven, B & Cartwright, D 1959, 'The bases of social power', *Classics of organization theory*, vol. 7, pp. 311-20.

Frich, JC & Spehar, I 2018, 'Physician leadership development: towards multidisciplinary programs?', *BMJ Leader*, vol. 2, no. 3, pp. 91-4.

Gaertner, FC, Okamoto, S, Shiga, T, Ito, YM, Uchiyama, Y, Manabe, O et al. 2015, 'FDG PET performed at thyroid remnant ablation has a higher predictive value for long-term survival of high-risk patients with well-differentiated thyroid cancer than radioiodine uptake', *Clinical nuclear medicine*, vol. 40, no. 5, pp. 378-83.

Gandhi, P 2011, 'Clinical research methodology', *Indian J Phar Edu Res*, vol. 45, no. 2, pp. 199-209.

Gandolfi, F & Stone, S 2018, 'Leadership, leadership styles, and servant leadership', *Journal of Management Research*, vol. 18, no. 4, pp. 261-9.

Gardner, WL, Cogliser, CC, Davis, KM & Dickens, MP 2011, 'Authentic leadership: A review of the literature and research agenda', *The Leadership Quarterly*, vol. 22, no. 6, pp. 1120-45.

Gareis, R 2013, 'Changes of Projects by Considering the Principles of Sustainable Development Case Study: Transforming the Project Hospital North', in *Sustainability Integration for Effective Project Management*, IGI Global, pp. 304-20.

Garling, P 2008, *Final Report of the Special Commission of Inquiry: Final report of the Special Commission of Inquiry: Acute Care Services in NSW Public Hospitals, volume 2*, NSW Department of Premier and Cabinet.

Geerts, JM, Goodall, AH & Agius, S 2020, 'Evidence-based leadership development for physicians: a systematic literature review', *Social Science & Medicine*, vol. 246, p. 112709.

General Medical Council 2022, *Good medical practice*, General Medical Council, viewed 10 February 2022, <<u>https://www.gmc-uk.org/ethical-guidance/ethical-guidance/ethical-guidance-for-doctors/leadership-and-management-for-all-doctors</u>>.

Germov, J 2019, 'Challenges to medical dominance', in *Second opinion: An introduction to health sociology*, Oxford University Press, USA, pp. 478-503.

Gerring, J 2016, *Case study research: Principles and practices*, 2nd edn, Cambridge University Press, Cambridge United Kingdom.

Ghani, FA, Derani, NES, Aznam, N, Mohamad, N, Zakaria, SAA & Toolib, SN 2018, 'An empirical investigation of the relationship between transformational, transactional female leadership styles and employee engagement', *Global Business and Management Research*, vol. 10, no. 3, p. 724.

Gharouni Jafari, K & Noorzai, E 2021, 'Selecting the most appropriate project manager to improve the performance of the occupational groups in road construction projects in warm regions', *Journal of Construction Engineering and Management*, vol. 147, no. 10, p. 04021131.

Ghasabeh, MS, Soosay, C & Reaiche, C 2015, 'The emerging role of transformational leadership', *The Journal of Developing Areas*, vol. 49, no. 6, pp. 459-67.

Ghazzawi, K, Shoughari, RE & Osta, BE 2017, 'Situational leadership and its effectiveness in rising employee productivity: A study on North Lebanon organization', *Human Resource Management Research*, vol. 7, no. 3, pp. 102-10.

Gilbert, JW, Johnson, KM, Larkin, GL & Moore, CL 2012, 'Atraumatic headache in US emergency departments: recent trends in CT/MRI utilisation and factors associated with severe intracranial pathology', *Emergency medicine journal*, vol. 29, no. 7, pp. 576-81.

Gilson, L 2016, 'Everyday politics and the leadership of health policy implementation', *Health Systems & Reform*, vol. 2, no. 3, pp. 187-93.

Glinka, B & Hensel, P 2018, 'What Should Be Avoided During Qualitative Research?', in *Qualitative Methodologies in Organization Studies*, Springer, pp. 245-57.

Głód, W 2018, 'Transformational leadership style in the relationship between innovation and efficiency of healthcare units in Poland', *Oeconomia copernicana*, vol. 9, no. 4, pp. 731-53.

Glouberman, S & Zimmerman, B 2016, 'Complicated and Complex Systems: What Would Successful Reform of Medicare Look Like?', in *Changing Health Care in Canada*, University of Toronto Press, Toronto Canada, pp. 21-53.

Goh, SC 2020, 'The influence of Senge's book The Fifth Discipline on an academic career: a research journey into the learning organization and some personal reflections', *The Learning Organization*.

Goldstein, KM, Gierisch, JM, Tucker, M, Williams, JW, Dolor, RJ & Henderson, W 2021, 'Options for Meaningful Engagement in Clinical Research for Busy Frontline Clinicians', *Journal of General Internal Medicine*, vol. 36, no. 7, pp. 2100-4, DOI 10.1007/s11606-020-06587-3, <<u>https://doi.org/10.1007/s11606-020-06587-3</u>>.

Goleman, D 2017, *Leadership that gets results (Harvard business review classics)*, Harvard Business Publishing Corporation, Boston, Massachusetts.

Golla, EJ 2012, *The relationship between transformational and transactional leadership styles and innovation commitment and output at commercial software companies*, University of Phoenix.

Gomes, CM, Kneipp, JM, Kruglianskas, I, da Rosa, LAB & Bichueti, RS 2014, 'Management for sustainability in companies of the mining sector: an analysis of the main factors related with the business performance', *Journal of Cleaner Production*, vol. 84, pp. 84-93.

Gong, CH & He, X 2019, 'Factors Predicting Voluntary and Involuntary Workforce Transitions at Mature Ages: Evidence from HILDA in Australia', *International Journal of Environmental Research and Public Health*, vol. 16, no. 19, ProQuest Central; SciTech Premium Collection, DOI <u>http://dx.doi.org/10.3390/ijerph16193769</u>. Gottfredson, RK, Wright, SL & Heaphy, ED 2020, 'A critique of the Leader-Member Exchange construct: Back to square one', *The Leadership Quarterly*, p. 101385.

Government of Western Australia 2021a, *Diagnostic Imaging Pathways* - *Common Procedures - Computed Tomography (CT)*, viewed 16 September 2021, <<u>http://imagingpathways.health.wa.gov.au/index.php/about-imaging/common-procedures/computed-tomography</u>>.

Government of Western Australia 2021b, *Diagnostic Imaging Pathways - Common Procedures - Nuclear Medicine*, viewed 16 September 2021, <<u>http://imagingpathways.health.wa.gov.au/index.php/about-imaging/common-procedures/nuclear-medicine</u>>.

Government of Western Australia 2021c, *Information for Consumers - Magnetic Resonance Imaging (MRI)*, viewed 16 September 2021, <<u>http://imagingpathways.health.wa.gov.au/index.php/about-imaging/common-procedures/magnetic-resonance-imaging?layout=edit&id=347</u>>.

Government of Western Australia 2021d, *Information for Consumers - Radiation Risks of X-Rays and Scans*, viewed 16 September 2021, <<u>http://imagingpathways.health.wa.gov.au/index.php/consumer-info/general-information-about-diagnostic-imaging/radiation-risks-of-x-rays-and-scans</u>>.

Graen, GB 1976, 'Role-making processes within complex organizations', *Handbook of industrial and organizational psychology*.

Graen, GB & Uhl-Bien, M 1991, 'The transformation of professionals into selfmanaging and partially self-designing contributors: Toward a theory of leadership-making'.

Grant-Smith, DCC & Colley, LK 2018, 'Of 'strong' leadership, crisis communication, and pooper scoopers: Change in the Queensland Public Service under Newman', *Australian journal of public administration*, vol. 77, no. 2, pp. 236–52-52, DOI 10.1111/1467-8500.12311.

Greenleaf, RK 1977, *Servant leadership: A journey into the nature of legitimate power and greatness*, 1st edn, Paulist Press, Mahwah NJ.

Griffin, J & Treanor, D 2017, 'Digital pathology in clinical use: where are we now and what is holding us back?', *Histopathology*, vol. 70, no. 1, pp. 134-45.

Grisaffe, DB, VanMeter, R & Chonko, LB 2016, 'Serving first for the benefit of others: Preliminary evidence for a hierarchical conceptualization of servant leadership', *Journal of Personal Selling & Sales Management*, vol. 36, no. 1, pp. 40-58.

Guttmann, A, Schull, MJ, Vermeulen, MJ & Stukel, TA 2011, 'Association between waiting times and short term mortality and hospital admission after departure from emergency department: population based cohort study from Ontario, Canada', *BMJ*, vol. 342.

Güven, A 2013, 'The Effects Of Manager Behavior On Motivation Of The Workers: An Application On Teachers Working In The National Education Department', *Journal of International Management Educational and Economics Perspectives*, vol. 1, no. 1, pp. 1-20.

Hallward, MC & Bekdash Muellers, H 2020, 'Women's Leadership in Oman: An Intersectional and Transnational Perspective', *Social Politics: International Studies in Gender, State & Society*, vol. 27, no. 2, pp. 361-84, viewed 1/12/2022, DOI 10.1093/sp/jxz027, <<u>https://doi.org/10.1093/sp/jxz027</u>>.

Hamstra, MR 2014, ''Big'men: Male leaders' height positively relates to followers' perception of charisma', *Personality and Individual Differences*, vol. 56, pp. 190-2.

Hanse, JJ, Harlin, U, Jarebrant, C, Ulin, K & Winkel, J 2016, 'The impact of servant leadership dimensions on leader–member exchange among health care professionals', *J Nurs Manag*, vol. 24, no. 2, pp. 228-34.

Haski-Leventhal, D 2020, 'Leading the Change', in *The Purpose-Driven University*, Emerald Publishing Limited, pp. 103-17.

Haslam, SA, Reicher, SD & Platow, MJ 2020, *The new psychology of leadership: ldentity, influence and power*, 2nd edn, Routledge, London United Kingdom.

Hass, KBK 2011, 'How to Manage the Complexities of Large, Diverse Project Teams', *The Good, The Bad and The Complex. Project Perspectives, International Project Management Association*, vol. 33, p. 31, viewed 17 September 2020, <<u>https://www.projecttimes.com/articles/how-to-manage-the-complexities-of-</u> <u>large-diverse-project-</u> teams html2utm_modium_linkedip?

teams.html?utm medium=linkedin&utm medium=linkedin>.

Hassan, MM, Bashir, S & Abbas, SM 2017, 'The impact of project managers' personality on project success in NGOs: The mediating role of transformational leadership', *Project Management Journal*, vol. 48, no. 2, pp. 74-87.

Havig, K, Pharris, A, McLeod, DA, Natale, AP & Miller-Cribbs, J 2020, 'Assessing new child welfare worker competency through social simulation with standardized clients: rubric development and pilot testing', *Journal of Public Child Welfare*, vol. 14, no. 5, pp. 531-52.

Hazzi, O & Maldaon, I 2015, 'A pilot study: Vital methodological issues', *Business: Theory and Practice*, vol. 16, no. 1, pp. 53-62.

Hedley, KM 2021, *Exploring the Effects of Generation and Leadership Style on Nurses' Organizational Commitment*, Walden University.

Helm, E, Krauss, O, Lin, AM, Pointner, A, Schuler, A & Küng, J 2020, 'Process Mining on FHIR-An Open Standards-Based Process Analytics Approach for Healthcare', *Springer*, pp. 343-55.

Hempenstall, A, Tomlinson, J & Bismark, MM 2019, 'Gender inequity in medicine and medical leadership', *Med J Aust*, vol. 211, no. 10, pp. 475-.

Hermanowicz, JC 2013, 'The longitudinal qualitative interview', *Qualitative sociology*, vol. 36, no. 2, pp. 189-208.

Hersey, P, Blanchard, KH & Natemeyer, WE 1979, 'Situational leadership, perception, and the impact of power', *Group & Organization Management*, vol. 4, no. 4, pp. 418-28.

Hoffmeier, A, Sindermann, JR, Scheld, HH & Martens, S 2014, 'Cardiac tumors – diagnosis and surgical treatment', *Deutsches Ärzteblatt International*, vol. 111, no. 12, p. 205.

Holliday, EB, Siker, M, Chapman, CH, Jagsi, R, Bitterman, DS, Ahmed, AA et al. 2018, 'Achieving gender equity in the radiation oncology physician workforce', *Advances in radiation oncology*, vol. 3, no. 4, pp. 478-83.

Hollnagel, E & Braithwaite, J 2019, *Resilient health care*, 1st edn, CRC Press, Boca Raton USA.

Holman, L, Stuart-Fox, D & Hauser, CE 2018, 'The gender gap in science: How long until women are equally represented?', *PLoS Biology*, vol. 16, no. 4, ProQuest Central; SciTech Premium Collection, DOI <u>http://dx.doi.org/10.1371/journal.pbio.2004956</u>.

Homan, AC, Gündemir, S, Buengeler, C & van Kleef, GA 2020, 'Leading diversity: Towards a theory of functional leadership in diverse teams', *Journal of Applied Psychology*, vol. 105, no. 10, p. 1101.

Hornickel, J 2012, 'Consultative Leadership', *Training*, vol. 49, no. 3, pp. 12-,4, ProQuest Central; Social Science Premium Collection.

Hornstein, HA 2015, 'The integration of project management and organizational change management is now a necessity', *International Journal of Project Management*, vol. 33, no. 2, pp. 291-8.

House, RJ & Mitchell, TR 1975, *Path-goal theory of leadership*, Washington University Seattle <<u>https://apps.dtic.mil/sti/pdfs/ADA009513.pdf</u>>.

Howe, AL, King, DS, Ellis, JM, Wells, YD, Wei, Z & Teshuva, KA 2012, 'Stabilising the aged care workforce: an analysis of worker retention and intention', *Australian Health Review*, vol. 36, no. 1, pp. 83-91.

Hsiung, H-H & Bolino, MC 2018, 'The implications of perceived leader favouritism in the context of leader-member exchange relationships', *European Journal of Work and Organizational Psychology*, vol. 27, no. 1, pp. 88-99, DOI 10.1080/1359432X.2017.1395414,

<<u>https://doi.org/10.1080/1359432X.2017.1395414</u>>.

Hsu, S-W & Lamb, P 2020, 'Still in search of learning organization? Towards a radical account of The Fifth Discipline: The Art and Practice of the Learning Organization', *The Learning Organization*.

Hunter, EM, Neubert, MJ, Perry, SJ, Witt, L, Penney, LM & Weinberger, E 2013, 'Servant leaders inspire servant followers: Antecedents and outcomes for employees and the organization', *The Leadership Quarterly*, vol. 24, no. 2, pp. 316-31.

Hutchinson, M & Jackson, D 2013, 'Transformational leadership in nursing:
towards a more critical interpretation', *Nursing Inquiry*, vol. 20, no. 1, pp. 11-22,
rzh, EBSCOhost, DOI 10.1111/nin.12006,
<<u>http://ezproxy.uws.edu.au/login?url=http://search.ebscohost.com/login.aspx?di</u>
rect=true&db=rzh&AN=2011887669&site=ehost-live&scope=site>.

laquinto, BL 2018, '10. A Mixed Methods Approach in Tourism Research', in *Qualitative Methods in Tourism Research*, Channel View Publications, pp. 224-47.

Iliyasu, R & Etikan, I 2021, 'Comparison of quota sampling and stratified random sampling', *Biom. Biostat. Int. J. Rev*, vol. 10, pp. 24-7.

Independent Health Pricing Authority 2019, *Activity Based Funding*, Independent Health Pricing Authority, viewed 10 Feb 2022, <<u>https://www.ihpa.gov.au/what-we-do/activity-based-funding</u>>.

International Institute of Directors and Managers 2020, *The Difference Between Good And Great Leadership*, International Institute of Directors and Managers, viewed 22 October 2020 2020, <<u>https://www.iidmglobal.com/expert_talk/expert-talk-categories/leadership/leader_styles_qualities/id128804-the-difference-between-good-and-great-leadership.html</u>>.

Irfan, M, Khan, SZ, Hassan, N, Hassan, M, Habib, M, Khan, S et al. 2021, 'Role of project planning and project manager competencies on public sector project success', *Sustainability*, vol. 13, no. 3, p. 1421.

Itri, JN & Lawson, LM 2016, 'Ineffective leadership', *Journal of the American College of Radiology*, vol. 13, no. 7, pp. 849-55.

Javanparast, S, Maddern, J, Baum, F, Freeman, T, Lawless, A, Labonté, R et al. 2018, 'Change management in an environment of ongoing primary health care system reform: A case study of Australian primary health care services', *The International journal of health planning and management*, vol. 33, no. 1, pp. e76-e88.

Jones, GR & George, JM 2017, *Essentials of contemporary management*, 7th edn, McGraw-Hill Education.

Jowsey, T, Yen, L, Wells, R & Leeder, S 2011, 'National Health and Hospital Reform Commission final report and patient-centred suggestions for reform', *Aust J Prim Health*, vol. 17, no. 2, pp. 162-8, NLM, DOI 10.1071/py10033.

Kagan, SH & Melendez-Torres, G 2015, 'Ageism in nursing', *J Nurs Manag*, vol. 23, no. 5, pp. 644-50.

Kalaitzi, S, Czabanowska, K, Fowler-Davis, S & Brand, H 2017, 'Women leadership barriers in healthcare, academia and business', *Equality, Diversity and Inclusion: An International Journal*, vol. 36, no. 5, pp. 457-74, Ethnic NewsWatch; GenderWatch; ProQuest Central; Social Science Premium Collection, DOI <u>http://dx.doi.org/10.1108/EDI-03-2017-0058</u>.

Kang, JY, Lee, MK, Fairchild, EM, Caubet, SL, Peters, DE, Beliles, GR et al. 2020, 'Relationships among organizational values, employee engagement, and patient satisfaction in an academic medical center', *Mayo Clinic Proceedings: Innovations, Quality & Outcomes*, vol. 4, no. 1, pp. 8-20.

Kay, A & Boxall, AM 2015, 'Success and failure in public policy: Twin imposters or avenues for reform? Selected evidence from 40 years of health-care reform in Australia', *Australian journal of public administration*, vol. 74, no. 1, pp. 33-41.

Kayser, L, Kushniruk, A, Osborne, RH, Norgaard, O & Turner, P 2015, 'Enhancing the effectiveness of consumer-focused health information technology systems through eHealth literacy: a framework for understanding users' needs', *JMIR human factors*, vol. 2, no. 1, p. e9.

Kea, B, Gamarallage, R, Vairamuthu, H, Fortman, J, Lunney, K, Hendey, GW et al. 2013, 'What is the clinical significance of chest CT when the chest x-ray result is normal in patients with blunt trauma?', *The American journal of emergency medicine*, vol. 31, no. 8, pp. 1268-73.

Keiser, NL & Arthur Jr, W 2021, 'A meta-analysis of the effectiveness of the afteraction review (or debrief) and factors that influence its effectiveness', *Journal of Applied Psychology*, vol. 106, no. 7, p. 1007.

Kelly, C, Tumblety, J & Sheron, N 2016, 'Histories of medical lobbying', *The Lancet*, vol. 388, no. 10055, pp. 1976-7, DOI <u>https://doi.org/10.1016/S0140-6736(16)31860-8</u>,

<http://www.sciencedirect.com/science/article/pii/S0140673616318608>.

Kena, G, Musu-Gillette, L, Robinson, J, Wang, X, Rathbun, A, Zhang, J et al. 2015, 'The Condition of Education 2015. NCES 2015-144', *National Center for Education Statistics*.

Kessler, V 2017, ''Visionaries... psychiatric wards are full of them': Religious terms in management literature', *Verbum et Ecclesia*, vol. 38, no. 2, pp. 1-9.

Khan, J, Jaafar, M, Javed, B, Mubarak, N & Saudagar, T 2020, 'Does inclusive leadership affect project success? The mediating role of perceived psychological empowerment and psychological safety', *International Journal of Managing Projects in Business*.

Kippist, L 2012, *The poisoned chalice: The paradoxical role of the hybrid clincian manager*, University of Western Sydney.

Kippist, L & Fitzgerald, JA 2014, 'Professional identity: enabler or barrier to clinical engagement?', *Employment Relations Record*, vol. 14, no. 2, pp. 27-48.

Kissane-Lee, NA, Yule, S, Pozner, CN & Smink, DS 2016, 'Attending Surgeons' Leadership Style in the Operating Room: Comparing Junior Residents' Experiences and Preferences', *Journal of Surgical Education*, vol. 73, no. 1, pp. 40-4, DOI <u>https://doi.org/10.1016/j.jsurg.2015.08.009</u>, <<u>http://www.sciencedirect.com/science/article/pii/S1931720415002184</u>>.

Kitchin, R & Lauriault, TP 2015, 'Small data in the era of big data', *GeoJournal*, vol. 80, no. 4, pp. 463-75.

Knol, A, Janssen, M & Sol, H 2014, 'A taxonomy of management challenges for developing shared services arrangements', *European Management Journal*, vol. 32, no. 1, pp. 91-103.

Kotter, JP 1995, 'Leading change: Why transformation efforts fail', *harvard business review*, vol. 73, no. 2, pp. 59-67.

Kotter, JP 2008, *Force for change: How leadership differs from management*, Simon and Schuster, New York USA.

Kovach, M 2018, 'An examination of leadership theories in business and sport achievement contexts', *The Journal of Values-Based Leadership*, vol. 11, no. 2, p. 14.

Krepia, V, Katsaragakis, S, Kaitelidou, D & Prezerakos, P 2018, 'Transformational leadership and its evolution in nursing', *Progress in Health Sciences*, vol. 8, pp. 185-90.

Kruskal, JB, Reedy, A, Pascal, L, Rosen, MP & Boiselle, PM 2012, 'Quality initiatives: Lean approach to improving performance and efficiency in a radiology department', *Radiographics*, vol. 32, no. 2, pp. 573-87, Embase, Ovid Technologies, DOI <u>http://dx.doi.org/10.1148/rg.322115128</u>.

Kubik-Huch, RA, Vilgrain, V, Krestin, GP, Reiser, MF, Attenberger, UI, Muellner, AU et al. 2020, 'Women in radiology: gender diversity is not a metric-it is a tool for excellence', *European Radiology*, vol. 30, no. 3, pp. 1644-52, NLM, DOI 10.1007/s00330-019-06493-1.

Kuipers, BS, Higgs, M, Kickert, W, Tummers, L, Grandia, J & Van der Voet, J 2014, 'The management of change in public organizations: A literature review', *Public Administration*, vol. 92, no. 1, pp. 1-20.

Kumar, R 2019, *Research methodology: A step-by-step guide for beginners*, 5th edn, Sage Publications Limited, Carlifonia USA.

Kumaraswamy, KSN & Chitale, C 2012, 'Collaborative knowledge sharing strategy to enhance organizational learning', *The Journal of Management Development*, vol. 31, no. 3, p. 308.

Kuziemsky, C 2016, 'Decision-making in healthcare as a complex adaptive system', *SAGE Publications Sage CA: Los Angeles, CA*, vol. 29, no. 1, pp. 4-7.

Lacity, MC, Khan, SA & Yan, A 2017, 'Review of the empirical business services sourcing literature: an update and future directions', *Outsourcing and offshoring business services*, pp. 499-651.

Larson, L & DeChurch, LA 2020, 'Leading teams in the digital age: Four perspectives on technology and what they mean for leading teams', *The Leadership Quarterly*, vol. 31, no. 1, p. 101377.

Larsson, J, Eriksson, PE, Olofsson, T & Simonsson, P 2015, 'Leadership in civil engineering: Effects of project managers' leadership styles on project performance', *Journal of management in engineering*, vol. 31, no. 6, p. 04015011.

Laschinger, HK & Smith, LM 2013, 'The influence of authentic leadership and empowerment on new-graduate nurses' perceptions of interprofessional

collaboration', *JONA: The Journal of Nursing Administration*, vol. 43, no. 1, pp. 24-9.

Latham, JR 2013, 'A framework for leading the transformation to performance excellence part I: CEO perspectives on forces, facilitators, and strategic leadership systems', *Quality Management Journal*, vol. 20, no. 2, pp. 12-33.

Laver, KE, Prichard, IJ, Cations, M, Osenk, I, Govin, K & Coveney, JD 2018, 'A systematic review of interventions to support the careers of women in academic medicine and other disciplines', *BMJ open*, vol. 8, no. 3, p. e020380, DOI 10.1136/bmjopen-2017-020380,

<<u>https://bmjopen.bmj.com/content/bmjopen/8/3/e020380.full.pdf</u>>.

Lavoie-Tremblay, M, Richer, MC, Marchionni, C, Cyr, G, Biron, AD, Aubry, M et al. 2012, 'Implementation of Evidence-Based Practices in the Context of a Redevelopment Project in a Canadian Healthcare Organization', *Journal of Nursing Scholarship*, vol. 44, no. 4, pp. 418-27.

Lenthall, S, Wakerman, J, Opie, T, Dunn, S, MacLeod, M, Dollard, M et al. 2011, 'Nursing workforce in very remote Australia, characteristics and key issues', *The Australian journal of rural health*, vol. 19, no. 1, pp. 32-7, DOI 10.1111/j.1440-1584.2010.01174.x.

Leroy, H, Palanski, ME & Simons, T 2012, 'Authentic leadership and behavioral integrity as drivers of follower commitment and performance', *Journal of Business Ethics*, vol. 107, no. 3, pp. 255-64.

Lewin, K 1951, 'Field theory in social science: selected theoretical papers (Edited by Dorwin Cartwright.)'.

Lewin, K, Lippitt, R & White, RK 1939, 'Patterns of aggressive behavior in experimentally created "social climates"', *The Journal of Social Psychology*, vol. 10, no. 2, pp. 269-99.

Lewiss, RE, Silver, JK, Bernstein, CA, Mills, AM, Overholser, B & Spector, ND 2020, 'Is academic medicine making mid-career women physicians invisible?', *Journal of Women's Health*, vol. 29, no. 2, pp. 187-92.

Lexa, FJ & Fessell, DP 2019, 'Medical Leadership: A Manifesto for Radiologists', *Journal of the American College of Radiology*, vol. 16, no. 3, pp. 380-3.

Li, J, Long, X, Wang, X, Fang, F, Lv, X, Zhang, D et al. 2021, 'Radiology indispensable for tracking COVID-19', *Diagnostic and interventional imaging*, vol. 102, no. 2, pp. 69-75.

Li, L, Davis, M, Kim, N, Lipka, S, Branson, B, Amport, S et al. 2022, 'Clinical Redesign: An Innovative Approach to Leading Change at an Academic Healthcare System', *Journal of Healthcare Management*, vol. 67, no. 1, pp. 13-24.

Liang, ZM, Leggat, SG, Howard, PF & Koh, L 2013, 'What makes a hospital manager competent at the middle and senior levels?', *Australian Health Review*, vol. 37, no. 5, pp. 566-73.

Lin, P 2021, 'Activity Data, South Western Sydney LHD, NSW Health', 14 June 2021.

Liphadzi, M, Aigbavboa, C & Thwala, W 2017, 'A theoretical perspective on the difference between leadership and management', *Procedia Engineering*, vol. 196, pp. 478-82.

Liu, Y 2015, 'The review of empowerment leadership', *Open Journal of Business and Management*, vol. 3, no. 04, p. 476.

Liu, Z, Venkatesh, S, Murphy, SE & Riggio, RE 2020, 'Leader development across the lifespan: A dynamic experiences-grounded approach', *The Leadership Quarterly*, p. 101382, DOI <u>https://doi.org/10.1016/j.leaqua.2020.101382</u>, <<u>http://www.sciencedirect.com/science/article/pii/S1048984320300096</u>>.

Long, JC, Cunningham, FC & Braithwaite, J 2013, 'Bridges, brokers and boundary spanners in collaborative networks: a systematic review', *BMC Health Services Research*, vol. 13, no. 1, p. 158, DOI 10.1186/1472-6963-13-158, <<u>https://doi.org/10.1186/1472-6963-13-158</u>>.

Lord, RG, De Vader, CL & Alliger, GM 1986, 'A meta-analysis of the relation between personality traits and leadership perceptions: An application of validity generalization procedures', *Journal of Applied Psychology*, vol. 71, no. 3, p. 402.

Luhmann, N 2018, *Trust and power*, John Wiley & Sons.

Lunenburg, F 2013, 'Leadership versus managment: A key distinction – at least in theory', *Main Issues Of Pedagogy And Psychology*, vol. 3, no. 3, pp. 15-8.

Luthans, F, Youssef, CM & Avolio, BJ 2015, *Psychological capital and beyond*, Oxford University Press, USA.

Luzinski, C 2012, 'Transformational Leadership and Navigating Change', *Journal of Nursing Administration*, vol. 42, no. 12, pp. 543-4.

MacCarrick, GR 2014, 'Professional medical leadership: a relational training model', *Leadership in Health Services*, vol. 27, no. 4, pp. 343-54, ProQuest Central; Social Science Premium Collection.

Mafle Ferreira Duarte, LG, Sahakian, M & Ferreira Neto, JL 2021, 'Empowerment in an ecovillage: Unveiling the role of power relations in social practices', *Community Development*, pp. 1-15.

Mafora, PP 2020, *Leadership at public hospitals: a case study of the Matlala District Hospital.*

Magableh, GM 2021, 'Supply Chains and the COVID-19 Pandemic: A Comprehensive Framework', *European Management Review*, p. 10.1111/emre.12449, PMC, DOI 10.1111/emre.12449, <<u>https://www.ncbi.nlm.nih.gov/pmc/articles/PMC8014293/</u>>.

Maghsoudi, T, Cascon-Pereira, R & Beatriz Hernández Lara, A 2020, 'The Role of Collaborative Healthcare in Improving Social Sustainability: A Conceptual Framework', *Sustainability*, vol. 12, no. 8, p. 3195.

Maile, E, McKimm, J & Till, A 2019, 'Exploring medical leader identity and its formation', *Leadership in Health Services*.

Mak, K-HM, Kippist, L, Sloan, T & Eljiz, K 2019, 'What is the professional identity of allied health managers?', *Asia-Pacific Journal of Health Management*, vol. 14, no. 1, pp. 58-.

Malik, SH, Aziz, S & Hassan, H 2014, 'Leadership behavior and acceptance of leaders by subordinates: Application of path goal theory in telecom sector', *International Journal of Trade, Economics and Finance*, vol. 5, no. 2, p. 170.

Mangal, S & Mangal, S 2013, *Research methodology in behavioural sciences*, PHI Learning Pvt. Ltd.

Mango, E 2018, 'Rethinking leadership theories', *Open Journal of Leadership*, vol. 7, no. 01, p. 57.

Mangold, K, Denke, NJ, Gorombei, D, Ostroski, TL & Root, L 2014, 'Principles of successful partnerships', *Nursing administration quarterly*, vol. 38, no. 4, pp. 340-7.

Mann, A & DiPrete, TA 2013, 'Trends in gender segregation in the choice of science and engineering majors', *Social Science Research*, vol. 42, no. 6, pp. 1519-41, DOI <u>https://doi.org/10.1016/j.ssresearch.2013.07.002</u>, <<u>http://www.sciencedirect.com/science/article/pii/S0049089X13001051</u>>.

Mann, RD 1959, 'A review of the relationships between personality and performance in small groups', *Psychological Bulletin*, vol. 56, no. 4, p. 241.

Marino, D & Quattrone, G 2019, 'A proposal for a new index to evaluate hospital resource allocation: The case of Italian NHS rationalisation', *European Research*

on Management and Business Economics, vol. 25, no. 1, pp. 23-9, DOI <u>https://doi.org/10.1016/j.iedeen.2018.10.001</u>, https://www.sciencedirect.com/science/article/pii/S244488341830086X>.

Marmor, YN, Kemp, BJ, Huschka, TR, Ruter, RL, McConnell, DM & Rohleder, TR 2013, 'Improving patient access in nuclear medicine: a case study of PET scanner scheduling', *Quality Management in Healthcare*, vol. 22, no. 4, pp. 293-305.

Martínez-Costa, M, Jiménez-Jiménez, D & Dine Rabeh, HA 2019, 'The effect of organisational learning on interorganisational collaborations in innovation: an empirical study in SMEs', *Knowledge Management Research & Practice*, vol. 17, no. 2, pp. 137-50.

Mascia, D, Morandi, F & Cicchetti, A 2014, 'Hospital restructuring and physician job satisfaction: An empirical study', *Health Policy*, vol. 114, no. 2, pp. 118-27, DOI <u>https://doi.org/10.1016/j.healthpol.2013.11.001</u>, https://www.sciencedirect.com/science/article/pii/S0168851013002893>.

Matalon, SA, Howard, SA, Gaviola, GC, Johnson, OW, Phillips, CH, Smith, SE et al. 2018, 'Customized residency leadership tracks: a review of what works, what we're doing and ideas for the future', *Current problems in diagnostic radiology*, vol. 47, no. 6, pp. 359-63.

Mayowa-Adebara, O 2018, 'The influence of leadership style, organizational justice and human capital development on employee commitment in university libraries in south-west, Nigeria', *Library Philosophy & Practice*.

McCullough, K, Whitehead, L, Bayes, S, Williams, A & Cope, V 2020, 'The delivery of Primary Health Care in remote communities: A Grounded Theory study of the perspective of nurses', *International Journal of Nursing Studies*, vol. 102, p. 103474.

McGivern, G, Currie, G, Ferlie, E, Fitzgerald, L & Waring, J 2015, 'Hybrid manager–professionals'identity work: The maintenance and hybridization of medical professionalism in managerial contexts', *Public Administration*, vol. 93, no. 2, pp. 412-32.

McGregor, D 1960, 'The human side of enterprise', New York, vol. 21, p. 166.

McKenna, E 2020, *Business psychology and organizational behaviour*, 6th edn, Routledge, London UK.

McMillan, K & Perron, A 2020, 'Change fatigue in nurses: A qualitative study', *Journal of Advanced Nursing*, vol. 76, no. 10, pp. 2627-36, DOI 10.1111/jan.14454.

Meeks Jr, JB 2020, *The Leader's Role in Motivating Public Sector Workers*, South Carolina USA, Liberty University.

Mehregan, MR, Jamporazmey, M, Hosseinzadeh, M & Kazemi, A 2012, 'An integrated approach of critical success factors (CSFs) and grey relational analysis for ranking KM systems', *Procedia-Social and Behavioral Sciences*, vol. 41, pp. 402-9.

Mehta, S, Maheshwari, G & Sharma, S 2014, 'Role of leadership in leading successful change: An empirical study', *Journal of Contemporary Management Research*, vol. 8, no. 2.

Meng, J & Berger, BK 2019, 'The impact of organizational culture and leadership performance on PR professionals' job satisfaction: Testing the joint mediating effects of engagement and trust', *Public Relations Review*, vol. 45, no. 1, pp. 64-75.

Meng, Q, Mills, A, Wang, L & Han, Q 2019, 'What can we learn from China's health system reform?', *BMJ*, vol. 365, p. I2349, DOI 10.1136/bmj.I2349, <<u>http://www.bmj.com/content/365/bmj.I2349.abstract</u>>.

Merriam, SB & Grenier, RS 2019, *Qualitative research in practice: Examples for discussion and analysis*, John Wiley & Sons.

Mintz, LJ & Stoller, JK 2014, 'A systematic review of physician leadership and emotional intelligence', *Journal of Graduate Medical Education*, vol. 6, no. 1, pp. 21-31.

Morgan, JA 2020, *Was That in An Email? An Electronic Tool to Improve Nursing Communication Satisfaction*, South Dakota USA, South Dakota State University.

Morley, C, Unwin, M, Peterson, GM, Stankovich, J & Kinsman, L 2018, 'Emergency department crowding: a systematic review of causes, consequences and solutions', *PloS one*, vol. 13, no. 8, p. e0203316.

Morrow, V & Crivello, G 2015, 'What is the value of qualitative longitudinal research with children and young people for international development?', *International Journal of Social Research Methodology*, vol. 18, no. 3, pp. 267-80.

Morton, MJ & Korley, FK 2012, 'Head computed tomography use in the emergency department for mild traumatic brain injury: integrating evidence into practice for the resident physician', *Annals of emergency medicine*, vol. 60, no. 3, pp. 361-7.

Mousa, M, Boyle, J, Skouteris, H, Mullins, AK, Currie, G, Riach, K et al. 2021, 'Advancing women in healthcare leadership: A systematic review and metasynthesis of multi-sector evidence on organisational interventions', *EClinicalMedicine*, vol. 39, p. 101084, DOI https://doi.org/10.1016/j.eclinm.2021.101084, <https://www.sciencedirect.com/science/article/pii/S2589537021003643>.

Muenjohn, N & Armstrong, A 2015, 'Transformational leadership: The influence of culture on the leadership behaviours of expatriate managers', *International Journal of Business and information*, vol. 2, no. 2.

Müller, JM, Buliga, O & Voigt, K-I 2021, 'The role of absorptive capacity and innovation strategy in the design of industry 4.0 business Models-A comparison between SMEs and large enterprises', *European Management Journal*, vol. 39, no. 3, pp. 333-43.

Müller, R 2017, *Project governance*, 1st edn, Routledge, London UK.

Munir, F & Aboidullah, M 2018, 'Gender Differences in Transformational Leadership Behaviors of School Principals and Teachers' Academic Effectiveness', *Bulletin of Education and Research*, vol. 40, no. 1, pp. 99-113.

National Institute for Health Research 2013, *New Evidence on Management and Leadership*, National Health Service UK, viewed 10 April 2015, <<u>http://www.nets.nihr.ac.uk/ data/assets/pdf file/0003/95655/New-Evidence-on-Management-and-Leadership.pdf</u>>.

Neider, LL & Schriesheim, CA 2011, 'The Authentic Leadership Inventory (ALI): Development and empirical tests', *The Leadership Quarterly*, vol. 22, no. 6, pp. 1146-64, DOI <u>https://doi.org/10.1016/j.leaqua.2011.09.008</u>, <<u>http://www.sciencedirect.com/science/article/pii/S104898431100155X</u>>.

Nging, TK & Yazdanifard, R 2015, 'The general review of how different leadership styles cause the transformational change efforts to be successful', *International Journal of Management, Accounting and Economics*, vol. 2, no. 9, pp. 1130-40.

Nickel, S & Schmidt, U-A 2009, 'Process improvement in hospitals: a case study in a radiology department', *Quality Management in Healthcare*, vol. 18, no. 4, pp. 326-38.

Nie, D & Lämsä, A-m 2015, 'The Leader-Member Exchange Theory in the Chinese Context and the Ethical Challenge of Guanxi: JBE', *Journal of Business Ethics*, vol. 128, no. 4, pp. 851-61, Arts Premium Collection; ProQuest Central; Social Science Premium Collection, DOI <u>http://dx.doi.org/10.1007/s10551-013-1983-9</u>.

Northouse, PG 2012, *Leadership: Theory and Practice*, 6th edn, SAGE Publications Ltd (CA), Califonia USA.

Northouse, PG 2019, *Leadership: Theory and Practice*, 8th edn, Sage Publications, Michigan USA.

Northouse, PG 2021, *Leadership: Theory and practice*, 9th edn, Sage publications, Michigan USA.

NSW Health 1997, *Annual report 1996-1997*, NSW Health, Sydney<<u>http://www.health.nsw.gov.au/publications/Publications/Annual-Report-1996-97/Annual-Report-1996-97.pdf</u>>.

NSW Health 2015, *Local Health Districts and specialty networks*, NSW Health, viewed 10 April 2015, <<u>http://www.health.nsw.gov.au/policies/manuals/Documents/corporate-governance-compendium-section3.pdf</u>>.

NSW Health 2018, *Program Management Office*, NSW Health, viewed 30 November 2020, <<u>https://www.health.nsw.gov.au/Performance/Pages/program-management-office.aspx</u>>.

Nyikuri, M, Tsofa, B, Barasa, E, Okoth, P & Molyneux, S 2015, 'Crises and resilience at the frontline—public health facility managers under devolution in a Sub-County on the Kenyan Coast', *PloS one*, vol. 10, no. 12, p. e0144768.

Nzinga, J, McGivern, G & English, M 2018, 'Examining clinical leadership in Kenyan public hospitals through the distributed leadership lens', *Health Policy and Planning*, vol. 33, no. suppl_2, pp. ii27-ii34.

O'Connor, M 2013, 'Communication: Essential Strategies for Success', *Journal of Nursing Administration*, vol. 43, no. 6, pp. 323-5, DOI 10.1097/NNA.0b013e3182942ccf,

<<u>http://journals.lww.com/jonajournal/Fulltext/2013/06000/Communication Esse</u> <u>ntial Strategies for Success.6.aspx</u>>.

Ojo, Y 2019, *The Use of Hourly Rounding Using the 4Ps Decreases the Fall Rate Among Older Adults in a Skilled Nursing Facility*, Brandman University.

Okello, DR & Gilson, L 2015, 'Exploring the influence of trust relationships on motivation in the health sector: a systematic review', *Human resources for health*, vol. 13, no. 1, p. 16.

Okere, OO & Olorunfemi, M 2018, 'Academic library leadership in South West Nigeria and recession: An exploratory study', *International Information & Library Review*, vol. 50, no. 3, pp. 202-11.

Okpala, P 2018, 'Balancing quality healthcare services and costs through collaborative leadership', *Journal of Healthcare Management*, vol. 63, no. 6, pp. e148-e57.

Olley, R 2021, 'A focussed literature review of power and influence leadership theories', *Asia Pacific Journal of Health Management*, vol. 16, no. 2, pp. 7-17.

Oostra, RD 2016, 'Physician leadership: a central strategy to transforming healthcare', *Frontiers of health services management*, vol. 32, no. 3, pp. 15-26.

Orr, T & Cleveland-Innes, M 2015, 'Appreciative leadership: Supporting education innovation', *International Review of Research in Open and Distributed Learning*, vol. 16, no. 4.

Ovseiko, PV, Chapple, A, Edmunds, LD & Ziebland, S 2017, 'Advancing gender equality through the Athena SWAN Charter for Women in Science: an exploratory study of women's and men's perceptions', *Health research policy and systems*, vol. 15, no. 1, pp. 1-13.

Papa, A, Dezi, L, Gregori, GL, Mueller, J & Miglietta, N 2018, 'Improving innovation performance through knowledge acquisition: the moderating role of employee retention and human resource management practices', *Journal of Knowledge Management*, pp. 1-18.

Park, S & Kim, E-J 2018, 'Fostering organizational learning through leadership and knowledge sharing', *Journal of Knowledge Management*, vol. 22, no. 6, pp. 1408-23.

Parris, DL & Peachey, JW 2013, 'A Systematic Literature Review of Servant Leadership Theory in Organizational Contexts', *Journal of Business Ethics*, vol. 113, no. 3, pp. 377-93, DOI 10.1007/s10551-012-1322-6, <<u>https://doi.org/10.1007/s10551-012-1322-6</u>>.

Parthasarthy, R 2018, 'Lean Six Sigma for Hospitals: Improving Patient Safety, Patient Flow and the Bottom Line', *Quality Progress*, vol. 51, no. 6, pp. 61-.

Patel, V, Barker, W & Siminerio, E 2015, 'Trends in consumer access and use of electronic health information', *ONC Data Brief*, vol. 30.

Patrick, R 2014, 'Working on welfare: Findings from a qualitative longitudinal study into the lived experiences of welfare reform in the UK', *Journal of Social Policy*, vol. 43, no. 4, pp. 705-25.

Peirson, L, Ciliska, D, Dobbins, M & Mowat, D 2012, 'Building capacity for evidence informed decision making in public health: a case study of organizational change', *BMC Public Health*, vol. 12, no. 1, p. 137, < >.

Perez, J 2021, 'Leadership in Healthcare: Transitioning From Clinical Professional to Healthcare Leader', *Journal of Healthcare Management*, vol. 66, no. 4, pp. 280-302.

Peus, C, Wesche, JS, Streicher, B, Braun, S & Frey, D 2012, 'Authentic leadership: An empirical test of its antecedents, consequences, and mediating mechanisms', *Journal of Business Ethics*, vol. 107, no. 3, pp. 331-48.

Phalak, K, Gerlach, K & Parikh, JR 2020, 'Community outreach and integration of breast radiologists', *Clinical Imaging*, vol. 66, pp. 143-6.

Phillips, AW, Smith, SG & Straus, CM 2013, 'The role of radiology in preclinical anatomy: a critical review of the past, present, and future', *Academic Radiology*, vol. 20, no. 3, pp. 297-304. e1.

Pines, JM, Hilton, JA, Weber, EJ, Alkemade, AJ, Al Shabanah, H, Anderson, PD et al. 2011, 'International perspectives on emergency department crowding', *Academic Emergency Medicine*, vol. 18, no. 12, pp. 1358-70.

Pohjola, T, Suhonen, M, Mattila, KK & Meretoja, R 2016, 'The work done in healthcare projects', *Journal of nursing*.

Qamar, SR, Khurshid, K, Jalal, S, McInnes, MDF, Probyn, L, Finlay, K et al. 2020, 'Gender Disparity Among Leaders of Canadian Academic Radiology Departments', *AJR. American Journal of Roentgenology*, vol. 214, no. 1, pp. 3-9, NLM, DOI 10.2214/ajr.18.20992.

Ratnapalan, S & Lang, D 2020, 'Health care organizations as complex adaptive systems', *The health care manager*, vol. 39, no. 1, pp. 18-23.

Rebelo, T, Lourenço, PR & Dimas, ID 2019, 'The journey of team learning since "The Fifth Discipline"', *The Learning Organization*.

Redknap, R, Twigg, D, Rock, D & Towell, A 2015, 'N ursing practice environment: A strategy for mental health nurse retention?', *International journal of mental health nursing*, vol. 24, no. 3, pp. 262-71.

Regan, S, Laschinger, HK & Wong, CA 2016, 'The influence of empowerment, authentic leadership, and professional practice environments on nurses' perceived interprofessional collaboration', *J Nurs Manag*, vol. 24, no. 1, pp. E54-E61.

Rittinger, K 2014, 'Impact of Female Leadership on Employee engagement in German small and medium-sized Family Companies'.

Robbins, S, Bergman, R & Stagg, I 2014, *Management EBook*, Pearson Australia Pty Limited.

Robbins, S & Judge, T 2017, *Organizational behavior*, 7th edn, Pearson Education, Inc., Boston USA.

Roberts, JP, Fisher, TR, Trowbridge, MJ & Bent, C 2016, 'A design thinking framework for healthcare management and innovation', *Healthcare*, vol. 4, no. 1, pp. 11-4, DOI <u>https://doi.org/10.1016/j.hjdsi.2015.12.002</u>, <<u>https://www.sciencedirect.com/science/article/pii/S221307641500113X</u>>.

Robinson, OC 2014, 'Sampling in Interview-Based Qualitative Research: A Theoretical and Practical Guide', *Qualitative Research in Psychology*, vol. 11, no. 1, pp. 25-41, DOI 10.1080/14780887.2013.801543, <<u>https://doi.org/10.1080/14780887.2013.801543</u>>.

Rodriguez, RM, Anglin, D, Langdorf, MI, Baumann, BM, Hendey, GW, Bradley, RN et al. 2013, 'NEXUS chest: validation of a decision instrument for selective chest imaging in blunt trauma', *JAMA surgery*, vol. 148, no. 10, pp. 940-6.

Roof, R 2014, 'Authentic leadership questionnaire (ALQ) psychometrics', *Asian Journal of Business Ethics*, vol. 3, no. 1, pp. 57-64.

Rosenbaum, D, More, E & Steane, P 2018, 'Planned organisational change management: Forward to the past? An exploratory literature review', *Journal of Organizational Change Management*.

Royal Australia and New Zealand College of Radiology 2019, *Radiology at a Glance Australia*, Royal Australia and New Zealand College of Radiology, viewed 10 Feb 2022, <<u>https://www.ranzcr.com/college/document-library/radiology-at-a-glance-australia</u>>.

Sacks, D, van Overhagen, H, van Zwam, WH, Radvany, MG, Marx, MV, Morgan, RA et al. 2019, 'The Role of Interventional Radiologists in Acute Ischemic Stroke Interventions: A Joint Position Statement from the Society of Interventional Radiology, the Cardiovascular and Interventional Radiology Society of Europe, and the Interventional Radiology Society of Australasia', Journal of Vascular & Interventional Radiology, vol. 30, no. 2, pp. 131-3, NLM, DOL 10.1016/j.jvir.2018.09.035.

Sadiq, AW 2020, *Influence of Leadership Practices on Organizational Safety Performance*, University of Phoenix.

Saira, S, Mansoor, S & Ali, M 2020, 'Transformational leadership and employee outcomes: the mediating role of psychological empowerment', *Leadership and Organization Development Journal*, vol. 42, no. 1, pp. 130-43.

Salary, TK 2019, *The role of transformational and transactional leaders on the intrinsic and extrinsic motivation among employees in Atlanta: A phenomenological study*, Northcentral University.

Samaibekova, Z & Kocherbaeva, A 2019, 'Leveraging universally desirable and undesirable leadership styles', vol. 359, pp. 30-4.

Sarid, A 2016, 'Integrating leadership constructs into the Schwartz value scale: Methodological implications for research', *Journal of Leadership Studies*, vol. 10, no. 1, pp. 8-17.

Sarwar, A & Imran, MK 2019, 'Exploring Women's multi-level career prospects in Pakistan: Barriers, interventions, and outcomes', *Frontiers in psychology*, vol. 10, p. 1376.

Savage, M, Savage, C, Brommels, M & Mazzocato, P 2020, 'Medical leadership: boon or barrier to organisational performance? A thematic synthesis of the literature', *BMJ open*, vol. 10, no. 7, p. e035542.

Scharmer, CO & Kaufer, K 2013, *Leading from the emerging future: From ego-system to eco-system economies*, Berrett-Koehler Publishers.

Schnackenberg, AK & Tomlinson, EC 2016, 'Organizational transparency: A new perspective on managing trust in organization-stakeholder relationships', *Journal of management*, vol. 42, no. 7, pp. 1784-810.

Schneper, WD 2020, 'Common thread: The Fifth Discipline from a liberal arts college perspective', *The Learning Organization*.

Scott, V, Mathews, V & Gilson, L 2012, 'Constraints to implementing an equitypromoting staff allocation policy: understanding mid-level managers' and nurses' perspectives affecting implementation in South Africa', *Health Policy and Planning*, vol. 27, no. 2, pp. 138-46.

Sebastian, A, Fulop, L, Dadich, A, Fitzgerald, A, Kippist, L & Smyth, A 2014, 'Health LEADS Australia and implications for medical leadership', *Leadership in Health Services*, vol. 27, no. 4, pp. 355-70, ProQuest Central; Social Science Premium Collection.

See, EJ, Lussier, S & Jones, D 2021, 'Understanding factors contributing to the underrepresentation of female co-authors in intensive care publications', *Australian Critical Care*, vol. 34, no. 6, pp. 519-21.

Seelen, MT, Edwards, K, Bruce, A, Rogers, K, Landrigan, C, Dufrene, C et al. 2020, 'Reducing undocumented supplies in hybrid ORs through multidisciplinary process improvement', *Perioperative Care and Operating Room Management*, vol. 18, p. 100087.

Seltzer, S, Kelly, P, Adams, D, Chiango, B, Viera, M, Fener, E et al. 1994, 'Expediting the turnaround of radiology reports: use of total quality management to facilitate radiologists' report signing', *AJR. American Journal of Roentgenology*, vol. 162, no. 4, pp. 775-81. Serrador, P & Turner, R 2015, 'The relationship between project success and project efficiency', *Project Management Journal*, vol. 46, no. 1, pp. 30-9.

Shah, CJ, Sullivan, JR, Gonyo, MB, Wadhwa, A & DuBois, MS 2013, 'Practice policy and quality initiatives using lean principles to improve screening mammography workflow', *Radiographics*, vol. 33, no. 5, pp. 1505-17, Embase, Ovid Technologies, DOI <u>http://dx.doi.org/10.1148/rg.335135017</u>, <<u>http://acs.hcn.com.au?acc=36422&url=http://ovidsp.ovid.com/ovidweb.cgi?T=J</u> <u>S&CSC=Y&NEWS=N&PAGE=fulltext&D=emed11&AN=23813321</u><97. >>.

Shaikh, AT, Farhan, SA, Siddiqi, R, Fatima, K, Siddiqi, J & Khosa, F 2019, 'Disparity in leadership in neurosurgical societies: a global breakdown', *World neurosurgery*, vol. 123, pp. 95-102.

Shatrov, K, Pessina, C, Huber, K, Thomet, B, Gutzeit, A & Blankart, CR 2021, 'Improving health care from the bottom up: Factors for the successful implementation of kaizen in acute care hospitals', *PloS one*, vol. 16, no. 9, p. e0257412.

Shaw, AS & Dixon, AK 2015, 'Computed tomography', eLS.

Shaw, G, Smith, N, Khowaja, A, Mitton, C, Denis, J-L & Lovato, C 2019, 'Strategies to increase physician engagement in acute care settings: a scoping review', *Journal of health organization and management*.

Shen, H 2013, 'MIND THE GENDER GAP', *Nature*, vol. 495, no. 7439, pp. 22-4, ProQuest Central; SciTech Premium Collection.

Siegal, D, Stratchko, LM & DeRoo, C 2017, 'The role of radiology in diagnostic error: a medical malpractice claims review', *Diagnosis*, vol. 4, no. 3, pp. 125-31.

Silva, A 2016, 'What is leadership?', *Journal of Business Studies Quarterly*, vol. 8, no. 1, p. 1.

Silvius, G & Schipper, R 2015, 'Developing a maturity model for assessing sustainable project management', *The Journal of Modern Project Management*, vol. 3, no. 1.

Simons, J & Lomax, H 2021, 'Leadership and vision', in *Leading managing caring:*, Routledge, pp. 61-88.

Simpson, S 2012, 'Top Ten Leadership Skills–Book', Ventus Publishing ApS.

Smith, A, Latter, S & Blenkinsopp, A 2014, 'Safety and quality of nurse independent prescribing: a national study of experiences of education, continuing professional development clinical governance', *Journal of Advanced Nursing*, vol. 70, no. 11, pp. 2506-17.

Smith, RH 2015, 'The Best Time to Plant a Tree Was 20 Years Ago, No Matter', *Psychology Today*, viewed 18 September 2020, <<u>https://www.psychologytoday.com/za/blog/joy-and-pain/201504/the-best-time-plant-tree-was-20-years-ago-no-matter</u>>.

Smith, TD, Eldridge, F & DeJoy, DM 2016, 'Safety-specific transformational and passive leadership influences on firefighter safety climate perceptions and safety behavior outcomes', *Safety Science*, vol. 86, pp. 92-7, DOI <u>https://doi.org/10.1016/j.ssci.2016.02.019</u>, http://www.sciencedirect.com/science/article/pii/S0925753516000631>.

Soomro, ZA, Ahmed, J, Muhammad, R, Hayes, D & Shah, MH 2017, 'Critical success factors in implementing an e-rostering system in a healthcare organisation', *Health Services Management Research*, vol. 31, no. 3, pp. 130-7, viewed 2020/10/22, DOI 10.1177/0951484817745695, https://doi.org/10.1177/0951484817745695,

Sorensen, R, Paull, G, Magann, L & Davis, J 2013, 'Managing between the agendas: implementing health care reform policy in an acute care hospital', *Journal of health organization and management*.

South Western Sydney Local Health District 2019, *About Us*, South Western Sydney Local Health District, viewed 27 Jul 2021, <<u>https://www.swslhd.health.nsw.gov.au/about.html</u>>.

South Western Sydney Local Health District 2020, *Services*, South Western Sydney Local Health District, viewed 16 September 2021, <<u>https://www.swslhd.health.nsw.gov.au/services.html</u>>.

Sparh, P 2016, 'What is transactional leadership? How structure leads to results. Leadership is learned', *Retrieved on*, vol. 22, no. 03, p. 2017, <<u>https://online.stu.edu/articles/education/what-is-transactional-</u> leadership.aspx>.

Sparrowe, RT & Liden, RC 1997, 'Process and structure in leader-member exchange', *Academy of management review*, vol. 22, no. 2, pp. 522-52.

Spears, LC & Lawrence, M 2016, *Practicing servant-leadership: Succeeding through trust, bravery, and forgiveness*, 1st edn, John Wiley & Sons, Germany.

Specchia, ML, Cozzolino, MR, Carini, E, Di Pilla, A, Galletti, C, Ricciardi, W et al. 2021, 'Leadership Styles and Nurses' Job Satisfaction. Results of a Systematic Review', *International Journal of Environmental Research and Public Health*, vol. 18, no. 4, p. 1552.

Spurgeon, P, Clark, J, Ham, C & Keogh, B 2017, *Medical Leadership: From the Dark Side to Centre Stage: From the Dark Side to Centre Stage*, CRC Press, London UK.

Stanley, D 2016, *Clinical leadership in nursing and healthcare: Values into action*, 2nd edn, John Wiley & Sons, West Sussex UK.

Stanley, D, Blanchard, D, Hohol, A, Hutton, M & McDonald, A 2017, 'Health professionals' perceptions of clinical leadership. A pilot study', *Cogent Medicine*, vol. 4, no. 1, p. 1321193.

Stanley, D & Stanley, K 2018, 'Clinical leadership and nursing explored: A literature search', *Journal of clinical nursing*, vol. 27, no. 9-10, pp. 1730-43.

Stefanini, A, Aloini, D, Benevento, E, Dulmin, R & Mininno, V 2020, 'A data-driven methodology for supporting resource planning of health services', *Socio-Economic Planning Sciences*, vol. 70, p. 100744.

Steinberg, DM 2015, *The social work student's research handbook*, 2nd edn, Routledge, Oxon and New York.

Stensaker, IG & Meyer, CB 2012, 'Change experience and employee reactions: developing capabilities for change', *Personnel review*.

Stephens, EH, Heisler, CA, Temkin, SM & Miller, P 2020, 'The current status of women in surgery: how to affect the future', *JAMA surgery*, vol. 155, no. 9, pp. 876-85.

Stewart, DW & Shamdasani, PN 2014, *Focus groups: Theory and practice*, 3rd edn, vol. 20, Sage publications, Califonia USA.

Stogdill, RM 1948, 'Personal factors associated with leadership: A survey of the literature', *The Journal of psychology*, vol. 25, no. 1, pp. 35-71.

Stogdill, RM & Bass, BM 1981, 'Stogdill's handbook of leadership'.

Stoller, FL 2012, 'Innovation as the hallmark of effective leadership', in *Leadership in English Language Education*, Routledge, pp. 87-98.

Stouten, J, Rousseau, DM & De Cremer, D 2018, 'Successful organizational change: Integrating the management practice and scholarly literatures', *Academy of Management Annals*, vol. 12, no. 2, pp. 752-88.

Strauss, SE 2011, *Evidence-based medicine: how to practice and teach it;[includes CD-ROM]*, Churchill Livingstone, Elsevier.

Suárez-Barraza, MF & Miguel-Davila, JA 2020, 'Kaizen–Kata, a problem-solving approach to public service health care in Mexico. A multiple-case study', *International Journal of Environmental Research and Public Health*, vol. 17, no. 9, p. 3297.

Sun, BC, Hsia, RY, Weiss, RE, Zingmond, D, Liang, L-J, Han, W et al. 2013, 'Effect of emergency department crowding on outcomes of admitted patients', *Annals of emergency medicine*, vol. 61, no. 6, pp. 605-11. e6.

Sydney Local Health District 2017, *Fellow Program*, Sydney Local Health District, viewed 11 Nov 2020, <<u>https://www.slhd.nsw.gov.au/rpa/neonatal/fellow.html</u>>.

Taherdoost, H 2016, 'Sampling methods in research methodology; how to choose a sampling technique for research', *How to Choose a Sampling Technique for Research (April 10, 2016)*.

Tajasom, A, Hung, DKM, Nikbin, D & Hyun, SS 2015, 'The role of transformational leadership in innovation performance of Malaysian SMEs', *Asian Journal of Technology Innovation*, vol. 23, no. 2, pp. 172-88, DOI 10.1080/19761597.2015.1074513, https://doi.org/10.1080/19761597.2015.1074513

Tannenbaum, SI & Cerasoli, CP 2013, 'Do team and individual debriefs enhance performance? A meta-analysis', *Human factors*, vol. 55, no. 1, pp. 231-45.

Teede, HJ 2019, 'Advancing women in medical leadership', *Medical Journal of Australia*, vol. 211, no. 9, pp. 392-4. e1.

Thompson, JD, Zald, MN & Scott, WR 2017, *Organizations in action: Social science bases of administrative theory*, 1st edn, Routledge, New York USA.

Thomson III, NB, Rawson, JV, Slade, CP & Bledsoe, M 2016, 'Transformation and Transformational Leadership:: A Review of the Current and Relevant Literature for Academic Radiologists', *Academic Radiology*, vol. 23, no. 5, pp. 592-9.

Thomson, R & McLeod, J 2015, 'New frontiers in qualitative longitudinal research: An agenda for research', *International Journal of Social Research Methodology*, vol. 18, no. 3, pp. 243-50.

Trastek, VFMD, Hamilton, NWJD & Niles, EEBSJD 2014, 'Leadership Models in Health Care-A Case for Servant Leadership', *Mayo Clinic Proceedings*, vol. 89, no. 3, pp. 374-81, ProQuest Central; SciTech Premium Collection.

Trivedi, T 2021, 'The Uncertain Future of Banking: People and Change Management', *AIMS International Journal of Management. 15. 137. 10.26573/2021.15*, vol. 2.

Tsai, TC, Jha, AK, Gawande, AA, Huckman, RS, Bloom, N & Sadun, R 2015, 'Hospital board and management practices are strongly related to hospital performance on clinical quality metrics', *Health Affairs*, vol. 34, no. 8, pp. 1304-11.

Tuckett, A, Winters-Chang, P, Bogossian, F & Wood, M 2015, ''Why nurses are leaving the profession... lack of support from managers': What nurses from an e-cohort study said', *International journal of nursing practice*, vol. 21, no. 4, pp. 359-66.

Tudor, L 2014, 'Change management-Challenge and opportunity for sustainable development of Romanian companies', vol. 8, no. 1, pp. 466-76.

Uhl-Bien, M & Arena, M 2018, 'Leadership for organizational adaptability: A theoretical synthesis and integrative framework', *The Leadership Quarterly*, vol. 29, no. 1, pp. 89-104.

Uhl-Bien, M, Riggio, RE, Lowe, KB & Carsten, MK 2014, 'Followership theory: A review and research agenda', *The Leadership Quarterly*, vol. 25, no. 1, pp. 83-104, DOI <u>https://doi.org/10.1016/j.leaqua.2013.11.007</u>, <<u>https://www.sciencedirect.com/science/article/pii/S1048984313001227</u>>.

Ulrich, TH 2020, 'The Background in Leadership Theory: Authentic Leadership', in *Spiritual Leadership*, Springer, pp. 11-5.

Utecht, R & Heier, W 1976, 'The contingency model and successful military leadership', *Academy of Management Journal*, vol. 19, no. 4, pp. 606-18.

Uvhagen, H, Hasson, H, Hansson, J & von Knorring, M 2018, 'Leading top-down implementation processes: a qualitative study on the role of managers', *BMC Health Services Research*, vol. 18, no. 1, p. 562.

Vainieri, M, Ferre, F, Giacomelli, G & Nuti, S 2019, 'Explaining performance in health care: How and when top management competencies make the difference', *Health Care Management Review*, vol. 44, no. 4, p. 306.

Van Rossum, L, Aij, KH, Simons, FE, van der Eng, N & Ten Have, WD 2016, 'Lean healthcare from a change management perspective: the role of leadership and workforce flexibility in an operating theatre', *Journal of health organization and management*.

VanVactor, JD 2012, 'Collaborative leadership model in the management of health care', *Journal of Business Research*, vol. 65, no. 4, pp. 555-61.

Veronesi, G, Harley, K, Dugdale, P & Short, SD 2014, 'Governance, transparency and alignment in the Council of Australian Governments (COAG) 2011 national health reform agreement', *Australian Health Review*, vol. 38, no. 3, pp. 288-94. Vigersky, RA, Velado, K, Zhong, A, Agrawal, P & Cordero, TL 2021, 'The effectiveness of virtual training on the MiniMed[™] 670G system in people with type 1 diabetes during the COVID-19 pandemic', *Diabetes technology & therapeutics*, vol. 23, no. 2, pp. 104-9.

Vito, R 2020, 'Key variations in organizational culture and leadership influence: A comparison between three children's mental health and child welfare agencies', *Children and Youth Services Review*, vol. 108, p. 104600.

Vogel, B, Reichard, RJ, Batistič, S & Černe, M 2020, 'A bibliometric review of the leadership development field: How we got here, where we are, and where we are headed', *The Leadership Quarterly*, p. 101381.

Vrchota, J, Řehoř, P, Maříková, M & Pech, M 2021, 'Critical success factors of the project management in relation to industry 4.0 for sustainability of projects', *Sustainability*, vol. 13, no. 1, p. 281.

Vrontis, D, Thrassou, A, Santoro, G & Papa, A 2017, 'Ambidexterity, external knowledge and performance in knowledge-intensive firms', *The Journal of Technology Transfer*, vol. 42, no. 2, pp. 374-88.

Walshe, K & Chambers, N 2017, 'Clinical governance and the role of NHS boards: learning lessons from the case of lan Paterson', British Medical Journal Publishing Group.

Walumbwa, FO, Avolio, BJ, Gardner, WL, Wernsing, TS & Peterson, SJ 2008, 'Authentic leadership: Development and validation of a theory-based measuret', *Journal of management*, vol. 34, no. 1, pp. 89-126.

Wang, B, Qian, J, Ou, R, Huang, C, Xu, B & Xia, Y 2016, 'Transformational leadership and employees' feedback seeking: The mediating role of trust in leader', *Social Behavior and Personality: an international journal*, vol. 44, no. 7, pp. 1201-8.

Wang, D-S & Hsieh, C-C 2013, 'The effect of authentic leadership on employee trust and employee engagement', *Social Behavior and Personality: an international journal*, vol. 41, no. 4, pp. 613-24.

Wang, M-t & Degol, JL 2017, 'Gender Gap in Science, Technology, Engineering, and Mathematics (STEM): Current Knowledge, Implications for Practice, Policy, and Future Directions', *Educational Psychology Review*, vol. 29, no. 1, pp. 119-40, ProQuest Central; Social Science Premium Collection, DOI <u>http://dx.doi.org/10.1007/s10648-015-9355-x</u>.

Wayne, SJ & Green, SA 1993, 'The effects of leader-member exchange on employee citizenship and impression management behavior', *Human Relations*, vol. 46, no. 12, pp. 1431-40.

Weale, VP, Wells, YD & Oakman, J 2019, 'The work-life interface: a critical factor between work stressors and job satisfaction', *Personnel review*.

Wehner, MR, Nead, KT, Linos, K & Linos, E 2015, 'Plenty of moustaches but not enough women: cross sectional study of medical leaders', *BMJ*, vol. 351.

Weigel, KS, Kubik-Huch, RA & Gebhard, C 2020, 'Women in radiology: why is the pipeline still leaking and how can we plug it?', *Acta Radiol*, vol. 61, no. 6, pp. 743-8, NLM, DOI 10.1177/0284185119881723.

Weiss, YG & Weissman, C 2020, 'Relocating to a New OR Suite: Practical Observations', *AORN journal*, vol. 111, no. 5, pp. 515-26.

Wen, H 2014, 'The nature, characteristics and ten strategies of learning organization', *International Journal of Educational Management*.

West, M, Armit, K, Loewenthal, L, Eckert, R, West, T & Lee, A 2015, 'Leadership and leadership development in health care: the evidence base'.

Western, S 2019, *Leadership: A critical text*, 3rd edn, Sage, London UK.

Williams, B, Perillo, S & Brown, T 2015, 'What are the factors of organisational culture in health care settings that act as barriers to the implementation of evidence-based practice? A scoping review', *Nurse education today*, vol. 35, no. 2, pp. e34-e41.

Willis, E 2020, *Medical dominance*, 2nd edn, Routledge, Oxon UK and New York USA.

Wodak, R & Meyer, M 2015, *Methods of critical discourse studies*, 3rd edn, Sage, London UK.

Wong, CA & Spence-Laschinger, HK 2013, 'Authentic leadership, performance, and job satisfaction: the mediating role of empowerment', *Journal of Advanced Nursing*, vol. 69, no. 4, pp. 947-59, rzh, EBSCOhost, DOI 10.1111/j.1365-2648.2012.06089.x,

<<u>http://ezproxy.uws.edu.au/login?url=http://search.ebscohost.com/login.aspx?di</u> rect=true&db=rzh&AN=2012041494&site=ehost-live&scope=site>.

Workplace Gender Equality Agency 2019, *Gender Segregation in Australia's Workforce* viewed 17 September 2020, <<u>https://www.wgea.gov.au/data/fact-sheets/gender-segregation-in-australias-workforce</u>>.

Wu, F, Dixon-Woods, M, Aveling, E-L, Campbell, A, Willars, J, Tarrant, C et al. 2021, 'The role of the informal and formal organisation in voice about concerns in healthcare: A qualitative interview study', *Social Science & Medicine*, vol. 280, p. 114050.

Yahaya, R & Ebrahim, F 2016, 'Leadership styles and organizational commitment: literature review', *The Journal of Management Development*, vol. 35, no. 2, pp. 190-216, ProQuest Central; Social Science Premium Collection, DOI <u>http://dx.doi.org/10.1108/JMD-01-2015-0004</u>.

Yergler, JD 2012, 'Organizational culture and leadership', *Leadership & Organization Development Journal*.

Yukl, G 2010, 'Leadership in organizations, Upper Saddler River', *New Jersey: Person*.

Yukl, G 2013a, *Leadership in organisations*, Pearson Global Edition edn, University of Albany, State University of New York.

Yukl, G 2013b, *Leadership in Organizations Global Edition*, Pearson Education Limited.

Yukl, G 2019, *Leadership in Organizations*, Pearson Education, Incorporated.

Zehir, C, Ertosun, ÖG, Zehir, S & Müceldili, B 2011, 'The effects of leadership styles and organizational culture over firm performance: Multi-National companies in istanbul', *Procedia-Social and Behavioral Sciences*, vol. 24, pp. 1460-74.

Zhao, X, Hwang, B-G & Lee, HN 2016, 'Identifying critical leadership styles of project managers for green building projects', *International Journal of Construction Management*, vol. 16, no. 2, pp. 150-60, DOI 10.1080/15623599.2015.1130602,

<<u>https://doi.org/10.1080/15623599.2015.1130602</u>>.

Zuraik, A & Kelly, L 2018, 'The role of CEO transformational leadership and innovation climate in exploration and exploitation', *European Journal of Innovation Management*, vol. 22, no. 1, pp. 84-104.

List of Appendices

Appendix A – Functions of the NSW Entities	276
Appendix B - Ethics Approval	279
Appendix C - Ethics Approval – Site Specific Authorisation	282
Appendix D – Participant Consent Form	284
Appendix E - Multifactor Leadership Questionnaire	286
Appendix F - Authentic Leadership Questionnaire	291
Appendix G – MLQ Sample Results Interpretation	292
Appendix H – MLQ International Normative Samples – Oceania	304
Appendix I - Categorisation Of Results From The Completed MLQ	305
Appendix J - Interview Questionnaire Round 1	306
Appendix K - Interview Questionnaire Round 2	307
Appendix L - Interview Questionnaire Round 3	308
Appendix M – Interview Transcript Coding	310
Appendix N - Round 1 – Sample Interview Transcript	311

Appendix A – Functions of the NSW Entities

State-wide Health Services

Entity	Functions
NSW Ambulance	is responsible for providing responsive, high quality clinical care in emergency situations, including pre-hospital care, rescue, retrieval and patient transport services
Health Infrastructure	is responsible for the delivery of the NSW Government's major works hospital building program, under the auspices of a Board appointed by the Secretary. The Board Members each offer specialised expertise in areas of health and infrastructure delivery.
HealthShare NSW	provides corporate services and information technology services to public health organisations across NSW under the auspices of a Board appointed by the Secretary. The mission of HealthShare NSW is to provide common shared services across corporate, technology and disability services to NSW Health customers in an innovative, efficient, and cost-effective manner. HealthShare NSW extensive state-wide services include implementing the latest IT health information systems, providing payroll and accounts payable functions, supporting patient care through food and linen services, and assisting people with a disability to live and participate in the community.
NSW Health Pathology	is a state-wide clinical and scientific service that provides quality, reliable public pathology, forensic and analytical science services across NSW through four pathology networks. Our pathologists are medically trained clinicians who work in public hospitals and modern laboratories, providing critical testing for NSW Health, the NSW Food Authority and local government bodies. It also supports the state's judicial system by providing independent, objective analysis to law enforcement bodies including the police, local coroners and public prosecutors
eHealth NSW	is a distinct organisation within the NSW Ministry of Health providing state-wide leadership on the
	shape, delivery and management of Information Communication Technology (ICT) led healthcare.
-----------------------	--
	eHealth NSW is responsible for setting eHealth strategy, policy and standards, and works with Local Health Districts (LHDs) and Health Agencies to implement state-wide core systems and ensure compliance with state-wide standards.
Health Protection NSW	Reporting to the Chief Health Officer, Health Protection NSW is responsible for surveillance and public health response in NSW including monitoring the incidence of notifiable infectious diseases and taking appropriate action to control the spread of diseases. It also provides public health advice and response to environmental issues affecting human health.

Source: Developed for this research from information on the NSW Health Website as viewed on 04/08/2021

Pillar Organisations

Entity	Function
Agency for Clinical Innovation (ACI)	is a board-governed statutory health corporation responsible for reviewing clinical variation and supporting clinical networks in clinical guideline/pathway development with encouragement toward standardised clinical approaches based on best evidence. The Agency for Clinical Innovation works with clinicians, consumers and managers to design and promote better healthcare for NSW.
Bureau of Health Information (BHI)	is a board-governed organisation that provides independent reports to government, the community and healthcare professionals about the performance of the NSW public healthcare system, including safety and quality, effectiveness, efficiency, cost and responsiveness of the system to the health needs of the people of NSW.
Cancer Institute NSW	is Australia's first statewide government cancer agency. Established under the Cancer Institute (NSW) Act 2003 to lessen the impact of cancer across the State, its statutory objectives are to reduce the incidence of cancer in the community, increase survival from cancer, and improve the quality of life for people with cancer and their

	carers and provide a source of expertise on cancer control for the government, health service providers, medical researchers and the general community. The Cancer Institute NSW leads the development and delivery of the state-wide NSW Cancer Plan, which sets out a coordinated and collaborative approach to cancer control, involving people affected by cancer, government and non-government organisations, health professionals and researchers. The Cancer Institute drives initiatives to reduce unwarranted variations in outcomes cross diverse cultural and geographic communities, report on the performance of cancer services and enhance cancer research capabilities across NSW.
Clinical Excellence Commission (CEC)	is a board-governed statutory health corporation, responsible for leading safety and quality improvement in the NSW public health system. It was established in 2004 to reduce adverse events in public hospitals, support improvements in transparency and review of these events in the health system and promote improved clinical care, safety and quality in health services across NSW.
Health Education and Training Institute (HETI)	is a Chief Executive-governed statutory health corporation which coordinates education and training for NSW Health staff. The Institute works closely with local health districts, specialty health networks, other public health organisations and health education and training providers to ensure that world-class education and training resources are available to support the full range of roles across the public health system including patient care, administration and support services.

Source: Developed for this research from information on the NSW Health Website as viewed on 04/08/2021

Appendix B - Ethics Approval



Research and Ethics Office Leadership • Quality • Governance Locked Bag 7103, LIVERPOOL BC, NSW, 1871 Phone: 02 8738 8304 Facsimile: 02 8738 8310 http://www.sswahs.nsw.gov.au/swslhd/ethics/default.html

12 May 2016

A/Professor Terry Sloan Human Resources Department University of Western Sydney Locked Bag 1797 PENRITH NSW 2751 CC: Munyaradzi Gwede

THIS LETTER CONSTITUTES ETHICAL APPROVAL ONLY. THIS RESEARCH PROJECT MUST NOT COMMENCE AT A SITE UNTIL SEPARATE AUTHORISATION FROM THE CHIEF EXECUTIVE OR DELEGATE OF THAT SITE HAS BEEN OBTAINED. *

Dear A/Professor Sloan,

Project Title:	Leadership for change management; critical success factors in medical imaging departments in major
HREC Reference:	teaching hospitals in NSW. HREC/15/LPOOL/360
Local Project Number:	15/201

Thank you for meeting with the Human Research Ethics Committee on 21 March 2016. This Human Research Ethics Committee is constituted and operates in accordance with the National Health and Medical Research Council's National Statement on Ethical Conduct in Research Involving Humans and the CPMP/ICH Note for Guidance on Good Clinical Practice.

I am pleased to advise that the Committee has granted ethical approval of the above project.

The following documentation has been reviewed and approved:

Document	Version	Date
Cover Letter	N/A	21.07.2015
National Ethics Application Form	AU/1/B140217	23.07.2015
Protocol	1.1	23.06.2015
MASTER Participant Information Sheet	2.0	08.09.2015
MASTER Consent Form	1.0	13.07.2015
Bulundidi Gundaga – 3 Year survey	1.0	Undated
Questionnaire – 3.5 years of age	Not Provided	June 2009
Questionnaire – 4 years of age	2.0	October 2009
Questionnaire - 4.5 years of age	2.0	October 2009
Support Letter from Tharawal Aboriginal Corporation	N/A	09.06.2015
Ethics Further Information Letter 1	N/A	17.08.2015
Researchers response	N/A	06.09.2015
Protocol (Tracked and clean)	2.0	08.09.2015
Ethics Further Information letter 2		11.09.2015

Page 1 of 3

<u>Please ensure for all future documents submitted for review include a document</u> version number, document date and page numbering.

Monitoring Requirements: (National Statement Chapters 2.1 and 5.5)

The Committee has classified this project as:



- Monitoring required for this study will be:
 - Submission of Annual Progress Reports with the first report due 12 May 2017 and annually thereafter for the duration of the approval period. Please note that ongoing approval for this study will be granted pending review and approval of the first Annual Progress Report.
 - The first progress report is to include further information on the methodology and analysis which is to be reviewed the HREC before ongoing approval is granted.

Approval has been granted for the following site(s):

- Liverpool Hospital
- Bankstown Lidcombe Hospital
- Fairfield Hospital
- Campbelltown Hospital

Please note the following conditions of approval:

- 1. The Principal Investigator will immediately report anything which might warrant review of ethical approval of the project in the specified format, including:
 - any serious or unexpected adverse events; and
 - unforeseen events that might affect continued ethical acceptability of the project.
- 2. The Principal Investigator will report proposed changes to the research protocol, conduct of the research, or length of HREC approval to the HREC in the specified format, for review. For multi-centre studies, the Chief Investigator should submit to the Lead HREC and then send the amendment approval letter to the investigators at each sites so that they can notify their Research Governance Officer.
- The Principal Investigator will inform the HREC, giving reasons, if the project is discontinued before the expected date of completion.
- The Principal Investigator will provide an annual report to the HREC and at completion of the study in the specified format.
- 5. The Principal Investigator must reassure participants about confidentiality of the data.
- Proposed changes to the personnel involved in the study are submitted to the HREC accompanied by a CV where applicable.
- The Principal Investigator is responsible for ensuring the research project is conducted in line with relevant NSW Health, South Western Sydney Local Health District and Hospital policies available from: <u>http://www.sswahs.nsw.gov.au/swslhd/ethics/policies.html</u>

HREC approval is valid for one year, subject to review Annual Progress Reports. If the study is ongoing at the conclusion of the five year approval period, a full resubmission may be required. Ethics approval will continue during the re-approval process.

Page 2 of 3

The South Western Sydney Local Health District Human Research Ethics Committee has been accredited by the NSW Ministry of Health to provide single ethical and scientific review of research proposals conducted within the NSW public health system and Victorian and Queensland Public Health Organisations participating in the Mutual Acceptance Scheme.

You are reminded that this letter constitutes ethical approval only. This research project must not commence at a site until separate authorisation from the Chief Executive or delegate of that site has been obtained. It is your responsibility to forward a copy of this letter together with any approved documents as enumerated above, to all site investigators for submission to the site's Research Governance Officer.

Should you have any queries about your project please contact Annamarie D'Souza on the telephone number listed above. The HREC Terms of Reference, Standard Operating Procedures, membership and standard forms are available from the SWSLHD website: <u>http://www.sswahs.nsw.gov.au/swslhd/ethics/default.html</u>

Please quote the Local HREC reference 15/201 in all correspondence. The HREC wishes you every success in your research

Yours faithfully



Annamarie D'Souza on behalf of Professor Jeremy Wilson Chairperson, SWSLHD Human Research Ethics Committee

This HREC is constituted and operates in accordance with the National Health and Medical Research Council's (NHMRC) National Statement on Ethical Conduct in Human Research (2007). The processes used by this HREC to review multi-centre research proposals have been certified by the National Health and Medical Research Council.

Page 3 of 3

Appendix C - Ethics Approval – Site Specific Authorisation



Health South Western Sydney Local Health District

Research and Ethics Office Leadership · Quality · Governance Locked Bag 7103, LIVERPOOL BC, NSW, 1871 Phone: 02 8738 8304 Facsimile: 02 8738 8310 http://www.sswahs.nsw.gov.au/swslhd/ethics/default.html

12 May 2016

A/Professor Terry Sloan Human Resources Department University of Western Sydney Locked Bag 1797 PENRITH NSW 2751 CC: Munyaradzi Gwede

Dear A/Professor Sloan,

Project Title:	Leadership for change management; critical success factors in medical imaging departments in major teaching hospitals in NSW.
HREC Reference:	HREC/15/LPOOL/360
SSA Reference:	SSA/15/LPOOL/362 – Bankstown
	SSA/15/LPOOL/363 – Campbelltown
	SSA/15/LPOOL/364 – Fairfield
	SSA/15/LPOOL/363 – Liverpool
Local Project Number:	15/201

SITE SPECIFIC AUTHORISATION

Thank you for your correspondence received 12 May 2016 in response to our request for further information dated 8 April 2016.

I am pleased to inform you that the Chief Executive has granted authorisation for this study to take place at the following site(s):

Liverpool Hospital

- Campbelltown Hospital
- Bankstown-Lidcombe Hospital
- Fairfield Hospital

The following are authorised for use at the Liverpool Hospital, Campbelltown Hospital, Bankstown-Lidcombe Hospital and Fairfield Hospital sites:

	M	aster	Site S	Specific
Document	Version	Date	Version	Date
MASTER Participant Information Sheet	2.0	08.09.2015	1.0	15/04/2016
MASTER Consent Form	1.0	13.07.2015	1.0	12/05/2016

The following conditions apply to this research project. These are additional to those conditions imposed by the Human Research Ethics Committee that granted ethical approval:

- Proposed amendments to the research protocol or conduct of the research which may affect the ethical acceptability of the project, and which are submitted to the lead HREC for review, are copied to this office.
- Proposed amendments to the research protocol or conduct of the research which may affect the ongoing site acceptability of the project, are to be submitted to this office.

Page 1 of 2

- Please note that you are responsible for making the necessary arrangements (e.g. identity
 pass and vaccine compliance as per NSW Health Policy Directive PD2011_005) for any
 researcher who is not employed by the South Western Sydney Local Health District and is
 conducting the research on-site.
- 4. The Principal Investigator is responsible for ensuring the research project is conducted in line with relevant NSW Health, South Western Sydney Local Health District and Hospital policies available from: <u>http://www.sswahs.nsw.gov.au/swslhd/ethics/policies.html</u>
- 5. Proposed changes to the personnel involved in the study at South Western Sydney Local Health District sites are submitted to the South Western Sydney Local Health District Research and Ethics Office accompanied by the required supporting documents. A list of the documentation required to add an Investigator to a study is located on the South Western Sydney Local Health District Research and Ethics Office website: <u>http://www.swslhd.nsw.qov.au/ethics/forms.html</u>

Yours sincerely,

Annamarie D'Souza Manager, Research and Ethics Office South Western Sydney Local Health District (SWSLHD)

Appendix D – Participant Consent Form

CONSENT FORM

[To be used in conjunction with a Participant Information Sheet]

Leadership for change management; critical success factors in medical imaging departments in major teaching hospitals in NSW.

I,..... of.....

agree to participate in the study described in the participant information statement attached to this form.

I acknowledge that I have read the participant information statement, which explains why I have been selected, the aims of the study and the nature and the possible risks of the investigation, and the statement has been explained to me to my satisfaction.

Before signing this consent form, I have been given the opportunity of asking any questions relating to any possible physical and mental harm I might suffer as a result of my participation and I have received satisfactory answers.

I understand that I can withdraw from the study at any time without prejudice to my relationship with the South Western Sydney Hospital Local Health District.

I agree that research data gathered from the results of the study may be published, provided that I cannot be identified.

I understand that if I have any questions relating to my participation in this research, I may contact Munyaradzi Gwede on telephone 040 657 8752, who will be happy to answer them.

I acknowledge receipt of a copy of this Consent Form and the Participant Information Statement.

Signature of participant Please PRINT name Date

Signature of witness Please PRINT name Date

Signature of investigator (if applicable) Please PRINT name Date

As a Leader	Frequently, if not always Fairly often Sometimes Once in a while Not at all Unsure
 I provide others with assistance in exchange for their efforts.* 	
 I re-examine critical assumptions to question whether they are appropriate.* 	
 I fail to interfere until problems become serious.* 	
 I focus attention on irregularities, mistakes, exceptions, and deviations from standards.* 	
 I avoid getting involved when important issues arise.* 	
 6. 6l talk about my most important values and beliefs.* 	
7. 7I am absent when needed.*	
 8I seek differing perspectives when solving problems.* 	
 9. 9I talk optimistically about the future.* 	
10.10I instill pride in others for being associated with me.	

Appendix E - Multifactor Leadership Questionnaire

As a Leader	Frequently, if not always Fairly often Sometimes Once in a while Not at all Unsure
	\circ \circ \circ \circ \circ \circ
11.1 discuss in specific terms who is responsible for achieving performance targets.*	
12.1 wait for things to go wrong before taking action.*	
13.1 talk enthusiastically about what needs to be accomplished.*	
14.1 specify the importance of having a strong sense of purpose.*	
15.I spend time teaching and coaching.*	
16.1 make clear what one can expect to receive when performance goals are achieved.*	
17.I show that I am a firm believer in "If it ain't broke, don't fix it."*	
18.I go beyond self-interest for the good of the group.*	
19.I treat others as individuals rather than just as members of the group.*	
20.1 demonstrate that problems must become chronic before I take action.	

Items marked by * are required.	
As a Leader	Frequently, if not always Fairly often Sometimes Once in a while Not at all Unsure
21.I act in ways that build others' respect for me.*	
22.I concentrate my full attention on dealing with mistakes, complaints, and failures.*	
23.1 consider the moral and ethical consequences of decisions.*	
24.1 keep track of all mistakes.*	
25.1 display a sense of power and confidence.*	
26.1 articulate a compelling vision of the future.*	
27.I direct my attention toward failures to meet standards.*	
28.1 avoid making decisions.*	
29.I consider each individual as having different needs, abilities, and aspirations from others.*	
30.I get others to look at problems from many different angles.*	

Items marked by * are required.	
As a Leader	Frequently, if not always Fairly often Sometimes Once in a while Not at all Unsure
31.I help others to develop their strengths.*	
32.I suggest new ways of looking at how to complete assignments.*	
33.I delay responding to urgent questions.*	
34.1 emphasize the importance of having a collective sense of mission.*	
35.1 express satisfaction when others meet expectations.*	
36.1 express confidence that goals will be achieved.*	
37.1 am effective in meeting others' job-related needs.*	
38.1 use methods of leadership that are satisfying.*	
39.I get others to do more than they expected to do.*	
40.1 am effective in representing my group to higher authority.	

Items marked by * are required.	
As a Leader	Frequently, if not always Fairly often Sometimes Once in a while Not at all Unsure
41.1 work with others in a	
42.I heighten others' desire to succeed.*	
43.1 am effective in meeting organizational requirements.*	
44.1 increase others' willingness to try harder.*	
45.I lead a group that is effective.*	
Items marked by * are required.	

Appendix F - Authentic Leadership Questionnaire

As a Leader I	
	Frequently if not always
	Fairly often
	Sometimes
	Once in a while
	Not at all
1. sav exactly what I mean*	
2. admit mistakes when they are	
made*	
3. encourage everyone to speak their	
mind*	
4. tell others the hard truth*	
5. display emotions exactly in line	
with feelings*	
6. demonstrate beliefs that are	
consistent with actions*	
7. make decisions based on my core	
values*	
8. ask others to take positions that	
support their core values*	
9. make difficult decisions based on	
high standards of ethical conduct*	
10. solicit views that challenge my	
deeply held positions*	
11. analyze relevant data before	
coming to a decision*	
12. listen carefully to different points	
of view before coming to	
conclusions*	
13. seek feedback to improve	
interactions with others*	
14. accurately describe how others	
view my capabilities*	
15. know when it is time to re-	
evaluate my position on important	
Issues*	
16. show I understand how specific	
actions impact others*	
Items marked by * are required.	

Appendix G – MLQ Sample Results Interpretation

MLQ	Multifactor Leadership Questionnaire
SELF-EVALUATION	DEVELOPED BY BRUCE AVOLIO & BERNARD BASS
	REPORT PREPARED FOR: JANE SAMPLE Gender: FEMALE Age: 22
GUESTIONNAIRE USED UNDER LICENSE BY: Specialist: DRAGOS ILIESCU Administered an: 25/05/2007 Report generated on date/hour: 30.05.2007 / 11:14:38 Identifier/serial number: 00067175	

INTRODUCTION TO UNDERSTANDING THIS REPORT

This report will help in the better understanding of the preferences, attitudes, and behaviors characteristic for the evaluated person in key dimensions of his/her leadership style.

This report is intended for the use of psychologists, consultants or other specialised professionals in their work with clients.

These results should only be considered in conjunction with professional judgment, after a careful and detailed analysis, and only after corroborating these data with the results of an interview and of possible other psychometric instruments. Results contained in this report may be subject to alterations and special highlights as a function of such corroborations made by a specialised professional.

This report is based on the MLQ (Multifactor Leadership Questionnaire, Form 5X), a psychometric instrument that has been validated in a wide variety of research programs, including extensive studies in Romania.

The Multifactor Leadership Questionnaire (MLQ) is a structured, verbal, omnibus measure of leadership styles, . The questionnaire consists of 45 items, covering what is known as the "full-range" leadership model.

GETTING THE MOST OF THIS REPORT

In order to get the most of this report, one should first understand the principles of the Full Range Leadership model and the distinctions between transformational leadership, transactional management, and passive/avoidant leader behavior, as well as the outcomes of leadership. All these are explained in the section labeled "Full Range Leadership Model".

The section labeled "Snapshot of the Leadership Profile" visualizes an overview of the concepts of the Full Range Leadership model. These concepts include Transformational Leadership, Transactional Leadership and Passive / Avoidant Leadership. The Snapshot also visualizes the passive vs. active forms of leadership, as well as the more effective vs. rather ineffective forms. The Snapshot is a good place to start in interpreting the results of this report.

Then, the scores for all leadership scales and their subscales should be considered, as reported to the selected norm. Raw scores are in this matter of little importance, so norm-referenced scores should be considered. All these are pictured in the section labeled 'Drilldown of the Leadership Profile'.

Finally, the "Norm-Referenced MLQ Profile" is intended for the use of researchers or specialists who want to build on an indepth norm-referenced image of the profile. The profile is represented in standardised T scores.

Also, this report has at the end a section labeled "Modus Operandi", where individual responses to the items of the questionnaire may be viewed and compared, as well as statistics about general response style, missing items etc.

MLQ Self-Evaluation / Form SX

THE FULL RANGE LEADERSHIP MODEL

Over the past 20 years, there has been considerable interest in the new paradigm of transformational and transactional leadership (Avolio, 1999; Bass, 1998).

Previous leadership models have fallen short in explaining a 'full range' of leadership styles, ranging from the charismatic and inspirational leaders to avoidant laissez-faire leaders.

The full range model of leadership was developed to broaden the range of leadership styles typically investigated in the field. The model was labeled 'full range' to challenge the leadership field to broaden its thinking about what constitutes a much broader range of leadership styles than the paradigms of initiation of structure and consideration.

The full range model of leadership assumes the existence of differences in the effectiveness of leadership styles, based on the active/passive distinction. Broad categories of leadership range thus from Passive / Avoidant Leadership (Laissez-Faire), through the classical model of Transactional Leadership and up to Transformational Leadership.

THE MLQ AS PART OF THE FULL-RANGE LEADERSHIP MODEL

The Multifactor Leadership Questionnaire is closely linked to the concepts of Transformational Leadership and of Full-Range Leadership.

The MLQ and MLQ Report have evolved over the last 25 years based on numerous investigations of leaders in public and private organizations, from CEOs of major corporations to non-supervisory project leaders. The major leadership constructs – transformational leadership, transactional leadership, and passive/avoidant leadership – form a new paradigm for understanding both the lower and higher order effects of leadership style. This paradigm builds on earlier leadership paradigms-such as those of autooratic versus democratic leadership, directive versus participative leadership, and task-versus relationship oriented leadership-which have dominated selection, training, development, and research in this field for the past half century.

The MLQ and MLQ Report were developed to expand the dimensions of leadership measured by previous leadership surveys and to provide a concise computerized feedback form that can be used for individual, team, and organizational development as well as individual counseling. Other leadership measures had generally ignored key factors, such as Inspirational Motivation, typically ascribed to successful leaders. Prior leadership research and training had concentrated on identifying and measuring behaviors that fell into a limited range that we have labeled Transactional leadership.

PASSIVE / AVOIDANT LEADERSHIP

Passive leaders avoid to identify and clarify potential problem areas, avoid to get involved, to set standards and to monitor for results. This leadership style has most of the time a negative effect on leadership results.

MLQ Self-Evaluation / Form 5X

TRANSACTIONAL LEADERSHIP

In its more constructive form, transactional leadership is supplemented by working with individuals and/or groups, setting up and defining agreements or contracts to achieve specific work objectives, discovering individuals capabilities, and specifying the compensation and rewards that can be expected upon successful completion of the tasks.

In its corrective form, it focuses on actively setting standards. In its passive form, it involves waiting for mistakes to occur before taking action. In its active form, there is closely monitoring for the occurrence of mistakes. In either its passive or active form, it focuses on identifying mistakes. Many consultants using the MLQ have found it useful to label Contingent Reward (CR) and Management-by-Exception: Active (MBEA) as Transactional Leadership and Management-by-Exception: Passive (MBEP) and Laissez Faire as Passive/Avoidant Leadership.

TRANSFORMATIONAL LEADERSHIP

Interestingly, when all levels of managers, students, and project leaders around the world were asked to describe the characteristics and behaviors of the most effective leaders they had worked with in their past, their descriptions encompassed much more than the reward for effort exchange behavior and corrective orientation that typifies transactional leadership. Specifically, they described leaders who had the greatest influence on them as transformational: inspirational, intellectually stimulating, challenging, visionary, development oriented, and determined to maximize performance. In many cases, the term "charisma" was used.

OUTCOMES OF LEADERSHIP

Transformational and Transactional leadership are both related to the success of the group. Success is measured with the MLQ by how often the raters perceive their leader to be motivating, how effective raters perceive their leader to be at different levels of the organisation, and how satisfied raters are with their leader's methods of working with others.

SOME RESEARCH FINDINGS

Transformational leaders created greater alignment around strategic visions and missions.

Transformational leadership behaviour factors are associated with organisational sales increases, market share, earnings and ROI.

Scores on transformational leadership predict individual and group performance.

Transformational leadership has been found to explain between 45% and 60% of organisational performance.

Transformational leaders created greater unit cohesion, commitment, and lower turnover.

Transformational leadership predicted higher levels of product innovation in R & D teams.

Transformational leaders created safer work environments.

Transformational leadership training has been shown to improve leadership and associated performance over time.

The nine factor Full Range Leadership Model has been shown to best represent the data collected by the MLQ when sample characteristics and situation factors are statistically controlled.

MLQ Self-Evaluation / Form 5X

SNAPSHOT OF THE LEADERSHIP PROFILE

(Profiled against a full range of leadership styles)

The Snapshot of the leadership profile visualizes an overview of the concepts of the Full Range Leadership model. These concepts include Transformational Leadership, Transactional Leadership and Passive / Avoidant Leadership. The Snapshot also visualises the passive vs. active forms of leadership, as well as the more effective vs. rather ineffective forms.

Laissez-Faire Leadership (LF) is typical for managers who avoid involvement. It is passive, avoidant and ineffective. Transactional Leadership, in its passive form involves waiting for mistakes to occur before taking action - It is called Management-by-Exception Passive (MBE-P) and is regarded as part of the Passive/Avoidant leadership style. In its active form, of either Management-by-Exception Active (MBE-A) or Contingent Reward (CR), it involves a close monitoring of behaviors and actively setting standards. Transformational leadership encompasses much more than the reward for effort exchange behavior and corrective orientation mentioned above. Transformational leadership is inspirational, intellectually stimulating, challenging, visionary, development oriented. It is regarded as the most active and effective form of leadership.



0 - Never / 1 - Once in a while / 2 - Sometimes / 3 - Fairly often / 4 - Frequently, if not always

MLQ Self-Evaluation / Form SX

DRILLDOWN OF THE LEADERSHIP PROFILE



MLQ Self-Evaluation / Form SX

IM: INSPIRATIONAL MOTIVATION RAW SCORE = 1.50

INSPIRES OTHERS

The INSPIRATIONAL MOTIVATION scale identifies leaders who INSPIRE OTHERS. Often, inspiration can occur without the need for identification of associates with the leader. Inspirational leaders articulate, in simple ways, shared goals and mutual understanding of what is right and important. They provide visions of what is possible and how to attain them. They enhance meaning and promote positive expectations about what needs to be done. The question one must ask is, "Whom are they inspiring - themselves or the greater good of their group, unit, organization, and/or community?"



IS: INTELLECTUAL STIMULATION RAW SCORE = 3.00

ENCOURAGES INNOVATIVE THINKING

The INTELLECTUAL STIMULATION scale identifies leaders who are able to ENCOURAGE INNOVATIVE THINKING. In addition to idealized Influence and inspirational Motivation, transformational leadership also involves the intellectual stimulation of associates' ideas and values. Through Intellectual Stimulation, transformational leadership others to think about old problems in new ways. They are enouraged to question their own beliefs, assumptions, and values, and, when appropriate, those of the leader, which may be outfaited or inappropriate for solving ourrent problems. As a consequence, associates develop the capacity to solve future problems unforeseen by the leader. Associates learn to tackle and solve problems on their own by being oreative and innovative. A key measure of a leader's effectiveness is how capable their associates are when operating without the leader's presence or direct involvement. An intellectually stimulating leader arouses in others a greater cognizance of problems, avareness of their own thoughts and imagination, and recognition of their beliefs and values.



IC: INDIVIDUAL CONSIDERATION RAW SCORE = 1.25

COACHES PEOPLE

The INDIVIDUAL CONSIDERATION scale identifies leaders who are able to COACH PEOPLE. Individualized Consideration is another aspect of transformational leadership. It means understanding and sharing in others' oonoems and developmental needs and treating each individual uniquely. In addition, Individualized Consideration represents an attempt on the part of leaders to not only recognize and statisfy their associates' ourrent needs, but also to expand and elevate those needs in an attempt to maximize and develop their full potential. This is one reason why transformational leaders set examples and assign tasks on an individual basis. Transformational leaders also provide opportunities and develop organizational outures supportive of individual growth.



MLQ Self-Evaluation / Form 5X

TRANSACTIONAL LEADERSHIP



Transactional leaders work toward recognizing the roles and tasks required for associates to reach desired outcomes; they also clarify these requirements for associates, thus creating the confidence they need to exert the necessary effort. Transactional leaders also recognize what associates need and desire, clarifying how those needs and desires will be satisfied if the associate expends the effort required by the task. Such motivation to perform will provide a sense of direction and help to energize others. This approach, currently stressed in most popular leadership training programs, is helpful but limited to first-order exchanges.

Note that the transactional process, in which the leader clarifies what the associates need to do for a reward, is nevertheless viewed here as an essential component of the full range of effective leadership. The newer paradigm adds transformational leadership to previous transactional leadership models. It is likely to have direct relevance particularly to the volunteer sector where the rewards are more personal and social and are based on commitment to ideals. In introducing the concept of transformational leadership, Burns (1978) pointed out that the moral movers and shakers of the world don't cater to their self-interest as much as they enable others to transcend their own self interest for the good of their group, organization, community, or society.

Transformational leadership does not replace transactional leadership, it augments transactional leadership in achieving the goals of the leader, associate, group, and organization. Although transformational leaders can be transactional when appropriate, transactional leadership is often a prescription for lower levels of performance or non-significant change.

me = 1.50	s				rew=3.50
rum = 3.08	and the second			Research veliciated be	norm = 3.08
	Not at all	Once in a while	Sametimes	Faitty offen	Frequently, if not always
MBE-A. MANA	CEMENT-BY-EXCEPT	ON: ACTIVE	MONITOR	OS MISTAKES	
MEE-A MANA NAW SCORE	CEMENT-BY-EXCEPT) = 3.00	ON ACTIVE	MONITOF	S MISTAKES	ñna landarr ura to
MBE-A MANA WW SCORE The MANAGE peaily with anotions for in quick and nistakes, oo	CEMENT-BY-EXCEPT = 3.00 EMENT-BY-EXCEPT blarity compliance s r not respecting the thorough corrective noentrate on errors ;	ON: ACTIVE CN: ACTIVE scale identifies les tandards, as well as vividly des set standards. This leadership e measures, if appropriate. Lea and exceptions from the rule ar	MONITOF aders who focus on MONITI onbe inefficient performan style focuses on a careful in ders practioning an active ins ind on the treatment of these	RS MISTAKES WING MISTAKES. High soo be: Following these descript controling of deviations, mis anagement by esseption use oversteps.	ring leaders use to ions, they focus on takes and errors and to keep track of
MEE-A MANA WW SCORE The MANAGE peakly with to anotions ho anotions	CEMENT-BY-EXCEPTD = 300 EMENT-BY-EXCEPTD clarity compliance so thorough corrective noembrate on errors a \$	ON: ACTIVE ON: ACTIVE scale identifies les tandards, as well as vividly des set standards. This leadership e measures, if appropriate. Lea and exceptions from the rule ar	MONITOF aders who focus on MONIT anibe inefficient performan style focuses on a caterul in ders practicing an active ins ind on the treatment of these	RS MISTAKES RING MISTAKES, High soo be Following these descript ontoring of deviations, mis anagement by esseption use oversteps.	ring leaders use to ions, they focus on takes and errors and to keep track of
MBE-A MANA WW SCORE the MANAGE peaity with anotions fro anotions for anotions for a	CEMENT-BY-EXCEPT = 3.00 EMENT-BY-EXCEPT olarity compliance s r not respecting the r thorough corrective noentrate on errors : \$	ON: ACTIVE CN: ACTIVE soale identifies les tandards, as well as vividly des set standards. This leadership e measures, if appropriate. Lea and exceptions from the rule ar	MONITOF aders who focus on MONITI onbe inefficient performan style focuses on a careful in ders practioning an active ins ind on the treatment of these	RS MISTAKES Wilkig Mistakes, High soo to Following these description intoring of deviations, mis magement by esseption use oversiteps.	fing leaders use to ions, they focus on takes and errors and to keep track of rew = 3.00 rem = 2.43

MLQ Self-Evaluation / Form SX





The MLQ discusses to special types of passive / avoidant leadership: Management by Exception: Passive (MBE-P) and Laissez-Faire Leadership (LF). Behaviors typical for Management by Exception: Passive are those behaviors where the leader takes corrective measure, but only after the fire has started. His behavior is thus not proactive, but reactive and focused on punishment. Behaviors typical for Laissez-Faire leadership are those behaviors where the leader avoids involvement completely, and does not even react to threats and problems arising. Most of the time this behavior states that a problem will disapear or solve itself in time.

Both passive / avoidant leadership styles (MBE-P and LF) have a negative impact on the performance of individuals, groups and organizations, as well as a strong emotional impact on followers and colleagues of leaders adopting these leadership styles.



LF: L	AISSE	Z-F	AIRE
RAW	scol	E.	2.00

AVOIDS INVOLVEMENT

The LAISSEZ-FAIRE soale identifies leaders who tend to AVOID INVOLVEMENT. This leadership style could be easily defined as "non-leadership" and is the exact opposite of an efficient transformational leadership style. Permissive leaders refuse to assume the responsibilities that are part of their position as leaders: they do not offer enough information to their followers, do not offer feedback, do not aoknowledge or work towards their followers' satisfaction. High scorers in this soale avoid approaching important problems, are absent when needed, avoid making decisions and have late reactions to urgent problems.

new = 2.00	*	1	2	*	4 mw=2.00
nam = 0.88	Research validated benchmark	Orrain antile	Samelines	Fallendar	nam = 0.88

MLQ Self-Evaluation / Form SX

Multifactor Leadership Questionnaire

Avolio & Bass, 1995, 2004

OUTCOMES OF LEADERSHIP



Both transformational and transactional leadership are closely related to individual, group and organizational success. Leadership efficiency is evaluated by the MLQ based on perceptions of followers, who should see their leaders as being good motivators, as having efficient interaction skills with different organizational levels and as generating satisfaction with work methods.

FF- EXTRA FF	FORT				
NAW SCORE	= 2.00		GENERA	IES EXTRA EFF	ORT
te EXTRA E feots of an forts, posit splity the v	EFORT scale identifies k efficient leadership styl ively exceeding legitima rish of their followers to 0	eaders who are able to BE (e, is defined as the wish of the behavioral expectations succeed and to overstep o	ENERATE EXTRA EFFORT followers to strive for supe of their leaders, their group bjeotives and induse position	in their tollowers. Extra effo nor performance by deployin or their organization. High s we supplementary behaviors.	rt, as one of the direct ng supplementary oorers in this soale
rane = 2.00				1.	rew = 2.00
nom = 3.03	No. of the second	20.000 00	- 14 I	Research yelduled ber	chmark = 3.03
	Not at all	Once in a while	Samelines	Fairly offen	Frequently, if not always
F: EFFECT	VENESS = 2.25		IS FEFICI	ENT	
AW SCORE to EFFECT so efficient enerally ger	IVENESS scale identifie dy represent the group i nerate a higher efficienc	s leaders who are able to B n front of the higher organs y in all the structures they :	E EFFICIENT. Efficient lead cabonal authority, are efficient are involved with.	ers satisty the protessional o ent in meeting organizational	f their followers. They objectives and
AW SCORE te EFFECT so efficient nerally ger raw = 2.25 nom = 3.08	VENESS soale identifie dy represent the group i nerate a higher efficience	s leaders who are able to B n front of the higher organs y in all the structures they :	EFFICIENT. Efficient lead attional authority, are efficient are involved with.	ens satisfy the professional o ent in meeting organizational	their followers. They objectives and rew=225 row=308
www.scole. he EFFECT iso efficient enerally ger now = 2.25 nom = 3.08	VENESS scale identifie dy represent the group i nerate a higher efficience 0	s leaders who are able to B n front of the higher organs y in all the structures they 	EFFICIENT. Efficient lead attional authority, are efficient are involved with.	ens satisfy the professional o ent in meeting organizational	raw = 225 ctruat Frequently, if not always
www.scole the EFFECT iso efficient enerally ger raw = 2.25 nom = 3.08 IAT: SATISFA WW SCORE	IVENESS scale identifie dy represent the group i nerate a higher efficience 0	s leaders who are able to B n front of the higher organic y in all the structures they : 	EFFICIENT. Efficient lead actional authority, are effor are involved with.	ers satisfy the professional o ent in meeting organizational Meeting visit and be Fatty often	f their followers. They objectives and rew = 225 com = 3.08 Frequently, if not always
AW SCORE the EFFECT so efficient enerally get rom = 2.25 nom = 3.08 AT: SATISFA AW SCORE the SATISFA the leaderst abstaction in dispoial sk	VENESS soale identifie dy represent the group i nerate a higher efficience a	s leaders who are able to B n front of the higher organs y in all the structures they : 	GENERA source of the source o	TES SATISFACTION in their following process, with	I their followers. They objectives and rew = 225 rom = 3.08 Frequently, if not always OW overs. Batistation nterpersonal good interpersonal
AW SCORE the EFFECT so efficient enerally get nom = 2.25 nom = 3.08 AT: SATISFA AW SCORE the SATISFA the Satisfaction of the Satisfaction of the Satisfaction of the Satisfaction of the Satisfaction of the Satisfaction of the Satisfaction of the Satisfaction of the Satisfaction of the Satisfaction of the Satisfaction of the Satis	VENESS soale identifie dy represent the group i nerate a higher efficience 0	s leaders who are able to B n front of the higher organs y in all the structures they : 	GENERAT who are able to GENERAT identifies with its higher si who are able to GENERAT identifies with its higher si in their followers.	ers satisfy the professional o ent in meeting organizational Promotive vehicuted for Faity often TES SATISFACTION E SATISFACTION in their follo cores leaders who generate in therito, honest persons, with	I their followers. They objectives and rew = 2:25 norm = 3:08 Frequently, if not always OW overs. Satisfaction nterpersonal good interpersonal
AT: SATISFA AT: SATISFA WW SCORE he SATISFA he SATISFA	VENESS scale identifie dy represent the group i nerate a higher efficience 0	s leaders who are able to B n front of the higher organs y in all the structures they : 	GENERA Sametimes GENERA who are able to GENERAT identifies with its higher so warm, nurburing, open, auf n their followers. 2	TES SATISFACTION in their followers with the most persons, with	Their followers. They objectives and raw = 2.25 rom = 3.08 Frequently, if not always OON owers. Batisfaction nterpersonal good interpersonal

MLQ Self-Evaluation / Form SX

NORM-REFERENCED MLQ PROFILE (T SCORES)

Scoring based on: THE ROMANIAN NORMS FOR LEADERS, SELF-ASSESSMENT (N=785ss)



MLQ Self-Evaluation / Form SX

MODUS OPERANDI

ANSWERS TO THE ITEMS

(1):14	(11):54	(21): '3	(31): "2	(41): '4
(2): '1'	(12): '1'	(22): "2	(32): 14	(42): '3'
(3): 2	(13): '3'	(23): '1'	(33): "1"	[43]: 2
(4): "4	(14): 2	(24):14	(34): '3'	[44]: '1'
(5): '1'	(15): '1'	(25): '3'	(35): "2	(45): '2
(6): '3	(16): '4'	(26): "1"	(36): "1"	
171:2	(17): '3'	(27): 2	(37): "3	
(8): "4"	(18): "2"	(28): 14	(38): "4"	
(9): '1'	(19): '1'	(29): '1'	(39): 2	
(10): 2	(20): '4'	(30): "3"	(40): "1"	

GENERAL ITEM STATISTICS

# '0' answers: 0 out of 45 (0.00%)	1
# '1' answers: 13 out of 45 (28.89%)	
# "2" answers: 12 out of 45 (26.67%)	_
# '3' answers: 9 out of 45 (20.00%)	
# '4' answers: 11 out of 45 (24.44%)	1
# missing answers: 0 out of 45 (0.00%)	

MISSING ITEMS												
Scale	ы	18	м	IS	IC	CR	MBE-A	MBE-P	LF	EE	EFF	SAT
Total number of items	4	4	4	4	4	4	4	4	4	з	4	2
Missing answers	0	0	0	0	0	0	0	0	0	0	0	0

303 | Page

MLQ Self-Evaluation / Form SX

11

Appendix H – MLQ International Normative Samples – Oceania

	Т	otal Sampi N=13488)	n	(8	eif, (N=11	12)	(Highe	r Level,N-	1860)	(Same	Level,N=2	341)	(Lowe	Level, N	4376)	(Other	Level, N	-2061)
Soale	Mean	SD	Range	Mean	SD	Range	Mean	SD	Range	Mean	SD	Range	Mean	SD	Range	Mean	SD	Range
Idealized Influence: Attributed	2.94	0.76	4.00	2.86	0.54	3.25	3.05	0.78	4.00	2.88	0.76	(4.00)	2.94	0.77	4.00	3.00	0.77	4.00
Behaviors	2.89	0.75	4.00	3.05	0.61	3.75	2.99	0.79	4.00	2.84	0.72	4.00	2.86	0.77	4.00	2.83	0.73	4.00
Motivation	2.99	0.77	4.00	3.07	0.59	3.50	3.02	0.83	4.00	2.87	0.77	4.00	3.05	0.77	4.00	3.00	0.75	4.00
Intellectual Stimulation	2.88	0.72	4.00	3.08	0.51	2.50	2.96	0.73	4.00	2.81	0.73	4.00	2.88	0.72	4.00	2.86	0.72	4.00
Consideration	2.89	0.78	4.00	3.20	0.49	2.75	2.98	0.74	4.00	2.80	0.75	4.00	2.85	0.83	4.00	2.87	0.75	4.00
Contingent Reward Management by	2.90	0.74	4.00	2.90	0.54	3.50	3.03	0.72	4.00	2.84	0.71	4.00	2.88	0.80	4.00	2.94	0.71	4.00
Exception: Active Management by	1.79	0.92	4.00	1.69	0.83	4.00	1.87	0.97	4.00	1.77	0.89	4.00	1.78	0.92	4.00	1.85	0.88	4.00
Exception: Passive	1.09	0.78	4.00	1.08	0.63	4.00	1.11	0.85	4.00	1.12	0.76	4.00	1.07	0.78	4.00	1.07	0.74	4.00
Laissez Faire	0.70	0.73	4.00	0.70	0.57	4.00	0.69	0.81	4.00	0.70	0.72	4.00	0.70	0.70	4.00	0.67	0.70	4.00
Extra Effectivenessort	2.63	0.90	4.00	2.68	0.61	3.67	2.76	0.86	4.00	2.47	0.92	4.00	2.69	0.91	4.00	2.63	0.87	4.00
Effectiveness	3.11	0.73	4.00	3.12	0.50	3.00	3.20	0.74	4.00	3.01	0.73	4.00	3.16	0.73	4.00	3.12	0.73	4.00
Satisfaction	3.14	0.83	4.00	3.12	0.53	3.00	3.20	0.84	4.00	3.07	0.81	4.00	3.14	0.87	4.00	3.18	0.82	4.00

Table 10c Oceana (Australia, New Zealand and Pacific Islands) Descriptive Statistics for MLQ 5X 2004 Normative Sample

Appendix I - Categorisation Of Results From The Completed MLQ

Characteristic	Scale Name	Scale Abbreviation	Items (questions measuring
			attribute)
Transformational	Idealised Attributes (IA) or Idealised	IA or II(A)	10,18,21,25
	Influence (Attributes) (II(A)		
Transformational	Idealised Behaviours (IB) or Idealised	IB or II(B)	6, 14, 23, 24
	Influence (Behaviour) (II(B)		
Transformational	Inspirational Motivation	IM	9, 13, 26, 36
Transformational	Intellectual Stimulation	IS	2, 8, 30, 32
Transformational	Individual Consideration	IC	15, 19, 29, 31
Transactional	Contigent Reward	CR	1, 11, 16, 35
Transactional	Management by Exception (Active)	MBEA	4, 22, 24, 27
Passive Avoindant	Management by Exception (Passive)	MBEP	3, 12, 17, 20
Passive Avoindant	Laissez -Faire	LF	5, 7, 28, 33

Characteristic	Scale Name	Scale Abbreviation	Items (questions measuring
			attribute)
Outcomes of Leadership	Extra Effort	EE	39, 42, 44
Outcomes of Leadership	Effectiveness	EFF	37, 40, 43, 45
Outcomes of Leadership	Satisfaction	SAT	38, 41

Appendix J - Interview Questionnaire Round 1

- 1. What change management projects have you undertaken?
- 2. Which staff occupations do you lead?
- 3. How long was your change management project process?
 What were the key success measures?
 Do you consider this project was successfully implemented?
 What were your key considerations for the change management project?
- 4. What strategies did you use for your change management project/process?
- 5. How are these strategies different from those you use daily to manage your staff?
- 6. What do you consider as your leadership style?
- 7. How long have you used this leadership style? Does this leadership style work for you in most of the situations?
- 8. Has your leadership style changed from the last project you lead? In what way has it changed?

Appendix K - Interview Questionnaire Round 2

- 1. What change management projects would you like to discuss for this interview round?
- 2. How long was your change management project process?
 - a. How was the key success criteria measured?
 - b. Do you consider this project was successfully implemented?
 - c. What were the key reasons for success/failure?
 - d. What were your key considerations for the change management project?
- 3. What strategies did you use for your change management project/process?
- 4. What key learning did you take from your previous project into this project? Are these key learning points documented somewhere where they are accessed by other staff that relieve your position?
- 5. Do you share your project successes or failures with others within your team or wider organisation? If not why? If so, how do you share them?
- 6. How are these strategies different from those that you used in your previous project? How are they also different from strategies you use daily to manage your staff?
- 7. Your self-reported leadership style was previouslyDo you consider that this leadership style has changed? In what way has it changed?
- 8. Have there been major changes or events that have affected your work environment? What are they?

Appendix L - Interview Questionnaire Round 3

- 1. What change management projects would you like to discuss for this interview round?
- 2. How long was your change management project process?
 - a. How was the key success criteria measured?
 - b. Do you consider this project was successfully implemented?
 - c. What were the key reasons for success/failure?
 - d. What were your key considerations for the change management project?
- 3. What strategies did you use for your change management project/process?
- 4. How have these strategies changed from those that you used in your previous project? How are they also different from strategies you use daily to manage your staff?
- Have you undertaken any change management training over the past 12 months to 2 years?
 Have you undertaken any in your working life?
- 6. Has your team changed over the change process?
- 7. What key learning did you take from your previous project into this project? Are these key learning points documented somewhere where they are accessed by other staff that relieve your position?
- 8. Do you share your project successes or failures with others within your team or wider organisation? If not, why? If so, how do you share them?
- 9. Your self-reported leadership style was previouslyDo you consider that this leadership style has changed? In what way has it changed?

- 10. Have there been major changes or events that have affected your work environment? What are they?
- 11. Have you completed the Final LSI questionnaires I send to you?

Appendix M – Interview Transcript Coding

Node Name		
Attitude to change		
Change affecting multiple stakeholders or systems		
Change leadership vs daily leadership		
Consultation		
Demographics		
Expected Outcomes		
Extra Comments		
Key Lessons Learnt		
Key Project considerations		
Leadership Style		
Management Responsibility		
Organisational Learning		
Process		
Projects		
Reason for project success		
Reasons for project failure		
Stakeholders involved		
Strategies used		
Trial before implementation		
Was Proiect Successful		

Appendix N - Round 1 – Sample Interview Transcript

- Interviewer: Alright. Thank you, Participant 2, for meeting me and allowing me the time to interview you for my PhD. As you are aware, my PhD topic is, "Leadership for Change Management, Critical success factors for managing change in Medical Imaging Departments in the public hospital system". And as you are aware also, we are using this study for the hospitals within the South Western Sydney LHD, with such as Campbelltown, Liverpool, Fairfield and Bankstown.
- Participant 2: and Camden
- Interviewer: no not Camden, Camden/Campbelltown is the same
- Participant 2: and Bowral?
- Interviewer: No, not Bowral. We don't manage the imaging.

So, I will give you the questions that we will walk through.

- Participant 2: Ok
- Interviewer: The purpose of this one is that it is a pilot study. And also it depend s whether I get enough responses or not, I might add the pilot to the final data collection.

So for the record, please describe your role in imaging?

- Participant 2: So I am the [position title] for South Western Sydney LHD. So I manage the [service type] aspects and [identifying word] manage.
- Interviewer: In terms of Change management, what projects have undertaken recently?
- Participant 2: Just for imaging?
- Interviewer: Yes just for Imaging.

- Participant 2: Uhh, I think we got the [....Manager]. That was quite hairy. What else have we done? Ahh we are currently doing the [a system name].
- Interviewer: Yes.
- Participant 2: And also we have done a lot of work on the back end of [a system] with notifications.
- Interviewer: Yes: Yes

Participant 2: [clear project description].

Interviewer: Right, thanks for that. So on those projects that you have led, perhaps we can discuss one for this interview.

- Participant 2: Yep.
- Interviewer: Perhaps we will leave the [a project name] for our second round of interviews.
- Participant 2: Yep that would be great.

Interviewer: So which one do you want to talk about?

- Participant 2: I don't know, you can choose. You were there. Do you want to talk about the back end of [a system]? We can do that one or do [a project name]?
- Interviewer: Let's do [a project name].
- Participant 2: Ok. Done
- Interviewer: Ok. So what sort of Project was it? Can you please describe the Priority Ordering project.
- Participant 2: So when we were having a look at our workflows in imaging, they all came in under the same, under basically where the patient was. Basically due to location. Round about 70% of all our ordering comes from ED and of course
we are getting busier especially at Liverpool, especially with one CT scanner.

Interviewer: Yes

Participant 2: That we were unable to look at getting these patients through especially within the 2 hour period, because that what they wanted us to do for all patients.

Interviewer: Yes.

Participant 2: There was no way for us to see which patients were more urgent than others or required a faster CT or any type of images from ED.

Interviewer: Uh humn.

Participant 2: So we had a look at changing the whole system to be based on priority rather than location based ordering.

Interviewer: Ok. So how long did was that change management process over?

Participant 2: It's still going on.

Interviewer: so you would say 1 to 2 years.

Participant 2: Yes.

Interviewer: So what were the key considerations for this project? I know you talked about the fact that you couldn't clearly identify ED patients, which ones were more urgent than the others

Participant 2: Yes

Interviewer: What are the key considerations therefore, for this project?

Participant 2: Umm the Interphase with [patient information system] was particularly important.

Involving all the key stakeholders especially ED and those requiring imaging such as ICU for MET calls etcetera.

Involving Clinicians outside of that.

So you had buy in by those Medical and Surgical Clinicians and as I said, eMR was a big thing.

- Interviewer: Yes
- Participant 2: And looking at our PACS/RIS system to see if it could actually do it.
- Interviewer: That's good. What sort of strategies did you use for this change management?

Participant 2: We had a Communication Plan.

- Interviewer: Yes.
- Participant 2: And the Communication Plan basically outlined our steps to get to the ending.
- Interviewer: Yes
- Participant 2: Lots and lots of meetings with the key stakeholders. Now that is also creating meetings ourselves, and involving people but also interphasing with existing meetings but also District meetings such as the [meting name] meeting.
- Interviewer: Yes
- Participant 2: Because we had to have a change management strategy with Sydney [LHD]. So we had to divorce from Sydney's eMR for ours. So tabling it with them and making them aware that this what we were going to do. Having a look at how we were going to do it first and then engaging.
- Interviewer: Ok. So from a governance perspective what would you think were the key considerations or the governance process for that project?

Participant 2: I think the first thing is to work out if you can do it, like what's going to happen? Can we do it? How is it going to look? Will it actually work? And I think that even when you are setting up the governance and structures. It's talking to the key stakeholders to begin with.

Interviewer: Yes

- Participant 2: Looking for those Champions. Seeing if you come up with an idea, seeing if it's actually going to benefit the other people. Because what you think may not actually benefit people at some stages.
- Interviewer: Yes
- Participant 2: So involving ED earlier on and involving eMR earlier on to say is this something you would like. Umm you know, we have looked into it, we know we can do this, is this something that you would like and then if they say it's something that we like then its developing that plan, having key milestones to hit.
- Interviewer: Yes
- Participant 2: Putting timelines around it.
- Interviewer: Yes
- Participant 2: And renegotiating those timelines if there is some issues. Having like a definite timeline for the project but also maintaining a bit of flexibility within those.
- Interviewer: Ok great. Thanks for that. In terms of your say from your management of staff. Are there strategies that you use to manage staff that are different to those you use to manage staff in projects?
- Participant 2: Umm, I think there is overlap.

Interviewer:	Yes
Participant 2:	There is definitely an overlap.
Interviewer:	Yes
Participant 2:	With managing staff on a day to day basis, I try to let them see the outcomes.
Interviewer:	Yaah
Participant 2:	So how are you going to get there. That's what you are doing in project management as well but you know, if they come across a problem then they should be able to see the outcomes or talking them through it so they can see it. How to get there. Looking at the barriers, looking at what options there are. Accountabilities, all those types of things.
Interviewer:	Yes
Participant 2:	But I think with a project of this size, you do tend to take the higher level stuff on yourself or involve your colleagues at that level to ensure that you are on the same page, to make sure that you have ticked all the really really big ones.
Interviewer:	Yes
Participant 2:	And then involve those key staff members to champion the

- Participant 2: And then involve those key staff members to champion the rest to work simultaneously with the other people with what they doing every day.
- Interviewer: In terms of the staff occupations, what sort of occupations do you manage or do you lead?
- Participant 2: Who do I manage, I don't manage anybody! Professionally manage?

Interviewer: Professionally manage and lead as well.

Participant 2: [profession], definitely of course, [another profession] including [another profession] of course. [different profession] as far as more of that background work, teaching and how to kind of get there.

Interviewer: Yes

Participant 2: Primarily [profession], because you are kind of writing things for them and showing them the way to do things. [Profession], I think that's about it.

Interviewer: Ok, great. Thank You. Now we will talk a little bit about your leadership style.

So Leadership style in the context of this project, I am looking at, I know you have undertaken some management studies. I am looking at whether your leadership style is perhaps; Laissez-fairer – you let things go or you are a participative leader who let staff and yourself participates as part of the leadership team or you are a democratic leader who allows you staff to give you their views before you make your final decision. Or you are an autocratic leader or a transformative leader?

Participant 2: Yeah yeah.

Interviewer: You want to come from one perspective and then arrive at a different perspective and take people through a journey and transform or whether we are looking in terms of authentic leadership where your views or moral campus sort of defines how you make your decisions subconsciously? You are thinking if it's me, having this, how would I like a decision being made this way and that way, it drives your decision making.

Could you please describe your leadership style in sort of that context?

Participant 2: I think that you stop and change depending on where you are.

Interviewer: Yes

Participant 2: And what you do. I think I have written lots of assignments on this. I don't think you are one type of leader all the times.
I think there is a place for all of those styles. But there's more places for some than others.

Interviewer: Yes

Participant 2: If I had to pick one, probably transformational.

Interviewer: Transformational

Participant 2: Transformational or what's the other one?

Interviewer: Transformation, Authentic, Democratic, Participative

- Participant 2: Participative
- Interviewer: ok

Participant 2: Yeah, because you do, you have to get people involved, but you have to get, the whole idea of a leader is not just to lead a team. The whole idea is to get people on board, have a look at their visions and your vision. Keep getting back to them and say, have I missed anything? How can I help you?

Interviewer: Yes

Participant 2: Because one day you want people to take your job.

Interviewer: Absolutely

Participant 2: You want to push them as far as they can go and support them.

Interviewer: Great, thanks for that.

So if we are talking transformational leadership and participative leadership, out of 100, what or where would

you scores be? Would you say you are 50% a transformational leader or you are 20% one or what would you say?

Participant 2: Probably 70/30.

Interviewer: 70% transformational and the other 30% more participative?

- Participant 2: Yeah
- Interviewer: ok

Participant 2: They kind of go hand in hand. Sometimes

Interviewer: So do you consider that this leadership style works for you n most situations?

Participant 2: Yeah in most situations. Sometimes it does not work.

Interviewer: Ok, in the cases that it has not worked, what have you tended to do? What sort of style have you adopted?

- Participant 2: I think when you come across people who do not want to go with you, who can't see vision, who are kind of trapped, like they don't want to be there.
- Interviewer: Yes
- Participant 2: In all seriousness it is easier to say maybe this is not the project for you to be involved in. but sometimes as you know in health, you are told, you are on the working party, that is it! You are there.

Interviewer: Yes

Participant 2: and you can't kind of swap and change. Then I think it becomes more autocratic. Not that you ever wanted to.

Interviewer: ok, fair enough

Participant 2: But that would be very very rare. You can normally find something that would engage that person, that would be

their passion. If you can find that person's passion and then you push them forward and kind of go, how are we going to do this? I think this is great if you can take on that role.

- Interviewer: Yes, ok, that's good. So from the project that we were talking about, before the [project name], do you consider your leadership style as having changed in that project, say from prior projects that you have led? Or did you try a certain style of leadership and then you decided after leading those projects that well, this style has not worked as well as it should. And for this new project, I will probably use this new tact?
- Participant 2: No, Not really. But then again I have done a few projects before in my time. So it's kind of not the first. But I think if it was earlier in down the track then I might swap and change.
 I think when I started doing projects, because you did not want to step on anyone's toes, you were very very laissezfaire.

Interviewer: Yes

Participant 2: But as time got on, you learnt how to engage better and not lead, but be a part of.

Interviewer: Oh Ok, that's good. so in leading projects, what do you think are the key considerations that you have embedded in your leadership for change management?

Participant 2: In what sense?

Interviewer: In the sense that perhaps that you collaborate more or you consult more or what key ingredients do you think contribute to the success or failure of your projects?

Participant 2: Collaboration, for sure. And I think that was in one of my earlier projects an issue. I didn't collaborate well and boy did I learn! Because everybody just went no!

Interviewer: Ok

Participant 2: So I think it's not only just the collaboration but it's also the feedback. That continual feedback loop from people, getting their ideas, getting their visions and the taking them with the group and embedding them in the way forward. Making sure everybody is kind of involved.

Interviewer: ok that's fine

So from you leading projects within this District, what's your experience from learning from past projects that we have done?

Participant 2: Definitely engagement and engagement across the board.

- Interviewer: No. I mean in terms of the organisation. Is the organisation an environment where perhaps we have a similar sort of project 2 years earlier and those learnings are carried forward to the next project? What's your experience with that?
- Participant 2: Yeah, I think they are embedded in life. So your projects that have been successful in uptake by people, are now part of the experience of Health really.

Interviewer: Ok, that's good.

Participant 2: I hope so.

Interviewer: Alright, the other thing that I will probably want to talk about is more around the interview itself. Is there anything that you were expecting to be asked that you were not asked?

Participant 2: No.

Interviewer: What would you like to see in next interview rounds?

- Participant 2: I think it would be good to, I know you are talking about Medical Imaging Leadership. I think it would be good to involve multi-disciplinary projects or multi stream projects. There is lots and lots of such projects that cross over and its really really hard [Interviewer] to say what type of leader you are.
- Interviewer: Ok
- Participant 2: With your style. So like put yourself into a box, because all the literature says this is the way to go. Somebody will come up with a new model of care and say this is the type of leader you are and this is this box.
- Interviewer: Absolutely, I agree entirely. So, I have done Myers Briggs and you tend to find that you are EJP or whatever, but there is times that I find myself that my leadership style changes just depending on the situation and what you have to lead and what you have to manage. It just changes all the time.
- Participant 2: It's hard to say which one you are but I think the bottom line is how you manage difficulties.
- Interviewer: Yes
- Participant 2: How you get through what you need to get through?
- Interviewer: So talking difficulties, what sort of experience do you have managing challenging situations. I know in health you get a lot of headstrong people?
- Participant 2: I think there is a thing that you need to do ask questions. What's the issue? What's the problem? What is it? And I think you will drill down to the bottom of it. Sometimes, there is a lot of fluff around it.

Interviewer: Yes

- Participant 2: Sometime people just want to cry. And you need to look at what are the barriers to something, like think of why is this a problem? Tell me about the barriers you have got. What are the options that we have to get out of this, what can you see happening and then say ok, if you can see this happening, would you like to lead this?
- Interviewer: Yes
- Participant 2: How can I help you to get there and then kind of sum everything up. I think that way they feel empowered because, they have heard their gripe, you have worked out their gripe, taken out all the fluff and emotion and worked out the options. Worked out how they are going to do it. So, they are engaged in their problem solving.
- Interviewer: Yes

Participant 2: And it's taken me years and years to work that out.

- Interviewer: That's good!
- Participant 2: Rather than just here is the problem, here is the solution! And I think that new leaders in particular, doesn't matter what project, it doesn't matter how many times you write a Gantt chart. If that person is not engaged or that person or area of specialty is not engaged, then they are just going to put up walls, walls and walls.
- Interviewer: Alright, thank you very much. That concludes our formal interview.
- Participant 2: Thanks [Interviewer].